**Table of Contents** 

### SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

# **SCHEDULE 13D**

**Under the Securities Exchange Act of 1934** 

# **METAVANTE TECHNOLOGIES, INC.**

(Name of Issuer)

COMMON STOCK, PAR VALUE \$0.01 PER SHARE (Title of Class of Securities)

> 591407101 (CUSIP Number)

SCOTT A. ARENARE, ESQ. WARBURG PINCUS LLC 466 LEXINGTON AVENUE NEW YORK, NY 10017 (212) 878-0600 (Name, Address and Telephone Number of Person Authorized to Receive Notices of Communication)

Copy to: ANDREW R. BROWNSTEIN, ESQ. IGOR KIRMAN, ESQ. WACHTELL, LIPTON, ROSEN & KATZ 51 WEST 52ND STREET NEW YORK, NY 10019 (212) 403-1000

If the filing person has previously filed a statement on Schedule 13G to report the acquisition that is the subject of this Schedule 13D, and is filing this schedule because of §§ 240.13d-1(e), 240.13d-1(f) or 240.13d-1(g), check the following box. o

Note: Schedules filed in paper format shall include a signed original and five copies of the schedule, including all exhibits. See § 240.13d-7 for other parties to whom copies are to be sent.

\* The remainder of this cover page shall be filled out for a reporting person's initial filing on this form with respect to the subject class of securities, and for any subsequent amendment containing information which would alter disclosures provided in a prior cover page.

The information required on the remainder of this cover page shall not be deemed to be "filed" for the purpose of Section 18 of the Securities Exchange Act of 1934 ("Act") or otherwise subject to the liabilities of that section of the Act but shall be subject to all other provisions of the Act (however, see the Notes).

November 1, 2007 (Date of Event Which Requires Filing of This Statement) statement on Schedule 13G to report the acquisition that is the subject of .13d-1(f) or 240.13d-1(g), check the following box. o

CUSIP N	No	<b>591</b> 4	07101				13D		Page	2	of	19
1	WPM, I	L.P. R <b>I.R.S.</b>	PORTING PEI IDENTIFICAT		F ABOVE	E PERSOI	NS					
2	CHECI (a) o (b) ☑	K THE	APPROPRIAT	E BOX IF A M	MEMBER	R OF A G	ROUP					
3	SEC US	SE ONI	X									
4	SOURC	CE OF I	FUNDS									
5	CHECI o	K BOX	IF DISCLOSU	RE OF LEGA	L PROC	CEEDING	S IS REQUIRED P	PURSUANT TO IT	ГЕМ 2(d)	) or 2(e)		
6	<b>CITIZI</b> Delawa		OR PLACE O	F ORGANIZA	ATION							
NUME	BER OF	7	SOLE VOTI	NG POWER								
BENEF	ARES ICIALLY ED BY	8	<b>SHARED VC</b> 29,732,214	DTING POWE	ÊR							
REPO	ACH RTING RSON	9	SOLE DISPO	DSITIVE POW	VER							
<b>W</b> ]	ITH	10	SHARED DI 29,732,214	SPOSITIVE P	OWER							
11	<b>AGGR</b> 29,732,2		AMOUNT BE	NEFICIALLY	( OWNEI	D BY EA	CH REPORTING	PERSON				
12	CHECI	K BOX	IF THE AGGR	EGATE AMO	OUNT IN	I ROW (11	I) EXCLUDES CE	RTAIN SHARES				
13	<b>PERCE</b> 25%	ENT OF	CLASS REPR	ESENTED BY	Y AMOU	UNT IN RO	OW (11)					
14	<b>TYPE</b>	OF REF	PORTING PER	SON								

CUSIP N	lo.	5914	07101				13D		Pa	age [	3	of	19
1	WPM, C	GP, LLC : <b>I.R.S.</b> 1	PORTING PEH		ABOVI	E PERS	SONS						
2	<ul> <li>CHECH</li> <li>(a) o</li> <li>(b) ☑</li> </ul>	K THE .	APPROPRIAT	E BOX IF A M	<b>IEMBE</b> I	ER OF A	GROUP						
3	SEC US	SE ONL	Y										
4	SOURC N/A	CE OF I	UNDS										
5	CHECH 0	K BOX	IF DISCLOSU	RE OF LEGAI	L PROC	CEEDIN	NGS IS REQUIRI	ED PURSUAN	T TO ITEM	2(d)	or 2(e)		
6	<b>CITIZE</b> Delawa		OR PLACE O	F ORGANIZA	ATION								
NUMB	ER OF	7	SOLE VOTIN	NG POWER									
BENEFI	RES CIALLY ED BY	8	<b>SHARED VC</b> 29,732,214	DTING POWE	R								
REPO	CH RTING SON	9	SOLE DISPO	DSITIVE POW	/ER								
WI	ТН	10	SHARED DIS 29,732,214	SPOSITIVE P	OWER								
11	AGGRI 29,732,2		AMOUNT BE	NEFICIALLY	OWNE	ED BY E	EACH REPORTI	NG PERSON					
12	CHECH o	K BOX	IF THE AGGR	EGATE AMO	OUNT IN	N ROW	(11) EXCLUDES	S CERTAIN SH	IARES				
13	<b>PERCE</b> 25%	NT OF	CLASS REPR	ESENTED BY	Y AMOU	UNT IN	ROW (11)						
14	TYPE (	OF REP	PORTING PER	SON									

CUSIP N	No.	5914	07101	]				13D				Page	4	0	f	19
	NAME	OF RE	PORTING PE	RSONS												
1	Warburg	Pincus	Private Equity	IX, L.P.												
1	S.S. OR	I.R.S.	IDENTIFICAT	TION NOS. OI	F ABOV	VE F	PERSON	IS								
	20-2975	990														
	CHECK	K THE	APPROPRIAT	E BOX IF A	MEMBI	BER (	OF A GI	ROUP								
2	(a) o															
	(b) 🗹															
3	SEC US	E ONL	Y													
4	SOURC	E OF I	FUNDS													
-	N/A															
5	CHECH	K BOX	IF DISCLOSU	RE OF LEGA	AL PRO	OCEI	EDINGS	5 IS REQU	JIRED PUI	RSUAN	T TO ITE	E <b>M 2(d)</b>	or 2(e)			
	0															
6	CITIZE	NSHI	OR PLACE (	)F ORGANIZ	ATION	N										
	Delawar	e														
		7	SOLE VOTI	NG POWER												
	BER OF		0													
	ARES ICIALLY	8	SHARED VO	OTING POWE	≤R											
	ED BY		29,732,214													
	ACH RTING	9	SOLE DISPO	OSITIVE POV	VER											
	SON		0													
W]	ITH	10	SHARED DI	SPOSITIVE F	POWER	R										
			29,732,214													
11	AGGRI	EGATE	AMOUNT BE	NEFICIALLY	YOWN	NED	BY EAC	CH REPO	RTING PE	RSON						
	29,732,2															
12	CHECH	K BOX	IF THE AGGE	<b>REGATE AM</b>	JUNT I	IN R	ROW (11)	) EXCLU	DES CERT	I'AIN SH	ARES					
	0															
13	PERCE	NT OF	CLASS REPF	ESENTED B	<b>ү</b> АМО	OUN	T IN RO	OW (11)								
	25%		000000000000000000000000000000000000000													
14	ТҮРЕ (	)F REF	ORTING PEF	ISON												
_	PN															

CUSIP I	No.	<b>591</b> 4	407101			13D		Page	5	of	19
	NAME	OF RE	PORTING PE	RSONS							
1	Warburg	g Pincus	IX LLC								
I	S.S. OR	I.R.S.	IDENTIFICAT	TON NOS. OF A	BOVE PERS	SONS					
	20-2975	945									
	CHECH	K THE	APPROPRIAT	E BOX IF A ME	MBER OF A	A GROUP					
2	(a) o (b) ☑										
	SEC US	SE ONI	Ŋ								
3											
4	SOURC	CE OF I	FUNDS								
4	N/A										
5	CHECH	K BOX	IF DISCLOSU	RE OF LEGAL I	PROCEEDI	NGS IS REQUIRED	PURSUANT TO IT	EM 2(d)	or 2(e)		
	0										
6			Y OR PLACE C	OF ORGANIZAT	ION						
	New Yo	rk	SOLE VOTI	NG POWER							 
NILIMI	BER OF	7	0								
	ARES		-	TING POWER							
	ICIALLY ED BY	8	29,732,214								
EA	СН	•	1	OSITIVE POWE	R						
	RTING RSON	9	0								
W	ITH	10	SHARED DI	SPOSITIVE POV	WER						
			29,732,214								
11	AGGRI	EGATE	AMOUNT BE	NEFICIALLY O	WNED BY I	EACH REPORTING	<b>G PERSON</b>				
	29,732,2										
12	CHECI	V ROX	IF THE AGGF	LEGATE AMOUI	NT IN ROW	' (11) EXCLUDES C	EKTAIN SHARES				
	o PERCE	NT OF	CLASS REPR	ESENTED BY A	MOUNT IN	ROW (11)					 
13	25%	01				()					
		OF REF	PORTING PER	SON							
14	CO										

CUSIP N	No	5914	07101	]			13D			Page	6	of	19
1	Warburg	g Pincus R <b>I.R.S.</b>	PORTING PE Partners, LLC IDENTIFICA		F ABOVI	E PERSC	DNS						
2	CHECI (a) o (b) ☑ SEC US			TE BOX IF A N	MEMBEI	CR OF A	GROUP						
3 4	SOURC N/A	CE OF I	FUNDS										 
5	CHECI o	K BOX	IF DISCLOS	URE OF LEGA	L PROC	CEEDIN	GS IS REQU	IRED PURSUA	ANT TO ITH	E <b>M 2(d)</b>	or 2(e)		
6	CITIZI New Yo		OR PLACE	OF ORGANIZ	ATION								
SHA BENEF	BER OF ARES ICIALLY	7 7 8	0 SHARED V	ING POWER	ER								
EA REPO PEF	ED BY ACH RTING RSON	9	0	OSITIVE POW									
<b>W</b>	AGGR	10 Egate	29,732,214	ISPOSITIVE P			ACH REPOR	TING PERSO	N				
11 12	29,732,7 CHECI		IF THE AGG	REGATE AMC	DUNT IN	N ROW (	11) EXCLUD	ES CERTAIN	SHARES				 
13	0 <b>PERCE</b> 25%	ENT OF	CLASS REP	RESENTED BY	Y AMOU	UNT IN F	ROW (11)						
14	TYPE C	OF REF	PORTING PE	RSON									

CUSIP	No.	5914	07101	]		13D		Pa	ge 7	of	19
			PORTING PE	RSONS							
1	Warburg			TION NOS. OF	ABOVE PI	ERSONS					
	13-6358			1011100101							
2	CHECK (a) o (b) ☑	THE .	APPROPRIAT	E BOX IF A M	IEMBER O	DF A GROUP					
3	SEC US	E ONL	X								
4	SOURC	E OF I	FUNDS								
5	CHECK	BOX	IF DISCLOSU	RE OF LEGA	L PROCEE	EDINGS IS REQ	UIRED PURSUANT	TO ITEM 2	(d) or 2(e)		
6	CITIZE New Yor		OR PLACE (	DF ORGANIZA	ATION						
NUMI	BER OF	7	SOLE VOTI	NG POWER							
BENEF	ARES ICIALLY ED BY	8	SHARED VC 29,732,214	DTING POWE	R						
REPO	ACH RTING RSON	9	SOLE DISPO	DSITIVE POW	VER						
W	ITH	10	SHARED DI 29,732,214	SPOSITIVE P	OWER						
11	AGGRE		AMOUNT BE	NEFICIALLY	Y OWNED B	<b>BY EACH REPC</b>	PRTING PERSON				
12	CHECK o	BOX	IF THE AGGE	LEGATE AMO	OUNT IN RO	OW (11) EXCLU	DES CERTAIN SHA	ARES			
13	<b>PERCE</b> 25%	NT OF	CLASS REPR	ESENTED BY	Y AMOUNT	「 IN ROW (11)					
14	TYPE C	)F REP	PORTING PER	SON							

CUSIP N	No.	5914	07101				13D		Page	8	of	19
		00.000										
			PORTING PEF	SUNS								
1	Warburg											
	S.S. OR	I.R.S.	IDENTIFICAT	ION NOS. OF	ABOVE	E PERSOI	NS					
	13-3536	050										
	CHECH	K THE	APPROPRIAT	E BOX IF A M	IEMBER	R OF A G	ROUP					
2	(a) o											
	(b) 🗹 SEC US	SE ONL	Y									
3												
4	SOURC	CE OF H	FUNDS									
4	N/A											
5	CHECH	K BOX	IF DISCLOSU	RE OF LEGAI	L PROCI	EEDING	S IS REQUIRED	PURSUANT TO IT	EM 2(d)	) or 2(e)		
3	0											
6	CITIZE	ENSHIP	OR PLACE O	F ORGANIZA	ATION							
•	New Yo	rk										
		7	SOLE VOTIN	NG POWER								
	BER OF		0									
-	ARES ICIALLY	8	SHARED VU	TING POWEI	К							
	ED BY CH		29,732,214	OSITIVE POW	/FD							
REPO	RTING	9		,5111VL10W	LIX							
	SON ITH		0 SHARED DIS	SPOSITIVE PO	OWER							
		10	29,732,214									
	AGGRI	EGATE		NEFICIALLY	OWNEI	D BY EA	CH REPORTING	PERSON				
11	29,732,2	214										
10			IF THE AGGR	EGATE AMO	UNT IN	<b>ROW (1</b> 1	1) EXCLUDES CI	ERTAIN SHARES				
12	0											
13	PERCE	NT OF	CLASS REPR	ESENTED BY	AMOU	INT IN RO	OW (11)					
13	25%											
14	ТҮРЕ (	OF REP	PORTING PER	SON								
	00											

CUSIP N	No.	5914	07101				13D			Page	<u> </u>	)	] of	19
	NAME	OF RE	PORTING PEF	SONS										
1	Charles 1	R. Kaye	2											
	S.S. OR	I.R.S. 1	IDENTIFICAT	ION NOS. OF	F ABOVE	E PERSC	ONS							
	CHECK	THE	APPROPRIAT	E BOX IF A N	<b>IEMBE</b>	R OF A	GROUP							
2	(a) o (b) ☑													
3	SEC US	E ONL	X											
	SOURC	E OF I	JUNDS											
4	N/A													
_		BOX	IF DISCLOSU	RE OF LEGA	L PROC	CEEDING	GS IS REQUIRI	ED PURSUAN	T TO ITE	M 2(d)	or 2(e	:)		
5	0													
		NSHIP	OR PLACE O	F ORGANIZ	ATION									
6	United S	tates of	America											
			SOLE VOTIN	NG POWER										
NUMF	BER OF	7	0											
	RES		SHARED VO	TING POWE	R									
	ICIALLY ED BY	8	29,732,214											
	CH		SOLE DISPO	SITIVE POW	VER									
	RTING SON	9	0											
	TH		SHARED DIS	SPOSITIVE P	OWER									
		10	29,732,214											
	AGGRE	GATE		NEFICIALLY	OWNE	ED BY EA	ACH REPORTI	NG PERSON						
11	29,732,2	14												
			IF THE AGGR	EGATE AMC	OUNT IN	N ROW (	11) EXCLUDES	CERTAIN SH	HARES					
12	0													
		NT OF	CLASS REPR	ESENTED BY	Y AMOU	UNT IN F	ROW (11)							
13	25%													
		F REP	ORTING PER	SON										
14	IN													

CUSIP N	Io	5914	07101									13D							Р	age		10	of	19	
	NAME	OF RE	PORTING PEH	RSONS																					
1	Joseph P	. Landy																							
	S.S. OR	<b>I.R.S.</b> ]	IDENTIFICAT	'ION NOS. O	F ABC	юv	VI	/E	PE	RS	ON	S													
_	CHECK	THE	APPROPRIAT	E BOX IF A	MEM	1BE	BEI	ER	OF	FA	GR	OUP													
2	(a) o (b) ☑																								
3	SEC US	E ONL	Y																						
	SOURC	E OF F	UNDS																				 	 	
4	N/A	-																							
5	CHECK	BOX	IF DISCLOSU	RE OF LEGA	AL PR	RO	DC	CE	EED	DIN	GS	IS R	EQUI	RED	) PUI	RSU	JANT	TO	TEM	2(d)	or 2	(e)			
5	0																								
6	CITIZE	NSHIP	OR PLACE O	F ORGANIZ	ZATIO	ON	N	[																	
U	United S	tates of	America																						
		-	SOLE VOTI	NG POWER																					
NUMB	ER OF	7	0																						
	RES		SHARED VO	TING POW	ER																				
BENEFI	CIALLY ED BY	8	29,732,214																						
EA	СН		SOLE DISPO	SITIVE PO	WER																				
	RTING	9																							
	SON TH		0 SHARED DIS	SDOSITIVE	DOWT		р	,															 	 	
**1	111	10	29,732,214	SPUSITIVE	FOWE	LK	Л																		
	AGGRE	GATE	AMOUNT BE	NEFICIALL	X OW	WNI	NE	ED	) BY	Y E.	AC	H RE	POR	ΓΙΝΟ	G PE	RSC	ON								
11	29,732,2																								
			IF THE AGGR	EGATE AM	OUNT	II T	IN	NI	RO	W (	(11)	EXC	LUDI	ES C	ERT	TAIN	N SHA	ARES	5						
12	0										. ,														
		NT OF	CLASS REPR	ESENTED B	BY AM	иоі	OU	UN	NT	IN I	RO	W (11	.)												
13	25%																								
	TYPE O	F REP	ORTING PER	SON																					
14	IN																								

Page 11 of 19

### **TABLE OF CONTENTS**

Item 1. Security and IssuerItem 2. Identity and BackgroundItem 3. Source and Amount of Funds or Other ConsiderationItem 4. Purpose of the TransactionItem 5. Interest in Securities of the IssuerItem 6. Contracts, Arrangements, Understandings or Relationships with Respect to Securities of the IssuerItem 7. Material To Be Filed as ExhibitsSIGNATURESINDEX OF EXHIBITSEX-99.1: Joint Filing AgreementEX-99.2: Investment AgreementEX-99.3: Shareholders AgreementEX-99.4: Stock Purchase Right Agreement

Information in respect of each Reporting Person is given solely by such Reporting Person and no Reporting Person has responsibility for the accuracy or completeness of information supplied by any other Reporting Person.

### Item 1. Security and Issuer

This statement on Schedule 13D (this "Statement") relates to the common stock, par value \$0.01 per share, of Metavante Technologies, Inc. (the "Common Stock"), a Wisconsin corporation formerly named Metavante Holding Company ("Metavante Technologies"). The principal executive offices of Metavante Technologies are located at 4900 West Brown Deer Road, Milwaukee, Wisconsin 53223.

### Item 2. Identity and Background

(a) This Statement is being filed on behalf of WPM, L.P., a Delaware limited partnership ("WPM"), WPM GP, LLC, a Delaware limited liability company and the sole general partner of WPM ("WPM GP"), Warburg Pincus Private Equity IX, L.P., a Delaware limited partnership and the sole member of WPM GP ("WP IX"), Warburg Pincus IX LLC, a New York limited liability company and the sole general partner of WP IX ("WP IX LLC"), Warburg Pincus Partners, LLC, a New York limited liability company and the sole member of WP IX LLC ("WP Partners"), Warburg Pincus & Co., a New York general partnership and the managing member of WP Partners ("WP"), Warburg Pincus LLC, a New York limited liability company that manages WP IX ("WP LLC"), and Messrs. Charles R. Kaye and Joseph P. Landy, each a Managing General Partner of WP and Managing Member and Co-President of WP LLC (each of the foregoing, a "Reporting Person," and collectively, the "Reporting Persons"). The agreement among the Reporting Persons to file this Statement jointly in accordance with Rule 13d-1(k) of the Securities Exchange Act of 1934, as amended (the "Exchange Act") is attached hereto as Exhibit 1.

(b) The address of the principal business and principal office of the Reporting Persons is c/o Warburg Pincus LLC, 466 Lexington Avenue, New York, New York 10017. The general partners of WP, the members and managing directors of WP LLC, the executive officers of WPM GP, and their respective business addresses, are set forth on Schedule I hereto, which is incorporated herein by reference.

(c) The principal business of WPM is to make and hold an equity investment in Metavante Technologies (see Item 4 hereto). The principal business of WPM GP is acting as general partner of WPM. The principal business of WP IX is that of making private equity and related investments. The principal business of WP IX LLC is acting as general partner of WP IX. The principal business of WP Partners is acting as general partner to certain private equity funds and as the sole member of WP IX LLC. The principal business of WP is acting as the managing member of WP Partners. The principal business of WP LLC is managing certain private equity funds, including WP IX. The principal businesses of each of Messrs. Kaye and Landy is acting as Managing General Partner of WP and Co-President and Managing Member of WP LLC. The principal occupation of each of the general partners of WP, the members and managing directors of WP LLC and the executive officers of WPM GP is set forth on Schedule I hereto, which is incorporated herein by reference.

(d) During the last five years, none of the Reporting Persons and, to the knowledge of the Reporting Persons, none of the partners, members and managing directors named on Schedule I, have been convicted in a criminal proceeding (excluding traffic violations or similar misdemeanors).

(e) During the last five years, none of the Reporting Persons and, to the knowledge of the Reporting

# CUSIP No. 591407101 13D Page 12 of 19

Persons, none of the partners, members and managing directors named on Schedule I, have been a party to a civil proceeding or a judicial or administrative body of competent jurisdiction and as result of such proceeding was or is subject to a judgment, decree or final order enjoining future violations of, or prohibiting or mandating activities subject to, federal or state securities laws or finding any violation with respect to such laws.

(f) WPM, WPM GP and WP IX are organized under the laws of Delaware. WP IX LLC, WP Partners, WP and WP LLC are organized under the laws of New York. Messrs. Kaye and Landy are citizens of the United States of America, and except as otherwise indicated on Schedule I, each of the individuals referred to on Schedule I hereto is a citizen of the United States of America.

### Item 3. Source and Amount of Funds or Other Consideration

The aggregate purchase price for the 29,732,214 shares of Common Stock acquired by WPM was \$625,000,000. WPM obtained its funds from a capital contribution from WP IX.

The above mentioned shares of Common Stock automatically converted on a one-for-one basis from shares of Metavante Technologies' Class A common stock, par value \$0.01 per share (the "Class A Common Stock"), that were initially issued and sold to WPM. Descriptions of the investment by WPM, the Class A Common Stock and its conversion into Common Stock are included in the proxy statement/prospectus—information statement (the "Prospectus") included as part of the Registration Statement on Form S-4, as amended (File No. 333-143143), of Metavante Technologies filed with the Securities Exchange Commission (the "SEC") and in Item 4 hereto.

### Item 4. Purpose of the Transaction

The purchase by WPM of the Class A Common Stock was effected because of the belief that the Common Stock represented, and continues to represent, an attractive investment. The Reporting Persons beneficially own the Common Stock as an investment. Subject to the limitations described below in this Item 4, the Reporting Persons from time to time may decide to increase or decrease their investment in Metavante Technologies by purchasing shares of Common Stock or other capital stock of Metavante Technologies in open market or private transactions or otherwise. The timing and amount of any such increase or decrease may depend upon the price and availability of shares of Metavante Technologies' capital stock, subsequent developments affecting Metavante Technologies, Metavante Technologies' business and prospects, other investment and business opportunities available to the Reporting Persons, general stock market and economic conditions, tax consideration and other factors considered relevant.

On April 3, 2007, WPM entered into an Investment Agreement (the "Investment Agreement") with M&I LLC (formerly known as Marshall & Ilsley Corporation ("Old M&I")) ("MI LLC"), Metavante Corporation ("Metavante"), Metavante Technologies, and Montana Merger Sub Inc. ("Merger Sub"). Pursuant to the terms of the Investment Agreement, on November 1, 2007, the date of the closing, (i) WPM purchased 29,732,214 shares of Class A Common Stock for an aggregate purchase price of \$625,000,000, (ii) Metavante Technologies and WPM entered into a Stock Purchase Right Agreement (the "Stock Purchase Right Agreement"), and (iii) Metavante Technologies and WPM entered into a Shareholders Agreement (the "Shareholders Agreement").

The following summary of the Investment Agreement, Shareholders Agreement and the Stock Purchase Right Agreement is not intended to be complete and is qualified in its entirety by reference to the full text of such agreements, copies of which are filed as Exhibits 2, 3 and 4 hereto, respectively, and which are incorporated herein by reference.

13D

#### Investment Agreement

Pursuant to the Investment Agreement and the transactions and transaction documents contemplated thereby, Old M&I and Metavante were split into two independent public companies, Marshall & Ilsley Corporation ("Marshall & Ilsley") and Metavante Technologies, and WPM paid Metavante Technologies \$625,000,000 for newly issued shares of Class A Common Stock, which converted automatically into shares of Common Stock at 12:01 a.m., Eastern Daylight Time, on November 2, 2007, and which shares of Common Stock represented approximately 25% of the Common Stock. The shareholders of Old M&I received the remaining 75% of the Common Stock. Except for WPM, holders of shares of Common Stock (i.e., former holders of shares of Old M&I common stock prior to the transactions contemplated by the Investment Agreement) as of the record date established therefor received three shares of Marshall & Ilsley common stock for each share of Common Stock held (the "Share Distribution") and retained their shares of Common Stock. The Class A Common Stock acquired by WPM was not entitled to receive shares of Marshall & Ilsley common stock in the Share Distribution. As a result of the transactions, Marshall & Ilsley now owns and operates Old M&I's banking business and Metavante Technologies now owns and operates Old M&I's Metavante business.

The Investment Agreement also provided for the manner of allocation of fees and expenses between the parties, certain governance matters in respect of the size and composition of the board of directors of Metavante Technologies, and the entry between Metavante Technologies and WPM into the Stock Purchase Right Agreement and the Shareholders Agreement.

### Shareholders Agreement

As noted above, on November 1, 2007, Metavante Technologies and WPM entered into the Shareholders Agreement.

Pursuant to the Shareholders Agreement, following the 2008 annual meeting of Metavante Technologies and until the earlier of (i) the date on which the Investor Percentage Interest (as defined in the Shareholder Agreement) is less than 7.5% and the fair market value of the shares of Common Stock held by WPM and/or its affiliates is less than \$150,000,000, (ii) ten years after the closing of the transactions or (iii) the termination of the Shareholders Agreement, WPM will have the right to designate for nomination and election, the following number of directors: (A) so long as the Investor Percentage Interest equals or exceeds 17.5%, WPM shall have the right to nominate three directors; (B) if the Investor Percentage Interest is less than 17.5% but equals or exceeds 7.5%, WPM shall have the right to nominate two directors; (C) if the Investor Percentage Interest is less than 7.5% but the fair market value of the shares of Common Stock held by WPM and/or its affiliates equals or exceeds \$150,000,000, WPM shall have the right to nominate on edirector; and (D) if the Investor Percentage Interest is less than 7.5% and the fair market value of the shares of Common Stock held by WPM and/or its affiliates equals or exceeds \$150,000,000, WPM shall have the right to nominate one director; and (D) if the Investor Percentage Interest is less than 7.5% and the fair market value of the shares of Common Stock held by WPM and/or its affiliates equals or exceeds \$150,000,000, WPM shall have the right to nominate one director; and (D) if the Investor Percentage Interest is less than 7.5% and the fair market value of the shares of Common Stock held by WPM and/or its affiliates is less than \$150,000,000, WPM shall not have the right to nominate any directors.

The Shareholders Agreement provides that the board of directors of Metavante Technologies will have three standing committees, an audit committee, a compensation committee, and a nominating and corporate governance committee. Each of the committees will be composed of three members, at least one of whom, to the extent permissible by law and the rules of the New York Stock Exchange ("NYSE"), will be a WPM designee. The Shareholders Agreement provides that the chairman of the compensation committee shall be a WPM designee. All of the members of each committee are required to be independent under the rules of the NYSE.

In addition, pursuant to the Shareholders Agreement, WPM has agreed to certain restrictions on its ability

13D

```
Page 14 of 19
```

to purchase additional shares of Common Stock or otherwise control Metavante Technologies, including an agreement not to acquire any additional Common Stock if such purchase would result in an Investor Percentage Interest in excess of 40%. These restrictions terminate on the earliest of (i) the two year anniversary of the closing date, (ii) the date on which any WPM designee that WPM is entitled to designate is not elected to the board of directors of Metavante Technologies and is not otherwise appointed to the board of directors, and (iii) the date of a change of control of Metavante Technologies. In addition, these restrictions do not apply at any time after (A) the board of directors resolves to pursue a transaction that is contemplated to result in a change of control of Metavante Technologies or (B) the board of directors approves, recommends or accepts a transaction that would result in a change of control proposed by any person other than WPM. These restrictions, however, become operative again once the board of directors (1) resolves not to pursue any such transaction or (2) rejects or announces that it has withdrawn its recommendation of any such transaction.

WPM further agreed in the Shareholders Agreement not to (i) sell its shares of Common Stock for a period of one year after the closing of the transactions and (ii) from the first anniversary of the closing to the second anniversary of the closing, sell its shares of Common Stock if such sales would result in an Investor Percentage Interest of less than 17.5%, in each case, subject to certain exceptions.

The Shareholders Agreement provides for demand registration rights for WPM that will require Metavante Technologies to, upon the request of WPM, register its shares of Common Stock with the SEC and permit WPM to sell such registered shares of Common Stock to the public, subject to specified conditions and a maximum of four such demand registrations. Furthermore, pursuant to the Shareholders Agreement, if Metavante Technologies proposes to register any of its securities on a form that may include registerable shares held by WPM, WPM will have the right to request that all or any part of its registrable shares be included in the registration, subject to specified conditions.

In addition to certain supermajority approval provisions, the Shareholders Agreement provides that until the Investor Percentage Interest is less than 10% of Metavante Technologies, if Metavante Technologies makes an offering of Common Stock or securities convertible into Common Stock solely for cash (other than offerings related to employee benefits plans or in connection with a merger or acquisition), WPM will have the right to acquire from Metavante Technologies for the same price and on the same terms as such securities are proposed to be offered to others, in the aggregate up to the amount of securities required to enable it to maintain its current percentage of ownership.

The Shareholders Agreement will terminate upon the earlier of: (i) its termination by the consent of all the parties to the Shareholders Agreement (with the consent of a majority of the independent directors of Metavante Technologies who are not WPM designees); (ii) the date on which WPM or its affiliates cease to hold any shares of registerable securities of Common Stock (except for those provisions that terminate as of a date specified in such provisions, which provisions shall terminate in accordance with the terms thereof); and (iii) the dissolution, liquidation or winding up of Metavante Technologies.

### Stock Purchase Right Agreement

As provided in the Investment Agreement, on November 1, 2007, Metavante Technologies and WPM entered into the Stock Purchase Right Agreement, giving WPM the right to purchase shares of Common Stock if employee stock options that were outstanding immediately following the Share Distribution are exercised after the Share Distribution. The Stock Purchase Right Agreement allows WPM to maintain its 25% interest in the Common Stock following the closing of the transactions notwithstanding the exercise of certain employee

CUSIP No. 591407101 13D Page 15 of	
	19

### options outstanding at the time of closing.

Pursuant to the terms of the Stock Purchase Right Agreement, subject to certain conditions, the total number of shares that may be purchased by WPM under the agreement will equal one-third of the aggregate number of shares of Common Stock that may be issued under specified Metavante Technologies employee options outstanding at the time of closing. Subject to the terms of the Stock Purchase Right Agreement, the stock purchase right may generally be exercised quarterly for one-third of the number of shares of Common Stock issued pursuant to such employee options during the preceding quarter, and the number of shares subject to the stock purchase right will be reduced by one-third of the number of shares subject to such employee options that expired unexercised in the preceding quarter. The Stock Purchase Right Agreement contains provisions allowing WPM to exercise a portion of its stock purchase rights in connection with certain transfers of Shares of Common Stock held by WPM.

### Additional Disclosure

Except as set forth above, none of the Reporting Persons nor, to the best of their knowledge, any person listed in Schedule I, has any plans or proposals which relate to or would result in: (a) the acquisition by any person of additional securities of Metavante Technologies, or the disposition of securities of Metavante Technologies; (b) an extraordinary corporate transaction, such as a merger, reorganization or liquidation, involving Metavante Technologies or any of its subsidiaries; (c) a sale or transfer of a material amount of assets of Metavante Technologies or any of its subsidiaries; (d) any change in the present board of directors or management of Metavante Technologies, including any plans or proposals to change the number or term of directors or to fill any existing vacancies on the board; (e) any material change in the present capitalization or dividend policy of Metavante Technologies; (f) any other material change in Metavante Technologies ' business or corporate structure; (g) any changes in Metavante Technologies' charter, bylaws or instruments corresponding thereto or other actions which may impede the acquisition of control of Metavante Technologies by any person; (h) causing a class of securities of Metavante Technologies to be delisted from a national securities exchange or to cease to be authorized to be quoted in an inter-dealer quotation system of a registered national securities association; (i) a class of equity securities of Metavante Technologies becoming eligible for termination of registration pursuant to Section 12(g)(4) of the Exchange Act; or (j) any action similar to any of those enumerated above.

#### Item 5. Interest in Securities of the Issuer

(a) WPM is the direct beneficial owner of 29,732,214 shares of Common Stock, representing approximately 25% of the outstanding shares of Common Stock (based on the number of shares of Common Stock outstanding as of October 31, 2007). Due to their respective relationships with WPM and each other, each of the Reporting Persons may be deemed to beneficially own, in the aggregate, 29,732,214 shares of Common Stock. Each of WPM GP, WP IX, WP IX LLC, WP Partners, WP, WP LLC, Messrs. Kaye and Landy and the individuals listed on Schedule I hereto disclaims beneficial ownership of the shares of Common Stock in which WPM has beneficial ownership, except to the extent of any indirect pecuniary interest therein. Except as described in this Item 5(a), no person listed in Item 2 of this Statement is a beneficial owner of the Common Stock in which WPM has beneficial ownership.

#### (b) See Item 5(a) above.

(c) On November 1, 2007, WPM acquired 29,732,214 shares of Class A Common Stock for an aggregate purchase price of \$625,000,000. At 12:01 a.m., Eastern Daylight Time, on November 2, 2007, each share of

13D



Class A Common Stock converted automatically into one share of Common Stock. Descriptions of the investment by WPM and of the Class A Common Stock and its conversion into Common Stock are included in the Prospectus and Item 4 hereto.

(d) Not applicable.

(e) Not applicable.

#### Item 6. Contracts, Arrangements, Understandings or Relationships with Respect to Securities of the Issuer

Pursuant to Rule 13d-1(k) promulgated under the Exchange Act, the Reporting Persons have entered into an agreement on November 7, 2007, with respect to the joint filing of this Statement and any amendment or amendments hereto (the "Joint Filing Agreement"). The Joint Filing Agreement is attached hereto as Exhibit 1 and incorporated herein by reference.

The responses set forth in Item 4 hereof are incorporated by reference in their entirety.

Except as referenced above or as described in Item 4 hereof, there are no contracts, arrangements, understandings or relationships among the persons named in Item 2 or between such persons and any other person with respect to any securities of Metavante Technologies.

### Item 7. Material To Be Filed as Exhibits

- Exhibit 1 Joint Filing Agreement, dated as of November 7, 2007, by and among WPM, L.P., WPM GP, LLC, Warburg Pincus Private Equity IX, L.P., Warburg Pincus IX LLC, Warburg Pincus Partners, LLC, Warburg Pincus & Co., Warburg Pincus LLC, Charles R. Kaye and Joseph P. Landy.
- Exhibit 2Investment Agreement, dated as of April 3, 2007, among M&I LLC (formerly known as Marshall & Ilsley Corporation), Metavante<br/>Corporation, Metavante Technologies, Inc. (formerly known as Metavante Holding Company), Montana Merger Sub Inc., and WPM, L.P.
- Exhibit 3 Shareholders Agreement, dated as of November 1, 2007, among Metavante Technologies, Inc. (formerly known as Metavante Holding Company) and WPM, L.P.
- Exhibit 4 Stock Purchase Right Agreement, dated as of November 1, 2007, among Metavante Technologies, Inc. (formerly known as Metavante Holding Company) and WPM, L.P.

13D

Page 17 of

19

### SIGNATURES

After reasonable inquiry and to the best of our knowledge and belief, the undersigned certify that the information set forth in this statement is true, complete and correct.

Dated November 7, 2007

#### WPM, L.P.

By: WPM GP, LLC, its general partner

By: /s/ Scott A. Arenare

 Name: Scott A. Arenare

 Title: Managing Director and Secretary

### WPM GP, LLC

By: /s/ Scott A. Arenare

Name: Scott A. Arenare Title: Managing Director and Secretary

### WARBURG PINCUS PRIVATE EQUITY IX, L.P.

- By: Warburg Pincus IX LLC, its general partner
- By: Warburg Pincus Partners, LLC, its general partner
- By: Warburg Pincus & Co., its managing member

### By: /s/ Scott A. Arenare Name: Scott A. Arenare Title: Partner

### WARBURG PINCUS IX, LLC

- By: Warburg Pincus Partners, LLC, its general partner
- By: Warburg Pincus & Co., its managing member
- By: /s/ Scott A. Arenare

Name: Scott A. Arenare Title: Partner

CUSIP No.	591407101	13D	Page [	18	of	19
		WARBURG PINCUS & CO.				
		By: /s/ Scott A. Arenare				
		Name: Scott A. Arenare Title: Partner				
		WARBURG PINCUS PARTNERS, LLC				
		By: Warburg Pincus & Co., its managing member				
		By: /s/ Scott A. Arenare				
		Name: Scott A. Arenare Title: Partner				
		WARBURG PINCUS LLC				
		By: /s/ Scott A. Arenare				
		Name: Scott A. Arenare Title: Managing Director				
		CHARLES R. KAYE				
		By: /s/ Scott A. Arenare				
		Scott A. Arenare, Attorney-in-fact*				
		JOSEPH P. LANDY				
		By: /s/ Scott A. Arenare				
		Scott A. Arenare, Attorney-in-fact**				

<sup>\*</sup> Power of Attorney given by Mr. Kaye was previously filed with the SEC on March 2, 2006, as an exhibit to a Schedule 13D filed by Building Products, LLC with respect to Builders FirstSource, Inc.

<sup>\*\*</sup> Power of Attorney given by Mr. Landy was previously filed with the SEC on March 2, 2006, as an exhibit to a Schedule 13D filed by Building Products, LLC with respect to Builders FirstSource, Inc.

CUSIP No.	591407101	13D	Page	19	] of	19
		INDEX OF EXHIBITS				
Exhibit 1	00	as of November 7, 2007, by and among WPM, L.P., WPM Warburg Pincus Partners, LLC, Warburg Pincus & Co., Wa	, , 0			1 5 .
Exhibit 2	8	as of April 3, 2007, among M&I LLC (formerly known as l ologies, Inc. (formerly known as Metavante Holding Comp	5	1 /		
Exhibit 3	Shareholders Agreement, dated Company) and WPM, L.P.	d as of November 1, 2007, among Metavante Technologies	, Inc. (formerly kno	own as Meta	avante	Holding
Exhibit 4	Stock Purchase Right Agreeme Holding Company) and WPM,	ent, dated as of November 1, 2007, among Metavante Tech , L.P.	nologies, Inc. (form	ierly knowi	1 as M	etavante

Set forth below are the name, position and present principal occupation of each of the general partners of Warburg Pincus & Co. ("WP"), members of Warburg Pincus LLC (including its subsidiaries, "WP LLC") and executive officers of WPM GP, LLC ("WPM GP"). WPM GP, WP and WP LLC are hereinafter collectively referred to as the "Reporting Entities". Except as otherwise indicated, the business address of each of such persons is 466 Lexington Avenue, New York, New York 10017, and each of such persons is a citizen of the United States.

### **GENERAL PARTNERS OF WP**

NAME	PRESENT PRINCIPAL OCCUPATION IN ADDITION TO POSITION WITH WP, AND POSITIONS
Joel Ackerman	WITH THE REPORTING ENTITIES Partner of WP; Member and Managing Director of WP LLC
Scott A. Arenare	Partner of WP; Member and Managing Director of WP LLC; Managing Director and Secretary of
Scott II. I Hentite	WPM GP
David Barr	Partner of WP; Member and Managing Director of WP LLC
Sean D. Carney	Partner of WP; Member and Managing Director of WP LLC
Mark Colodny	Partner of WP; Member and Managing Director of WP LLC
David A. Coulter	Partner of WP; Member and Managing Director of WP LLC
Timothy J. Curt	Partner of WP; Member and Managing Director of WP LLC; Managing Director and Treasurer of
·	WPM GP
W. Bowman Cutter	Partner of WP; Member and Managing Director of WP LLC
Cary J. Davis	Partner of WP; Member and Managing Director of WP LLC
David W. Dorman	Partner of WP; Member and Senior Advisor of WP LLC
Steven Glenn	Partner of WP; Member and Managing Director of WP LLC
Michael Graff	Partner of WP; Member and Managing Director of WP LLC
Patrick T. Hackett	Partner of WP; Member and Managing Director of WP LLC
E. Davisson Hardman	Partner of WP; Member and Managing Director of WP LLC
Jeffrey A. Harris	Partner of WP; Member and Managing Director of WP LLC
Stewart J. Hen	Partner of WP; Member and Managing Director of WP LLC
William H. Janeway	Partner of WP; Member and Senior Advisor of WP LLC
Julie A. Johnson Staples	Partner of WP; Member and Managing Director of WP LLC
Chansoo Joung	Partner of WP; Member and Managing Director of WP LLC
Peter R. Kagan	Partner of WP; Member and Managing Director of WP LLC
Charles R. Kaye	Managing General Partner of WP; Managing Member and Co-President of WP LLC
Henry Kressel	Partner of WP; Member and Managing Director of WP LLC
David Krieger	Partner of WP; Member and Managing Director of WP LLC
Kevin Kruse	Partner of WP; Member and Managing Director of WP LLC
Joseph P. Landy	Managing General Partner of WP; Managing Member and Co-President of WP LLC
Sidney Lapidus	Partner of WP; Member and Senior Advisor of WP LLC
Kewsong Lee	Partner of WP; Member and Managing Director of WP LLC
Jonathan S. Leff	Partner of WP; Member and Managing Director of WP LLC

NAME	PRESENT PRINCIPAL OCCUPATION IN ADDITION TO POSITION WITH WP, AND POSITIONS WITH THE REPORTING ENTITIES
Philip Mintz	Partner of WP; Member and Managing Director of WP LLC
James Neary	Partner of WP; Member and Managing Director of WP LLC; Managing Director of WPM GP
Bilge Ogut	Partner of WP; Member and Managing Director of WP LLC
Dalip Pathak	Partner of WP; Member and Managing Director of WP LLC
Michael F. Profenius	Partner of WP; Member and Managing Director of WP LLC
Stan Raatz	Partner of WP; Member and Managing Director of WP LLC
Justin Sadrian	Partner of WP; Member and Managing Director of WP LLC
Henry B. Schacht	Partner of WP; Member and Senior Advisor of WP LLC
Steven G. Schneider	Partner of WP; Member and Managing Director of WP LLC
John Shearburn	Partner of WP; Member and Managing Director of WP LLC
Mimi Strouse	Partner of WP; Member and Managing Director of WP LLC
Barry Taylor	Partner of WP; Member and Managing Director of WP LLC
Christopher H. Turner	Partner of WP; Member and Managing Director of WP LLC
John L. Vogelstein	Partner of WP; Member and Senior Advisor of WP LLC
Elizabeth H. Weatherman	Partner of WP; Member and Managing Director of WP LLC
David J. Wenstrup	Partner of WP; Member and Managing Director of WP LLC
Rosanne Zimmerman	Partner of WP; Member and Managing Director of WP LLC
Pincus & Company LLC*	
WP & Co. Partners, L.P.**	
Warburg Pincus Principal Partnership,	
L.P.***	
Warburg Pincus Real Estate Principal	
Partnership, L.P.***	
Warburg Pincus 2006	
Limited Partnership***	
-	

\* New York limited liability company; primary activity is ownership interest in WP and WP LLC

\*\* New York limited partnership; primary activity is ownership interest in WP

\*\*\* Delaware limited partnership; primary activity is ownership interest in WP

#### PRESENT PRINCIPAL OCCUPATION IN ADDITION TO POSITION WITH WP LLC, AND POSITIONS NAME WITH THE REPORTING ENTITIES Joel Ackerman Member and Managing Director of WP LLC; Partner of WP Member and Managing Director of WP LLC; Partner of WP; Managing Director and Secretary of Scott A. Arenare WPM GP Member and Managing Director of WP LLC; Partner of WP David Barr Sean D. Carney Member and Managing Director of WP LLC; Partner of WP Member and Managing Director of WP LLC Julian Cheng (1) Stephen John Coates (2) Member and Managing Director of WP LLC Member and Managing Director of WP LLC; Partner of WP Mark Colodny David A. Coulter Member and Managing Director of WP LLC; Partner of WP Timothy J. Curt Member and Managing Director of WP LLC; Partner of WP; Managing Director and Treasurer of WPM GP W. Bowman Cutter Member and Managing Director of WP LLC; Partner of WP Cary J. Davis Member and Managing Director of WP LLC; Partner of WP David W. Dorman Member and Senior Advisor of WP LLC: Partner of WP Rajiv Ghatalia (1) Member and Managing Director of WP LLC Steven Glenn Member and Managing Director of WP LLC; Partner of WP Michael Graff Member and Managing Director of WP LLC; Partner of WP Patrick T. Hackett Member and Managing Director of WP LLC; Partner of WP E. Davisson Hardman Member and Managing Director of WP LLC; Partner of WP Jeffrev A. Harris Member and Managing Director of WP LLC; Partner of WP Stewart J. Hen Member and Managing Director of WP LLC; Partner of WP William H. Janeway Member and Senior Advisor of WP LLC; Partner of WP Julie A. Johnson Staples Member and Managing Director of WP LLC; Partner of WP Member and Managing Director of WP LLC; Partner of WP Chansoo Joung Member and Managing Director of WP LLC; Partner of WP Peter R. Kagan Charles R. Kave Managing Member and Co-President of WP LLC; Managing General Partner of WP Rajesh Khanna (3) Member and Managing Director of WP LLC Member and Managing Director of WP LLC; Partner of WP Henry Kressel David Krieger Member and Managing Director of WP LLC; Partner of WP Kevin Kruse Member and Managing Director of WP LLC; Partner of WP Joseph P. Landy Managing Member and Co-President of WP LLC; Managing General Partner of WP Sidney Lapidus Member and Senior Advisor of WP LLC; Partner of WP Kewsong Lee Member and Managing Director of WP LLC; Partner of WP Jonathan S. Leff Member and Managing Director of WP LLC; Partner of WP David Li (1) Member and Managing Director of WP LLC Nicholas J. Lowcock (2) Member and Managing Director of WP LLC Niten Malhan (3) Member and Managing Director of WP LLC Philip Mintz Member and Managing Director of WP LLC; Partner of WP Member and Managing Director of WP LLC; Partner of WP; Managing Director of WPM GP James Neary Bilge Ogut Member and Managing Director of WP LLC; Partner of WP

### MEMBERS OF WP LLC

NAME
Dalip Pathak
Michael F. Profenius
Leo Puri (3)
Stan Raatz
Justin Sadrian
Henry B. Schacht
Steven G. Schneider
Joseph C. Schull (4)
John Shearburn
Mimi Strouse
Chang Q. Sun (1)
Barry Taylor
Christopher H. Turner
Simon Turton (2)
John L. Vogelstein
Elizabeth H. Weatherman
David J. Wenstrup
Peter Wilson (2)
Jeremy S. Young (2)
Rosanne Zimmerman
Pincus & Company LLC*

TO POSITION WITH WP LLC, AND POSITIONS WITH THE REPORTING ENTITIES Member and Managing Director of WP LLC; Partner of WP Member and Managing Director of WP LLC; Partner of WP Member and Managing Director of WP LLC Member and Managing Director of WP LLC; Partner of WP Member and Managing Director of WP LLC; Partner of WP Member and Senior Advisor of WP LLC; Partner of WP Member and Managing Director of WP LLC; Partner of WP Member and Managing Director of WP LLC Member and Managing Director of WP LLC; Partner of WP Member and Managing Director of WP LLC; Partner of WP Member and Managing Director of WP LLC Member and Managing Director of WP LLC; Partner of WP Member and Managing Director of WP LLC; Partner of WP Member and Managing Director of WP LLC Member and Senior Advisor of WP LLC; Partner of WP Member and Managing Director of WP LLC; Partner of WP Member and Managing Director of WP LLC; Partner of WP Member and Managing Director of WP LLC Member and Managing Director of WP LLC Member and Managing Director of WP LLC; Partner of WP

PRESENT PRINCIPAL OCCUPATION IN ADDITION

(1) Citizen of Hong Kong

- (2) Citizen of United Kingdom
- (3) Citizen of India

(4) Citizen of Canada

\* New York limited liability company; primary activity is ownership interest in WP and WP LLC

### **EXECUTIVE OFFICERS OF WPM GP**

### PRESENT PRINCIPAL OCCUPATION IN ADDITION TO POSITION WITH WPM GP, AND POSITIONS

NAME	WITH THE REPORTING ENTITIES
Scott A. Arenare	Managing Director and Secretary of WPM GP; Member and Managing Director of WP LLC; Partner of WP
Timothy J. Curt	Managing Director and Treasurer of WPM GP; Member and Managing Director of WP LLC; Partner of WP
James Neary	Managing Director of WPM GP; Member and Managing Director of WP LLC; Partner of WP
As of November 1, 2007	

### JOINT FILING AGREEMENT

Pursuant to Rule 13d-1(k) promulgated under the Securities Exchange Act of 1934, as amended, the undersigned hereby agree to the joint filing of this Statement on Schedule 13D including any amendments thereto. This Joint Filing Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all of which shall constitute one and the same instrument.

Date: November 7, 2007

WPM, L.P.

By: WPM GP, LLC, its general partner

By: /s/ Scott A. Arenare Name: Scott A. Arenare

Title: Managing Director and Secretary

WPM GP, LLC

By: /s/ Scott A. Arenare

Name: Scott A. Arenare Title: Managing Director and Secretary

WARBURG PINCUS PRIVATE EQUITY IX, L.P.

- By: Warburg Pincus IX LLC, its general partner
- By: Warburg Pincus Partners, LLC, its general partner
- By: Warburg Pincus & Co., its managing member

By: /s/ Scott A. Arenare

Name: Scott A. Arenare Title: Partner

WARBURG PINCUS IX, LLC

By: Warburg Pincus Partners, LLC, its general partner

By: Warburg Pincus & Co., its managing member

### By: /s/ Scott A. Arenare

Name: Scott A. Arenare Title: Partner

WARBURG PINCUS & CO.

By: /s/ Scott A. Arenare

Name: Scott A. Arenare Title: Partner

WARBURG PINCUS PARTNERS, LLC

By: Warburg Pincus & Co., its managing member

By: /s/ Scott A. Arenare Name: Scott A. Arenare Title: Partner

### WARBURG PINCUS LLC

By: /s/ Scott A. Arenare Name: Scott A. Arenare Title: Managing Director

### CHARLES R. KAYE

By: /s/ Scott A. Arenare Scott A. Arenare, Attorney-in-fact

JOSEPH P. LANDY

By: /s/ Scott A. Arenare

Scott A. Arenare, Attorney-in-fact

INVESTMENT AGREEMENT DATED AS OF APRIL 3, 2007 AMONG MARSHALL & ILSLEY CORPORATION, METAVANTE CORPORATION, METAVANTE HOLDING COMPANY, MONTANA MERGER SUB INC. AND WPM, L.P.

	Page
ARTICLE I DEFINITIONS	2
Section 1.1 Definitions	2
ARTICLE II SHARE PURCHASE	10
Section 2.1 Share Purchase	10
Section 2.2 Payment Terms of the Share Purchase	10
Section 2.3 Investor Share Number	10
ARTICLE III MERGER, OTHER TRANSACTIONS, EXCHANGE OF SHARES, ETC.	10
Section 3.1 Merger and Other Transactions	10
Section 3.2 Closing	12
Section 3.3 Effective Time of the MI Merger and the MI Conversion	12
Section 3.4 Effects of the MI Merger and the MI Conversion	12
Section 3.5 Organizational Documents of MI Corp., MVT Holding and MI LLC	12
Section 3.6 Effect on Capital Stock	13
Section 3.7 Exchange of Shares	14
Section 3.8 Directors and Officers of MVT Holding and MI Corp.	15
Section 3.9 Other Transaction Matters	15
Section 3.10 Options; Restricted Stock	15
Section 3.11 MVT Holding Class A Common Stock	16
Section 3.12 Initiation of the Closing Transactions	16
ARTICLE IV REPRESENTATIONS AND WARRANTIES	16
Section 4.1 Representations and Warranties of Investor	16
Section 4.2 Representations and Warranties of MI Corp	20
Section 4.3 Representations and Warranties of MI Corp., MVT Holding and MVT Corp.	27
Section 4.4 Representations and Warranties of MI Corp. and MVT Holding	38
ARTICLE V COVENANTS RELATING TO CONDUCT OF BUSINESS	41
Section 5.1 Covenants of MI Corp. and MVT Corp.	41
Section 5.2 Control of Other Party's Business	45
Section 5.3 Transfer of Investor Interests	45
ARTICLE VI ADDITIONAL AGREEMENTS	46
Section 6.1 Preparation of Form S-4, Proxy Statement and Form 10; Shareholders Meeting	46
Section 6.2 Governance Matters	47
Section 6.3 Access to Information	48
Section 6.4 Reasonable Best Efforts	49
Section 6.5 MI Acquisition Proposal; Change in Recommendation	51

## Table of Contents (continued)

	Page
Section 6.6 Fees and Expenses	55
Section 6.7 Public Announcements	56
Section 6.8 Takeover Statutes	56
Section 6.9 Advice of Changes	56
Section 6.10 Private Letter Ruling; Tax-Free Reorganization Treatment; Pre-Distribution Tax Returns	56
Section 6.11 Obligations under Separation Agreement	57
Section 6.12 Employee Benefits Matters	57
Section 6.13 Debt Financing	57
Section 6.14 Shareholders Agreement; Stock Purchase Right Agreement; Continuing Business Agreements	58
Section 6.15 Listing	59
Section 6.16 Investor Activity	59
Section 6.17 Valuation Firm	59
Section 6.18 Merger Sub Activity	60
Section 6.19 Sole Shareholder Approval	60
Section 6.20 Affiliates	60
Section 6.21 Non-Competition; Non-Solicitation	60
Section 6.22 Transaction Agreements	61
Section of 2 minuterior regreenents	01
ARTICLE VII CONDITIONS PRECEDENT	62
Section 7.1 Conditions to Each Party's Obligation to Effect the Transactions	62
Section 7.2 Additional Conditions to Obligations of Investor	64
Section 7.3 Additional Conditions to Obligations of MI Corp., MVT Corp. and MVT Holding	65
ARTICLE VIII TERMINATION AND AMENDMENT	66
Section 8.1 Termination	66
Section 8.2 Effect of Termination	67
ARTICLE IX GENERAL PROVISIONS	69
Section 9.1 Non-Survival of Representations and Warranties	69
Section 9.2 Notices	69
Section 9.3 Amendment	70
Section 9.4 Extension; Waiver	71
Section 9.5 Interpretation	71
Section 9.6 Counterparts	71
Section 9.7 Entire Agreement; No Third Party Beneficiaries	71
Section 9.8 Governing Law	71
Section 9.9 Severability	71
Section 9.10 Assignment	72
Section 9.11 Submission to Jurisdiction; Waivers	72
Section 9.12 Enforcement	73
	,5

# ii

Section 9.13 Disclosure Schedule Section 9.14 Mutual Drafting

### EXHIBITS

Exhibit A -Separation Agreement Exhibit B -**Employee Matters Agreement** Tax Allocation Agreement Exhibit C -Exhibit D -Form of Amended and Restated Articles of Incorporation of MVT Holding Exhibit E -Form of Amended and Restated By-laws of MVT Holding Exhibit F -Form of Shareholders Agreement Investor Share Number Calculation Exhibit G -Form of Stock Purchase Right Agreement Exhibit H -

Page 73 73

### **INVESTMENT AGREEMENT**

INVESTMENT AGREEMENT, dated as of April 3, 2007 (this "<u>Agreement</u>"), among MARSHALL & ILSLEY CORPORATION, a Wisconsin corporation ("<u>MI Corp.</u>"), METAVANTE CORPORATION, a Wisconsin corporation and, as of the date hereof, a wholly-owned subsidiary of MI Corp. ("<u>MVT Corp.</u>"), METAVANTE HOLDING COMPANY, a Wisconsin corporation and, as of the date hereof, a wholly-owned subsidiary of MI Corp. ("<u>MVT Holding</u>"), MONTANA MERGER SUB INC., a Wisconsin corporation and, as of the date hereof, a wholly-owned subsidiary of MVT Holding ("<u>Merger Sub</u>"), and WPM, L.P., a Delaware limited partnership ("<u>Investor</u>" and, collectively with MI Corp., MVT Corp., MVT Holding and Merger Sub, the "<u>Parties</u>").

### WITNESSETH:

WHEREAS, the Board of Directors of MI Corp. deems it advisable and in the best interests of MI Corp. and its shareholders that the following transactions be consummated subject to the terms and conditions hereof:

(a) the merger (the "<u>MI Merger</u>") of Merger Sub with and into MI Corp., whereby each share of common stock, \$1.00 par value per share, of MI Corp. (the "<u>MI Common Stock</u>"), shall be converted into one-third of a share of common stock, \$0.01 par value per share, of MVT Holding (the "<u>MVT Holding Common Stock</u>"), MI Corp. shall be the surviving corporation and MI Corp. shall become a wholly-owned subsidiary of MVT Holding;

(b) immediately after the effective time of the MI Merger, the conversion (the "<u>MI Conversion</u>") of MI Corp. into a limited liability company organized under the laws of the State of Wisconsin ("<u>MI LLC</u>"), whereby the outstanding shares of MI Common Stock shall be converted into membership interests in MI LLC;

(c) the distribution by MI LLC of all of the outstanding shares of common stock, \$0.01 par value per share, of MVT Corp. ("<u>MVT Common Stock</u>") to MVT Holding (the "<u>MVT Distribution</u>"), whereby MVT Corp. shall become a wholly-owned subsidiary of MVT Holding;

(d) the (i) issuance and sale by MVT Holding to Investor of a number of shares of Class A Common Stock, \$0.01 par value per share, of MVT Holding (the "<u>MVT Holding Class A Common Stock</u>") equal to the Investor Share Number (the "<u>Share Issuance</u>"), (ii) the consummation of the Debt Financing by MVT Holding and/or one or more of its wholly-owned Subsidiaries and (iii) the payment of the Intercompany Debt Amount to MI LLC;

(e) the payment by MVT Corp. to MVT Holding of a dividend in the amount of \$1,040,000,000 in cash (the "MVT Dividend");

(f) the contribution by MVT Holding to New M&I Corporation, a Wisconsin corporation and, as of the date hereof, a wholly-owned subsidiary of MVT

Holding ("<u>New MI Corp.</u>"), of (i) all the membership interests of MI LLC (the "<u>MI LLC Contribution</u>") and (ii) an amount in cash equal to the MI Cash Contribution (the MI Cash Contribution together with the MI LLC Contribution, the "<u>MI Contribution</u>"); and

(g) the distribution (the "<u>Share Distribution</u>") of all of the issued and outstanding shares of common stock, par value \$0.01 per share of New MI Corp. ("<u>New MI Corp. Common Stock</u>"), on a pro rata basis to the holders of record of MVT Holding Common Stock (based upon a distribution ratio of three shares of New MI Corp. Common Stock for every one Share of MVT Holding Common Stock) upon the terms and subject to the conditions set forth in the Separation Agreement, dated as of the date hereof, among MI Corp., MVT Holding, MVT Corp. and New MI Corp. and attached hereto as <u>Exhibit A</u> (the "<u>Separation Agreement</u>") (the transactions described in <u>clauses (a)</u> through (g) of this recital are referred to as the "<u>Closing Transactions</u>," and together with the other transactions contemplated hereby, the "<u>Transactions</u>");

WHEREAS, for United States federal income tax purposes, it is intended that (a) the MI Merger and the MI Conversion shall qualify as a reorganization under Section 368(a)(1)(F) of the Internal Revenue Code of 1986, as amended (the "<u>Code</u>"), and that this Agreement is intended to be, and is hereby adopted as, a plan of reorganization and (b) the MI Contribution and the Share Distribution shall qualify as a reorganization under Section 368(a)(1) (D) of the Code and a distribution eligible for nonrecognition under Sections 355(a) and 361(c) of the Code;

WHEREAS, MI Corp., MVT Corp., MVT Holding and Investor each desire to make certain representations, warranties, covenants and agreements in connection with the Transactions and to prescribe the various conditions to the Transactions.

NOW, THEREFORE, in consideration of the foregoing and the respective representations, warranties, covenants and agreements set forth in this Agreement, and intending to be legally bound hereby, the parties hereto agree as follows:

### ARTICLE I

### **DEFINITIONS**

Section 1.1 <u>Definitions</u>. Capitalized terms used in this Agreement have the meanings set forth in this Agreement or, when so indicated, in the applicable Transaction Agreement. As used in this Agreement:

(a) "<u>Additional Required Approvals</u>" means (i) approval of the MI Merger by MVT Holding, as the sole shareholder of Merger Sub, (ii) approval of the MI Conversion by the Board of Directors of MI Corp. and by MVT Holding, as the sole shareholder of MI Corp., (iii) approval of the MVT Distribution by the Board of Directors (or similar body) of MI LLC, (iv) approval of the MVT Dividend by the Board of Directors of MVT Corp., (v) approval of the MI LLC Contribution by the Board of Directors of MVT Holding, (vi) approval of the MI Cash Contribution by the

Board of Directors of MVT Holding, and (vii) approval of the Share Distribution by the Board of Directors of MVT Holding.

(b) "<u>Administrative Services Agreement</u>" means the Administrative Services Agreement, dated as of the date hereof, among MVT Holding, MVT Corp., New MI Corp. and MI Corp., in substantially the form provided to Investor.

(c) "<u>Affiliate</u>" means (except as specifically otherwise defined), with respect to any specified Person, any other Person that, directly or indirectly, controls, or is controlled by, or is under common control with, such specified Person. As used in this definition, "control" (including, with its correlative meanings, "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ownership of voting securities or partnership or other ownership interests, by contract or otherwise.

(d) "<u>Applicable Laws</u>" means all applicable laws, statutes, ordinances, orders, decrees, rules, regulations, policies or guidelines promulgated, or judgments, decisions, orders or arbitration awards entered, by any Governmental Entity.

(e) "Board of Directors" means the board of directors of any specified Person.

(f) "Business Day" means any day on which banks are not required or authorized to close in the City of New York.

(g) "<u>Computer Software</u>" means (i) computer programs, including any and all software implementations of algorithms, models and methodologies, whether in source code, object code, executable or binary code (ii) databases and compilations, including any and all data and collections of data, whether machine readable or otherwise, (iii) descriptions, flow-charts and other work product used to design, plan, organize, maintain, support or develop any of the foregoing, and (iv) all documentation, including programmers' notes and source code annotations, user manuals and training materials relating to any of the foregoing, including any translations thereof.

(h) "<u>Continuing Business Agreements</u>" means the Contracts set forth in <u>Section 1.1</u> of the MI Disclosure Schedule among certain members of the MI Group, on the one hand, and certain members of the MVT Group, on the other hand, which are being entered into on the date hereof in substantially the form provided to Investor (other than those Contracts identified in <u>Section 1.1</u> of the MI Disclosure Schedule which are to be entered into prior to Closing Date with the consent of Investor (such consent not to be unreasonably withheld)).

(i) "<u>Employee Matters Agreement</u>" means the Employee Matters Agreement, dated as of the date hereof, among MVT Holding, MVT Corp., New MI Corp. and MI Corp., in the form attached hereto as <u>Exhibit B</u>.

(j) "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

(k) "ERISA Affiliate" when used with respect to any Person, means any trade or business, whether or not incorporated, that, together with such Person, would be deemed to be a "single employer" within the meaning of Section 4001(b) of ERISA.

(1) "Form 10" means the registration statement with respect to the shares of New MI Corp. Common Stock to be distributed in the Share Distribution.

(m) "GAAP" means United States generally accepted accounting principles, consistently applied.

(n) "<u>Intercompany Debt Amount</u>" means the aggregate principal amount of indebtedness for borrowed money owed to MI Corp. by MVT Corp. pursuant to the MVT Notes, plus, in each case, accrued and unpaid interest thereon as of the Closing Date.

(o) "<u>Investor Tax Affiliate</u>" means any entity or individual (i) whose ownership of stock would be attributable to or aggregated with Investor under Section 355(e)(4)(C) of the Code, (ii) who is a member of any "coordinating group" (within the meaning of Treasury Regulation Section 1.355-7(h)(4)) that includes Investor, or (iii) who is acting pursuant to a "plan or arrangement" (within the meaning of Section 355(d)(7)(B) of the Code) with Investor.

(p) "<u>Known</u>" or "<u>Knowledge</u>" means, (i) with respect to MVT Holding, MI Corp. or MVT Corp., the actual knowledge, after reasonable inquiry, of any of the persons set forth in <u>Section 1.1</u> of the MI Disclosure Schedule and (ii) with respect to Investor, the actual knowledge of any of the following persons: James Neary, Adarsh Sarma or Alex Berzofsky.

(q) "Limited Guarantee" means the Limited Guarantee, dated as of the date hereof, by the Equity Fund as guarantor of Investor in favor of MI Corp. and MVT Corp. as the guaranteed parties.

(r) "<u>Material Adverse Effect</u>" means any effect, change, circumstance or development that, individually or in the aggregate with other such effects, changes, circumstances or developments, is both material and adverse to, (i) with respect to any Person, the ability of such Person (or, in the case of MI Corp. or MVT Corp., any member of either of the MI Group or the MVT Group) to consummate the Transactions or (ii) with respect to MVT Corp., the business, financial condition, operations, results of operations, properties, assets or liabilities of MVT Corp. and its Subsidiaries (or with respect to the businesses engaged in prior to the MVT Distribution Time by the MVT Group that constitute MI Corp.'s Metavante Corporation segment for reporting purposes as listed in the latest Annual Report on Form 10-K of MI Corp. included in the MI Corp. Filed SEC Reports) taken as a whole, other than, in the case of this clause (ii), any effect, change, circumstance or development (A) resulting from the

announcement of the execution of this Agreement or any of the Transaction Agreements or of the Transactions, (B) relating to state, national or international political, social, general business or economic conditions (but only if such Person and its Subsidiaries, taken as a whole, are not disproportionately affected thereby in any material respect as compared to other comparable companies in their industry), (C) relating in general to the industries in which such Person and its Subsidiaries operate (but only if such Person and its Subsidiaries, taken as a whole, are not disproportionately affected thereby in any material respect as compared to other comparable companies in their industry), (D) relating to any action of MVT Holding, MI Corp., MVT Corp. or any Subsidiary of any of them taken with the express prior written consent of Investor after the date hereof, (E) relating to the commencement, occurrence or continuation of any war, armed hostilities or acts of terrorism involving or affecting the United States of America or any other jurisdiction in which such Person or any of its Subsidiaries operates (but only if such Person and its Subsidiaries, taken as a whole, are not disproportionately affected thereby in any material respect as compared to other comparable companies in their industry), (F) relating to financial, banking, or securities markets (including any disruption thereof and any decline in the price of any security or any market index), (G) relating to changes after the date hereof in United States GAAP or the accounting rules and regulations of the SEC (but only if MVT Corp. and its Subsidiaries, taken as a whole, are not disproportionately affected thereby in any material respect as compared to other comparable companies in their industry) or (H) relating to changes in Applicable Laws.

(s) "<u>MI Cash Contribution</u>" means an amount in cash equal to \$1,665,000,000.

(t) "MI Group" means New MI Corp., MI Corp. and each of the Subsidiaries of MI Corp. (after giving effect to the MVT Distribution).

(u) "Multiemployer Plan" means any "multiemployer plan" within the meaning of Section 4001(a)(3) of ERISA.

(v) "<u>MVT Distribution Time</u>" means the time at which the distribution of all outstanding MVT Common Stock to MVT Holding is effective as determined by the MI Corp. Board of Directors.

(w) "MVT Group," means MVT Holding, MVT Corp. and each Subsidiary of MVT Corp. or MVT Holding other than a member of the MI Group.

(x) "<u>MVT Notes</u>" means the (A) Unsecured Note, dated April 1, 2003, in the aggregate principal amount of \$67,000,000, issued by MVT Corp. to MI Corp., (B) Unsecured Note, dated May 27, 2004, in the aggregate principal amount of \$165,000,000, issued by MVT Corp. to MI Corp., (C) Unsecured Note, dated July 1, 2004, in the aggregate principal amount of \$140,000,000, issued by MVT Corp. to MI Corp. and (D) Unsecured Note, dated July 30, 2004, in the aggregate principal amount of \$610,000,000, issued by MVT Corp. to MI Corp.

(y) "<u>MVT Plan</u>" means any material employee benefit plan, program, policy, practice or other arrangement providing compensation or benefits to any current or former MVT Employee or any beneficiary or dependent thereof that is sponsored, maintained or contributed to by MI Corp. or MVT Corp. or any of its other ERISA Affiliates or other Subsidiaries or to which MI Corp. or any of its ERISA Affiliates or other Subsidiaries or is obligated to contribute, whether or not written, including any material employee benefit plan within the meaning of Section 3(3) of ERISA (whether or not such plan is subject to ERISA) and any material bonus, incentive, deferred compensation, welfare, paid time off, bereavement, tuition, employee assistance, relocation, vacation, stock purchase, stock option, equity or equity-based compensation, severance, termination, employment, change of control or fringe benefit plan, program or agreement.

(z) "<u>NYSE</u>" means The New York Stock Exchange, Inc.

(aa) "<u>Person</u>" means an individual, corporation, limited liability company, partnership, association, joint venture, trust, unincorporated organization, other entity or group (as defined in the Exchange Act), including any Governmental Entity.

(bb) "SEC" means the Securities and Exchange Commission.

(cc) "<u>Stock Purchase Right Agreement</u>" means the Stock Purchase Right Agreement between MVT Holding and Investor in the form attached as <u>Exhibit H</u>.

(dd) "<u>Subsidiary</u>" means, when used with respect to any Person, any corporation or other organization, whether incorporated or unincorporated, at least a majority of the securities or other interests of which having by their terms ordinary voting power to elect a majority of the Board of Directors or others performing similar functions with respect to such corporation or other organization is directly or indirectly owned or controlled by such Person or by any one or more of its Subsidiaries, or by such Person and one or more of its Subsidiaries.

(ee) "Successor Fund" means one or more successor funds to the Investor Fund, each of which is controlled (as defined in Section <u>1.1(c)</u>) by Warburg Pincus LLC and/or Warburg Pincus & Co. (or a controlled Affiliate of one of such entities) and is managed by Warburg Pincus LLC or its Affiliates.

(ff) "Tax" (and, with correlative meaning, "Taxes" and "Taxable") means any tax, in any sense, including (i) any federal, state, municipal, county, local, foreign or other Governmental Entity net income, gross income, receipts, windfall profit, severance, real, personal, tangible, escheatable, unclaimed or abandoned property, goods and services, value added, estimated, capital stock, production, sales, use, license, excise, franchise, employment, unemployment, social security, payroll, withholding, alternative or add-on minimum, ad valorem, transfer, stamp, or environmental tax, or any other tax, customs, duty or other like assessment or charge of any kind whatsoever, together with any interest or penalty, addition to tax or additional

amount imposed by any Governmental Entity, including any fines, penalties or interest arising under ERISA; and (ii) any liability for payments of a type described in <u>clause (i)</u> as a result of being or having been a member of an affiliated, consolidated, combined, unitary or similar group.

(gg) "<u>Tax Allocation Agreement</u>" means the Tax Allocation Agreement, dated as of the date hereof, among MVT Holding, MVT Corp., New MI Corp. and MI Corp., in the form attached hereto as <u>Exhibit C</u>.

(hh) "<u>Tax Return</u>" means any return, report or similar statement required to be filed with respect to any Taxes (including any attached schedules), including any information return, claim for refund, amended return or declaration of estimated Tax.

(ii) "<u>Transaction Agreements</u>" means collectively, the Separation Agreement, the Employee Matters Agreement, the Tax Allocation Agreement, the Shareholders Agreement, the Stock Purchase Right Agreement, the Limited Guarantee, the Administrative Services Agreement, the Continuing Business Agreements and the other agreements, if any, entered into or to be entered into in connection with the Transactions.

Each of the following terms is defined in the Section of this Agreement or the agreement set forth opposite such term:

Action	4.1(d)(i)
Agreement	Preamble
Alternative Debt Financing	6.13(b)
BHC Act	4.2(a)
Сар	4.2(d) 8.2(f)
Capitalization Date	4.2(b)
Change in the MI Recommendation	4.2(0) 6.5(c)
Closing	3.2
Closing Date	3.2
Closing Price	3.2 3.6(c)
	Recitals
Closing Transactions	Recitals
Code	
Confidentiality Agreement	6.3
Contract	4.1(b)(ii)
Conversion Effective Time	3.3
Customer Contracts	4.3(k)(i)
Debt Commitment Letter	6.13(a)
Debt Financing	6.13(a)
DOJ	6.4(b)
Environmental Laws	4.3(h)
Environmental Liabilities	4.3(h)
Equity Commitment Letter	4.1(f)
Equity Financing	4.1(f)
Equity Fund	4.1(f)
Exchange Act	4.1(b)(iii)
Exchange Agent	3.7(a)
Exchange Fund	3.7(a)
Executive Committee	6.21(b)
Expenses	6.6
Force the MI Vote Notice	6.1(b)
Form S-4	6.1(a)
FRB	6.4(b)
FTC	6.4(b)
Governmental Entity	4.1(b)(iii)
Hazardous Materials	4.3(h)
HSR Act	4.1(b)(iii)
Independent Director	6.2(a)
Injunction	7.1(b)
Intellectual Property	4.3(i)
Investor	Preamble
Investor Necessary Consents	4.1(b)(iii)
Investor Share Number	Exhibit G
IRS	4.2(h)
Liens	4.1(b)(ii)
Merger Sub	Preamble
7	

Merger Sub Common Stock	3.6(d)
MI Acquisition Agreement	6.5(a)
MI Acquisition Proposal	6.5(b)
MI Articles of Merger	3.3
MI Cash Contribution	Recitals
MI Certificate	3.6(b)
MI Certificate of Conversion	3.3
MI Certificate of Formation	3.3
MI Common Stock	Recitals
MI Contribution	Recitals
MI Conversion	Recitals
MI Corp.	Recitals
MI Corp. Sale Transaction	6.4(e)
MI Disclosure Schedule	4.2
MI Effective Time	4.2
MI Filed SEC Reports	4.3(d)(iii)
MILLC	Recitals
MI LLC Contribution	Recitals
MI Merger	Recitals
MI Necessary Consents	4.2(c)(iv)
MI Options	4.2(b)(i)
MI Option Plans	4.2(b)(i)
MI Permits	4.2(i)
MI Preferred Stock	4.2(b)(i)
MI Recommendation	6.1(b)
MI Restricted Stock	4.2(b)(i)
MI Restricted Stock Plans	4.2(b)(i)
MI SEC Reports	4.2(d)(i)
MI SPACES Plan	4.2(b)(i)
MI Stock Purchase Plan	4.2(b)(i)
MI Shareholders Meeting	6.1(b)
MI Transaction Approval	4.2(c)(i)
MI Vote	6.1(b)
MI Voting Debt	4.2(b)(ii)
MVT Acquisition Proposal	6.5(a)
MVT Business Personnel	4.3(m)
MVT Business	Separation Agreement
MVT CEO	6.2(a)
MVT CEO MVT Common Stock	6.2(a) Recitals
MVT Common Stock	Recitals
MVT Common Stock MVT Corp.	Recitals Preamble
MVT Common Stock MVT Corp. MVT Disclosure Schedule	Recitals Preamble 4.3
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution	Recitals Preamble 4.3 Recitals
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend	Recitals Preamble 4.3 Recitals Recitals
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution	Recitals Preamble 4.3 Recitals Recitals Employee Matters
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee	Recitals Preamble 4.3 Recitals Recitals Employee Matters Agreement
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements	Recitals Preamble 4.3 Recitals Recitals Employee Matters Agreement 4.3(d)(i)
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding	Recitals Preamble 4.3 Recitals Recitals Employee Matters Agreement 4.3(d)(i) Preamble
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock	Recitals Preamble 4.3 Recitals Recitals Employee Matters Agreement 4.3(d)(i) Preamble Recitals
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock	Recitals Preamble 4.3 Recitals Recitals Employee Matters Agreement 4.3(d)(i) Preamble Recitals Recitals
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents	Recitals Preamble 4.3 Recitals Recitals Employee Matters Agreement 4.3(d)(i) Preamble Recitals Recitals Actials 4.4(c)(iii)
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Prospectus	Recitals Preamble 4.3 Recitals Recitals Employee Matters Agreement 4.3(d)(i) Preamble Recitals Recitals Recitals 4.4(c)(iii) 6.1(a)
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Prospectus MVT Holding Voting Debt	Recitals Preamble 4.3 Recitals Recitals Employee Matters Agreement 4.3(d)(i) Preamble Recitals Recitals Recitals 4.4(c)(iii) 6.1(a) 4.4(b)(ii)
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Prospectus MVT Holding Voting Debt MVT Lease	Recitals Preamble 4.3 Recitals Recitals Employee Matters Agreement 4.3(d)(i) Preamble Recitals Recitals Recitals 4.4(c)(iii) 6.1(a) 4.4(b)(ii) 4.3(p)
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Prospectus MVT Holding Voting Debt MVT Lease MVT Lease MVT Material Contracts	$\begin{array}{c} \text{Recitals} \\ \text{Preamble} \\ 4.3 \\ \text{Recitals} \\ \text{Recitals} \\ \text{Employee Matters} \\ \text{Agreement} \\ 4.3(d)(i) \\ \text{Preamble} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Alt}(c)(iii) \\ 6.1(a) \\ 4.4(b)(ii) \\ 4.3(p) \\ 4.3(k)(i) \end{array}$
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Prospectus MVT Holding Voting Debt MVT Lease MVT Lease MVT Material Contracts MVT Necessary Consents	Recitals Preamble 4.3 Recitals Recitals Employee Matters Agreement 4.3(d)(i) Preamble Recitals Recitals Recitals 4.4(c)(iii) 6.1(a) 4.4(b)(ii) 4.3(p) 4.3(k)(i) 4.3(c)(iv)
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Voting Debt MVT Holding Voting Debt MVT Lease MVT Material Contracts MVT Necessary Consents MVT Option	Recitals Preamble 4.3 Recitals Recitals Employee Matters Agreement 4.3(d)(i) Preamble Recitals Recitals Recitals 4.4(c)(iii) 6.1(a) 4.4(b)(ii) 4.3(p) 4.3(k)(i) 4.3(c)(iv) 3.10(a)
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Voting Debt MVT Holding Voting Debt MVT Lease MVT Material Contracts MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents	$\begin{array}{c} \text{Recitals} \\ \text{Preamble} \\ 4.3 \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Employee Matters} \\ \text{Agreement} \\ 4.3(d)(i) \\ \text{Preamble} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ 4.4(c)(iii) \\ 6.1(a) \\ 4.4(b)(ii) \\ 4.3(p) \\ 4.3(k)(i) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(f)(ii) \end{array}$
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Prospectus MVT Holding Debt MVT Lease MVT Material Contracts MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Permits MVT Permits MVT Significant Subsidiary	$\begin{array}{c} \text{Recitals} \\ \text{Preamble} \\ 4.3 \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Employee Matters} \\ \text{Agreement} \\ 4.3(d)(i) \\ \text{Preamble} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ 4.4(c)(iii) \\ 6.1(a) \\ 4.4(b)(ii) \\ 4.3(p) \\ 4.3(k)(i) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(f)(ii) \\ 4.3(a)(ii) \\ \end{array}$
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Voting Debt MVT Material Contracts MVT Necessary Consents MVT Necessary Consents MVT Neterial Contracts MVT Permits MVT Permits MVT Significant Subsidiary MVT Voting Debt	$\begin{array}{c} \text{Recitals} \\ \text{Preamble} \\ 4.3 \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Employee Matters} \\ \text{Agreement} \\ 4.3(d)(i) \\ \text{Preamble} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ 4.4(c)(iii) \\ 6.1(a) \\ 4.4(b)(ii) \\ 4.3(p) \\ 4.3(k)(i) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(f)(ii) \end{array}$
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Prospectus MVT Holding Debt MVT Lease MVT Material Contracts MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Permits MVT Permits MVT Significant Subsidiary	Recitals           Preamble           4.3           Recitals           Recitals           Employee Matters           Agreement           4.3(d)(i)           Preamble           Recitals           Recitals           Recitals           Recitals           Agreement           4.3(d)(i)           Preamble           Recitals           Recitals           Alge           4.4(c)(iii)           6.1(a)           4.4(b)(ii)           4.3(p)           4.3(k)(i)           4.3(c)(iv)           3.10(a)           4.3(c)(ii)           4.3(a)(ii)           4.3(b)(ii)           4.3(c)(iv)
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Distribution MVT Employee MVT Financial Statements MVT Financial Statements MVT Holding Class A Common Stock MVT Holding Class A Common Stock MVT Holding Necessary Consents MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Voting Debt MVT Lease MVT Lease MVT Material Contracts MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Significant Subsidiary MVT Voting Debt NASD New MI Corp.	Recitals           Preamble           4.3           Recitals           Recitals           Employee Matters           Agreement           4.3(d)(i)           Preamble           Recitals           Recitals           Recitals           Recitals           Agreement           4.3(d)(i)           Preamble           Recitals           Recitals           Alters           4.4(c)(iii)           6.1(a)           4.4(b)(ii)           4.3(p)           4.3(k)(i)           4.3(c)(iv)           3.10(a)           4.3(c)(ii)           4.3(a)(ii)           4.3(a)(ii)
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Noting Debt MVT Holding Voting Debt MVT Material Contracts MVT Necessary Consents MVT	$\begin{array}{c} \text{Recitals} \\ \text{Preamble} \\ 4.3 \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Employee Matters} \\ \text{Agreement} \\ 4.3(d)(i) \\ \text{Preamble} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ 4.4(c)(iii) \\ 6.1(a) \\ 4.4(b)(ii) \\ 6.1(a) \\ 4.3(p) \\ 4.3(p) \\ 4.3(k)(i) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(b)(ii) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(b)(ii) \\ 4.3(c)(iv) \\ 8.citals \\ \text{Recitals} \\ \text{Recitals} \end{array}$
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Distribution MVT Employee MVT Financial Statements MVT Financial Statements MVT Holding Class A Common Stock MVT Holding Class A Common Stock MVT Holding Necessary Consents MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Voting Debt MVT Lease MVT Lease MVT Material Contracts MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Significant Subsidiary MVT Voting Debt NASD New MI Corp.	$\begin{array}{c} \text{Recitals} \\ \text{Preamble} \\ 4.3 \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Employee Matters} \\ \text{Agreement} \\ 4.3(d)(i) \\ \text{Preamble} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Act(c)(iii)} \\ 6.1(a) \\ 4.4(c)(iii) \\ 6.1(a) \\ 4.4(b)(ii) \\ 4.3(p) \\ 4.3(p) \\ 4.3(k)(i) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(c)(iv) \\ 5.10(a) \\ 5.10($
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Distribution MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Voting Debt MVT Lease MVT Material Contracts MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Significant Subsidiary MVT Voting Debt NASD New MI Corp. New MI Corp.	$\begin{array}{c} \text{Recitals} \\ \text{Preamble} \\ 4.3 \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Employee Matters} \\ \text{Agreement} \\ 4.3(d)(i) \\ \text{Preamble} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ 4.4(c)(iii) \\ 6.1(a) \\ 4.4(b)(ii) \\ 6.1(a) \\ 4.3(p) \\ 4.3(p) \\ 4.3(k)(i) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(b)(ii) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(b)(ii) \\ 4.3(c)(iv) \\ 8.citals \\ \text{Recitals} \\ \text{Recitals} \end{array}$
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Distribution MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Prospectus MVT Holding Voting Debt MVT Lease MVT Material Contracts MVT Necessary Consents MVT Net	$\begin{array}{c} \text{Recitals} \\ \text{Preamble} \\ 4.3 \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Employee Matters} \\ \text{Agreement} \\ 4.3(d)(i) \\ \text{Preamble} \\ \text{Recitals} \\ \text{Recitals} \\ 4.4(c)(iii) \\ 6.1(a) \\ 4.4(c)(iii) \\ 6.1(a) \\ 4.3(p) \\ 4.3(p) \\ 4.3(p) \\ 4.3(k)(i) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(b)(ii) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(b)(ii) \\ 4.2(c)(iv) \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ 3.10(b) \end{array}$
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Voting Debt MVT Holding Voting Debt MVT Atterial Contracts MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Premits MVT Significant Subsidiary MVT Voting Debt NASD New MI Corp. New MI Corp. Common Stock New MI Corp. Option Parties	$\begin{array}{c} \text{Recitals} \\ \text{Preamble} \\ 4.3 \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Employee Matters} \\ \text{Agreement} \\ 4.3(d)(i) \\ \text{Preamble} \\ \text{Recitals} \\ \text{Recitals} \\ 4.4(c)(iii) \\ 6.1(a) \\ 4.4(c)(iii) \\ 6.1(a) \\ 4.4(b)(ii) \\ 4.3(p) \\ 4.3(p) \\ 4.3(k)(i) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(b)(ii) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(b)(ii) \\ 4.2(c)(iv) \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ 3.10(b) \\ \text{Preamble} \end{array}$
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Class A Common Stock MVT Holding Recessary Consents MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Noting Debt MVT Lease MVT Aterial Contracts MVT Necessary Consents MVT Necessary Consents MVT Option MVT Permits MVT Significant Subsidiary MVT Voting Debt NASD New MI Corp. New MI Corp. Common Stock New MI Corp. Option Parties Private Letter Ruling	$\begin{array}{c} \text{Recitals} \\ \text{Preamble} \\ 4.3 \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ \text{Employee Matters} \\ \text{Agreement} \\ 4.3(d)(i) \\ \text{Preamble} \\ \text{Recitals} \\ \text{Recitals} \\ 4.4(c)(iii) \\ 6.1(a) \\ 4.4(c)(iii) \\ 6.1(a) \\ 4.4(b)(ii) \\ 4.3(p) \\ 4.3(p) \\ 4.3(k)(i) \\ 4.3(c)(iv) \\ 3.10(a) \\ 4.3(f)(ii) \\ 4.3(a)(ii) \\ 4.3(a)(ii) \\ 4.3(b)(ii) \\ 4.3(b)(ii) \\ 4.2(c)(iv) \\ \text{Recitals} \\ \text{Recitals} \\ \text{Recitals} \\ 3.10(b) \\ \text{Preamble} \\ 7.1(f)(i) \end{array}$
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Distribution MVT Employee MVT Financial Statements MVT Holding Class A Common Stock MVT Holding Class A Common Stock MVT Holding Class A Common Stock MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Voting Debt MVT Holding Voting Debt MVT Holding Contracts MVT Naterial Contracts MVT Necessary Consents MVT Necessary Consents MVT Significant Subsidiary MVT Significant Subsidiary MVT Significant Subsidiary MVT Voting Debt NASD New MI Corp. New MI Corp. New MI Corp. Common Stock New MI Corp. Common Stock New MI Corp. Option Parties Private Letter Ruling Proxy Statement Purchase Price	$\begin{array}{c} \mbox{Recitals} \\ \mbox{Preamble} \\ \mbox{4.3} \\ \mbox{Recitals} \\ \mbox{Recitals} \\ \mbox{Recitals} \\ \mbox{Employee Matters} \\ \mbox{Agreement} \\ \mbox{4.3(d)(i)} \\ \mbox{Preamble} \\ \mbox{Recitals} \\ \mbox{Recitals} \\ \mbox{4.4(c)(iii)} \\ \mbox{6.1(a)} \\ \mbox{4.4(b)(ii)} \\ \mbox{4.3(p)} \\ \mbox{4.3(k)(i)} \\ \mbox{4.3(c)(iv)} \\ \mbox{3.10(a)} \\ \mbox{4.3(c)(iv)} \\ \mbox{3.10(a)} \\ \mbox{4.3(b)(ii)} \\ \mbox{4.3(b)(ii)} \\ \mbox{4.3(b)(ii)} \\ \mbox{4.3(b)(ii)} \\ \mbox{4.3(c)(iv)} \\ \mbox{3.10(b)} \\ \mbox{Preamble} \\ \mbox{7.1(f)(i)} \\ \mbox{6.1(a)} \\ \mbox{2.1} \end{array}$
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Financial Statements MVT Holding Class A Common Stock MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Necessary Consents MVT Holding Voting Debt MVT Lease MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Significant Subsidiary MVT Voting Debt NASD New MI Corp. New MI Corp. Common Stock New MI Corp. Option Parties Private Letter Ruling Proxy Statement Purchase Price Required Approvals	Recitals           Preamble           4.3           Recitals           Recitals           Employee Matters           Agreement           4.3(d)(i)           Preamble           Recitals           Recitals           Recitals           Recitals           Recitals           Recitals           Recitals           4.3(d)(i)           Preamble           Recitals           4.4(c)(iii)           6.1(a)           4.4(b)(ii)           4.3(p)           4.3(k)(i)           4.3(c)(iv)           3.10(a)           4.3(b)(ii)           4.3(c)(iv)           3.10(a)           4.3(b)(ii)           4.2(c)(iv)           Recitals           3.10(b)           Preamble           7.1(f)(i)           6.1(a)
MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding Class A Common Stock MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Necessary Consents MVT Holding Prospectus MVT Holding Voting Debt MVT Lease MVT Material Contracts MVT Necessary Consents MVT Option MVT Permits MVT Significant Subsidiary MVT Voting Debt NASD New MI Corp. New MI Corp. Common Stock New MI Corp. Option Parties Private Letter Ruling Proxy Statement Purchase Price Required Approvals Required Retained Cash	$\begin{array}{c} \mbox{Recitals} \\ \mbox{Preamble} \\ \mbox{4.3} \\ \mbox{Recitals} \\ \mbox{Recitals} \\ \mbox{Recitals} \\ \mbox{Employee Matters} \\ \mbox{Agreement} \\ \mbox{4.3(d)(i)} \\ \mbox{Preamble} \\ \mbox{Recitals} \\ \mbox{Recitals} \\ \mbox{4.4(c)(iii)} \\ \mbox{6.1(a)} \\ \mbox{4.4(b)(ii)} \\ \mbox{4.3(p)} \\ \mbox{4.3(k)(i)} \\ \mbox{4.3(c)(iv)} \\ \mbox{3.10(a)} \\ \mbox{4.3(c)(iv)} \\ \mbox{3.10(a)} \\ \mbox{4.3(b)(ii)} \\ \mbox{4.3(b)(ii)} \\ \mbox{4.3(b)(ii)} \\ \mbox{4.3(b)(ii)} \\ \mbox{4.3(c)(iv)} \\ \mbox{3.10(b)} \\ \mbox{Preamble} \\ \mbox{7.1(f)(i)} \\ \mbox{6.1(a)} \\ \mbox{2.1} \end{array}$
MVT Common Stock MVT Corp. MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Financial Statements MVT Holding Class A Common Stock MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Necessary Consents MVT Holding Voting Debt MVT Lease MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Necessary Consents MVT Significant Subsidiary MVT Voting Debt NASD New MI Corp. New MI Corp. Common Stock New MI Corp. Option Parties Private Letter Ruling Proxy Statement Purchase Price Required Approvals	$\begin{array}{c} {\rm Recitals} \\ {\rm Preamble} \\ {\rm 4.3} \\ {\rm Recitals} \\ {\rm Recitals} \\ {\rm Employee Matters} \\ {\rm Agreement} \\ {\rm 4.3(d)(i)} \\ {\rm Preamble} \\ {\rm Recitals} \\ {\rm 4.3(d)(i)} \\ {\rm Preamble} \\ {\rm Recitals} \\ {\rm 4.4(c)(iii)} \\ {\rm 6.1(a)} \\ {\rm 4.4(b)(ii)} \\ {\rm 4.3(p)} \\ {\rm 4.3(k)(i)} \\ {\rm 4.3(c)(iv)} \\ {\rm 3.10(a)} \\ {\rm 4.3(c)(iv)} \\ {\rm 3.10(a)} \\ {\rm 4.3(a)(ii)} \\ {\rm 4.3(a)(ii)} \\ {\rm 4.3(b)(ii)} \\ {\rm 4.2(c)(iv)} \\ {\rm Recitals} \\ {\rm Recitals} \\ {\rm Recitals} \\ {\rm 3.10(b)} \\ {\rm Preamble} \\ {\rm 7.1(f)(i)} \\ {\rm 6.1(a)} \\ {\rm 2.1} \\ {\rm 6.4(a)} \end{array}$
MVT Corp. MVT Disclosure Schedule MVT Disclosure Schedule MVT Distribution MVT Dividend MVT Employee MVT Financial Statements MVT Holding MVT Holding Class A Common Stock MVT Holding Common Stock MVT Holding Necessary Consents MVT Holding Necessary Consents MVT Holding Necessary Consents MVT Holding Voting Debt MVT Holding Voting Debt MVT Lease MVT Material Contracts MVT Necessary Consents MVT Option MVT Significant Subsidiary MVT Significant Subsidiary MVT Voting Debt NASD New MI Corp. New MI Corp. New MI Corp. Option Parties Private Letter Ruling Proxy Statement Proxy Statement Purchase Price Required Approvals Required Approvals	$\begin{array}{c} \mbox{Recitals} \\ \mbox{Preamble} \\ \mbox{4.3} \\ \mbox{Recitals} \\ \mbox{Recitals} \\ \mbox{Recitals} \\ \mbox{Employee Matters} \\ \mbox{Agreement} \\ \mbox{4.3(d)(i)} \\ \mbox{Preamble} \\ \mbox{Recitals} \\ \mbox{Recitals} \\ \mbox{4.4(c)(iii)} \\ \mbox{6.1(a)} \\ \mbox{4.4(b)(ii)} \\ \mbox{4.3(p)} \\ \mbox{4.3(k)(i)} \\ \mbox{4.3(c)(iv)} \\ \mbox{3.10(a)} \\ \mbox{4.3(c)(iv)} \\ \mbox{3.10(a)} \\ \mbox{4.3(b)(ii)} \\ \mbox{4.3(b)(ii)} \\ \mbox{4.3(b)(ii)} \\ \mbox{4.3(b)(ii)} \\ \mbox{4.3(c)(iv)} \\ \mbox{3.10(b)} \\ \mbox{Preamble} \\ \mbox{7.1(f)(i)} \\ \mbox{6.1(a)} \\ \mbox{2.1} \end{array}$

Sarbanes Act Securities Act Separation Agreement Share Distribution Share Distribution Time Share Issuance Shares Shareholders Agreement Superior MVT Acquisition Proposal Surplus and Solvency Opinions Surviving Corporation Termination Agreement 4.2(d)(i) 4.1(b)(iii) Recitals Recitals 3.1(h) Recitals 2.1 6.14 6.5(b) 6.17 3.1(a) 4.3(l)(vi)

Termination Date
Termination Fee
Transactions
Valuation Firm
Violation
WBCL
WDFI
WLLCL

### ARTICLE II

### SHARE PURCHASE

Section 2.1 <u>Share Purchase</u>. Upon the terms and subject to the conditions of this Agreement, at the Closing and immediately after the MVT Distribution, but prior to the MI Contribution, MVT Holding shall issue and sell to Investor and Investor shall purchase from MVT Holding a number of previously unissued shares of MVT Holding Class A Common Stock determined in accordance with <u>Exhibit G</u> (the "<u>Shares</u>"), which Shares shall be subject to the transfer restrictions and other terms set forth in the Shareholders Agreement. In consideration for the issuance and sale of the Shares, and upon the terms and subject to the conditions of this Agreement, at the Closing and immediately after the MVT Distribution, but prior to the MI Contribution, Investor shall pay, or cause to be paid, to MVT Holding an amount equal to \$625 million (the "<u>Purchase Price</u>") in accordance with <u>Section 2.2</u>.

Section 2.2 <u>Payment Terms of the Share Purchase</u>. At the Closing and immediately after the MVT Distribution, but prior to the MI Contribution, upon the terms and subject to the conditions of this Agreement MVT Holding shall issue and deliver to Investor a certificate representing the Shares duly registered in the name of Investor. Prior to the MI Merger, Investor shall deliver the Purchase Price provided for in <u>Section 2.1</u> by wire transfer of immediately available funds to an account designated by MVT Holding at least two days prior to the Closing Date and effective immediately after the MVT Distribution but prior to the MI Contribution such Purchase Price shall be deemed to have been paid in satisfaction of the Investor's obligations under <u>Section 2.1</u>.

Section 2.3 <u>Investor Share Number</u>. During the two trading days prior to the Closing Date, Investor and MI Corp. shall mutually determine in good faith the Investor Share Number in a manner consistent with the terms and methodology set forth on <u>Exhibit G</u>.

## ARTICLE III

# MERGER, OTHER TRANSACTIONS, EXCHANGE OF SHARES, ETC.

Section 3.1 <u>Merger and Other Transactions</u>. Upon the terms and subject to the conditions set forth in this Agreement, the transactions set forth in this <u>Section 3.1</u> shall take place in the order set forth below:

(a) <u>The MI Merger</u>. At the MI Effective Time, Merger Sub shall be merged with and into MI Corp. in accordance with the Wisconsin business corporation law (the "<u>WBCL</u>"). MI Corp. shall be the surviving corporation (the "<u>Surviving Corporation</u>") in the MI Merger and shall continue its corporate existence under the laws of the State of Wisconsin and shall succeed to and assume all of the rights and obligations of MI Corp. and Merger Sub in accordance with, and shall have the effects set forth in, this Agreement and the WBCL. As a result of the MI Merger, MI Corp. shall become a wholly-owned subsidiary of MVT Holding. Upon consummation of the MI Merger, the separate corporate existence of Merger Sub shall terminate.

(b) <u>The MI Conversion</u>. Immediately following the MI Effective Time, MI Corp. shall effect the MI Conversion pursuant to applicable provisions of the WBCL and the Wisconsin limited liability company law (the "<u>WLLCL</u>"). Following the MI Conversion, (i) MI LLC shall be a limited liability company which shall be disregarded as an entity separate from MVT Holding for U.S. federal income tax purposes and (ii) all of the membership interests of MI LLC shall be held by MVT Holding.

(c) <u>The MVT Distribution</u>. Immediately following the Conversion Effective Time, MI LLC shall distribute all of the outstanding shares of MVT Common Stock to MVT Holding. Following the MVT Distribution, all of the outstanding shares of capital stock of MVT Corp. shall be owned by MVT Holding.

(d) <u>Share Purchase; Debt Financing</u>. Immediately following the MVT Distribution Time, (i) Investor shall purchase the Shares and deliver the Purchase Price to MVT Holding in consideration of the Shares in accordance with <u>Article II</u> hereof and (ii) the Debt Financing shall be consummated by MVT Holding and/or one or more of its wholly-owned Subsidiaries.

(e) <u>Payment of the Intercompany Debt Amount</u>. Upon receipt of the proceeds from the Debt Financing, MVT Holding and/or one of its Subsidiaries shall pay to MI LLC an amount in cash equal to the Intercompany Debt Amount.

(f) <u>MVT Dividend</u>. Immediately following the purchase of Shares by Investor, the consummation of the Debt Financing and the payment of the Intercompany Debt Amount to MI LLC, MVT Corp. shall pay the MVT Dividend to MVT Holding.

(g) <u>The MI Contribution</u>. Immediately following the MVT Dividend, MVT Holding shall contribute to New MI Corp. (i) all of the membership interests of MI LLC and (ii) the MI Cash Contribution. Following the MI Contribution, all of the outstanding membership interests in MI LLC shall be owned by New MI Corp.

(h) <u>Amended and Restated Articles of Incorporation</u>. Prior to the Share Distribution Time, MVT Holding and New MI Corp. shall, or shall cause New MI Corp. to file amended and restated articles of incorporation of New MI Corp. with the Department of Financial Institutions of the State of Wisconsin (the "<u>WDFI</u>"), which, among other things, shall change the name of New MI Corp. to Marshall & Ilsley Corporation.

(i) <u>Share Distribution</u>. Upon the terms and subject to the conditions of the Separation Agreement, following the MI Contribution and at the time established by the Board of Directors of MVT Holding (the "<u>Share Distribution Time</u>"), MVT Holding shall effect the Share Distribution (based upon a distribution ratio of three shares of New MI Corp. Common Stock for every one share of MVT Holding Common Stock) as contemplated by the Separation Agreement. Effective immediately following

the Distribution Time, the Board of Directors of MVT Holding shall be reconstituted as contemplated by Section 6.2.

Section 3.2 <u>Closing</u>. The closing of the Closing Transactions (the "<u>Closing</u>") will take place as soon as practicable, but in any event within three Business Days after the satisfaction or waiver (subject to Applicable Laws) of the conditions (excluding conditions that, by their nature, cannot be satisfied until the Closing Date) set forth in <u>Article VII</u> (but not prior to October 2, 2007), unless this Agreement has been theretofore terminated pursuant to its terms or unless another time or date is agreed to in writing by the parties hereto (the actual time and date of the Closing being referred to herein as the "<u>Closing</u> <u>Date</u>")). The Closing shall be held at the offices of Sidley Austin LLP, One South Dearborn Street, Chicago, Illinois, unless another place is agreed to in writing by the parties hereto.

Section 3.3 <u>Effective Time of the MI Merger and the MI Conversion</u>. Upon the terms and subject to the conditions of this Agreement, (a) the MI Merger shall become effective as set forth in the articles of merger relating thereto (the "<u>MI Articles of Merger</u>") that shall be filed with the WDFI on or prior to the Closing Date and (b) the MI Conversion shall become effective as soon as practicable on the Closing Date following the MI Effective Time, as set forth in the certificate of conversion of MI Corp. (the "<u>MI Certificate of Conversion</u>") and the articles of organization of MI LLC (the "<u>MI Certificate of Formation</u>"), each of which shall be filed with the WDFI on or prior to the Closing Date. The time the MI Merger shall become effective as set forth in the MI Articles of Merger is referred to as the "<u>MI Effective Time</u>," and the time the MI Conversion shall become effective as set forth in the MI Certificate of Conversion and the MI Certificate of Formation is referred to as the "<u>Conversion Effective Time</u>."

Section 3.4 <u>Effects of the MI Merger and the MI Conversion</u>. The MI Merger and the MI Conversion shall have the effects set forth in this Agreement and the applicable provisions of the WBCL and the WLLCL, respectively.

## Section 3.5 Organizational Documents of MI Corp., MVT Holding and MI LLC.

(a) (i) At the MI Effective Time, (A) the articles of incorporation of MI Corp., as in effect immediately prior to the MI Effective Time, shall be the articles of incorporation of MI Corp., as the surviving corporation in the MI Merger and (B) the by-laws of MI Corp., as in effect immediately prior to the MI Effective Time, shall be the by-laws of MI Corp. as the surviving corporation in the MI Merger, in each case until superseded by the MI Certificate of Formation filed as part of the MI Conversion; and

(ii) At the Conversion Effective Time, MVT Holding shall cause MI LLC to adopt an operating agreement.

(b) MI Corp. and MVT Holding shall take all necessary action so that, at the MI Effective Time, (i) the articles of incorporation of MVT Holding shall be

in the form attached as Exhibit D hereto and (ii) the by-laws of MVT Holding shall be in the form attached as Exhibit E hereto.

Section 3.6 Effect on Capital Stock. At the MI Effective Time, by virtue of the MI Merger and without any action on the part of the holders of any capital stock of Merger Sub or MI Corp.:

(a) Each share of MI Common Stock issued and outstanding immediately prior to the MI Effective Time (other than shares of MI Common Stock that are held by MI Corp. in treasury or that are owned by any Subsidiary of MI Corp.) shall be automatically converted into one-third of a fully paid and nonassessable (except to the extent provided in former Section 180.0622(2)(b) of the WBCL) share of MVT Holding Common Stock. Each share of MI Common Stock held by MI Corp. in treasury or owned by MI Corp. or any of its Subsidiaries shall be cancelled and shall cease to exist and no shares of stock or other consideration shall be delivered in exchange therefor.

(b) All shares of MI Common Stock converted into MVT Holding Common Stock pursuant to this <u>Article III</u> shall no longer be outstanding and shall automatically be canceled and shall cease to exist, and each certificate or book-entry credit previously evidencing any such shares of MI Common Stock (an "<u>MI Certificate</u>") shall thereafter evidence only rights with respect to certificates or book-entry credits (or cash in lieu of fractional shares in accordance with <u>Section 3.6(c)</u>) representing the number of shares of MVT Holding Common Stock into which the shares of MI Common Stock formerly evidenced by such MI Certificate have been converted pursuant to this <u>Section 3.6</u>.

(c) No certificates for fractions of shares of MVT Holding Common Stock and no scrip or other certificates evidencing fractional interests in such shares shall be issued pursuant to <u>Article III</u>. If the conversion of a Person's aggregate holdings of MI Common Stock at any time results in a fractional share of MVT Holding Common Stock or interest therein, such Person shall, in lieu thereof, be entitled to be paid cash in an amount equal to the value of such fractional share or interest based on the Closing Price of MI Common Stock on the last Business Day prior to the MI Effective Time. Any person otherwise entitled to a fractional share or interest shall not be entitled by reason thereof to any voting, dividend or other rights as a stockholder of MVT Holding. The "<u>Closing Price</u>" of MI Common Stock on any Business Day shall for all purposes of this Agreement be the last sale price, or the closing bid price if no sale occurred, of MI Common Stock on The New York Stock Exchange, Inc. or such other national securities exchange on which MI Common Stock is then principally traded.

(d) Each share of common stock, par value \$0.01 per share, of Merger Sub ("<u>Merger Sub Common Stock</u>") issued and outstanding immediately prior to the MI Effective Time shall be automatically converted into one fully paid and nonassessable (except to the extent provided in former Section 180.0622(2)(b) of the

WBCL) share of common stock, par value \$0.01 per share, of the Surviving Corporation.

(e) Each share of MVT Holding Common Stock held by MI Corp. immediately prior to the MI Effective Time shall be cancelled and shall cease to exist and no stock or other consideration shall be delivered in exchange therefor.

## Section 3.7 Exchange of Shares.

(a) On or prior to the Closing Date, MI Corp. and MVT Holding shall deposit, or shall cause to be deposited, with Continental Stock Transfer and Trust Company or a bank or trust company appointed by MVT Holding prior to the Closing Date and reasonably acceptable to Investor (the "<u>Exchange Agent</u>"), for exchange in accordance with this <u>Article III</u>, certificates representing the shares of MVT Holding Common Stock to be issued in connection with the MI Merger pursuant to <u>Section 3.6</u> and delivered pursuant to <u>Section 3.7(b)</u> in exchange for MI Certificates and cash in lieu of fractional shares to be issued thereon in accordance with <u>Section 3.6(c)</u> (such shares of MVT Holding Common Stock and cash being referred to as the "<u>Exchange Fund</u>").

(b) MI Corp. and MVT Holding shall instruct the Exchange Agent to deliver after the MI Effective Time, to each holder of record of MVT Holding Common Stock as of the MI Effective Time, upon surrender of such holder's certificates representing shares of MI Common Stock, a certificate representing the number of shares of MVT Holding Common Stock and/or a check payable to such holder to effect the payment of cash in lieu of any fractional shares to be delivered in accordance with <u>Section 3.6(c)</u> into which such holder's shares of MI Common Stock shall have been converted pursuant to the provisions of this <u>Article III</u> and the MI Certificate as converted shall forthwith be canceled.

(c) Other than the Share Distribution, no dividends or other distributions will be declared on the MVT Holding Common Stock in connection with the Transactions. The shares of New MI Corp. Common Stock to be distributed in the Share Distribution shall be held by the Exchange Agent pursuant to <u>Article III</u> of the Separation Agreement.

(d) All shares of MVT Holding Common Stock and/or cash in lieu of fractional shares in accordance with <u>Section 3.6(c)</u> issued or delivered upon conversion of MI Common Stock in accordance with the terms of this <u>Article III</u> shall be deemed to have been issued in full satisfaction of all rights pertaining to shares of MI Common Stock converted in the MI Merger in accordance with <u>Section 3.6</u>. At the MI Effective Time, there shall be no further transfers on the stock transfer books of MI Corp. of the shares of MI Common Stock that were issued and outstanding immediately prior to the MI Effective Time.

(e) Any portion of the Exchange Fund that remains unclaimed by holders of MI Common Stock entitled thereto for twelve months after the MI Effective Time shall be delivered to MVT Holding, and any holders of record of MI Common

Stock as of the MI Effective Time who have not theretofore complied with this <u>Article III</u> shall thereafter look only to MVT Holding for payment of the shares of MVT Holding Common Stock and any unpaid dividends and distributions on MVT Holding Common Stock deliverable in respect of each share of MI Common Stock formerly evidenced by such MI Certificates as determined pursuant to this Agreement, without any interest thereon, and MVT Holding shall not charge such holders in connection with the delivery of such shares of MVT Holding Common Stock, unpaid dividends or distributions and cash in lieu of fractional shares. Any such portion of the Exchange Fund remaining unclaimed by holders of MI Common Stock entitled thereto five years after the MI Effective Time (or such earlier date immediately prior to such time as such amounts would otherwise escheat to or become property of any Governmental Entity) shall, to the extent permitted by Applicable Laws, become the property of MVT Holding free and clear of any claims or interest of any Person previously entitled thereto.

(f) None of Investor, MVT Holding, MI Corp., New MI Corp., Merger Sub, the Surviving Corporation, the Exchange Agent or any other Person shall be liable to any holder of MI Common Stock as of the MI Effective Time for any shares of MVT Holding Common Stock and any dividend or other distribution or cash in lieu of fractional shares with respect thereto delivered in good faith to a public official pursuant to applicable abandoned property, escheat or similar Applicable Laws.

## Section 3.8 Directors and Officers of MVT Holding and MI Corp.

(a) <u>MI Corp.</u> The officers of MI Corp. at the MI Effective Time shall be the initial officers of the Surviving Corporation and the directors of MI Corp. at the MI Effective Time shall be the initial directors of the Surviving Corporation, in each case, until such time as their respective successors are duly elected or appointed and qualified in the manner provided in the articles of incorporation or by-laws of the Surviving Corporation or as otherwise provided by Applicable Law.

(b) <u>MVT Holding</u>. The officers and directors of MVT Holding at the MI Effective Time shall be the officers and directors of MVT Holding immediately prior to the MI Effective Time, until the officers and directors are duly elected or appointed pursuant to <u>Section 6.2</u>.

### Section 3.9 Other Transaction Matters.

(a) <u>Principal Corporate Offices</u>. MVT Holding's headquarters and principal corporate offices immediately following the Transactions shall be the current MVT Corp., headquarters and principal corporate offices, and New MI Corp.'s headquarters and principal corporate offices shall be the current MI Corp. headquarters and principal corporate offices in Milwaukee, Wisconsin.

(b) <u>Ticker Symbol</u>. The Parties shall use their reasonable best efforts to cause New MI Corp. to have the NYSE ticker symbol "MI."

Section 3.10 Options; Restricted Stock.

(a) As of the MI Effective Time, each MI Option held by a MVT Employee that is outstanding as of the MI Effective Time shall be converted into an option to purchase shares of MVT Holding Common Stock (a "<u>MVT Option</u>"), as set forth in the Employee Matters Agreement. After the MI Effective Time, except as provided above in this <u>Section 3.10</u>, each MVT Option shall have the same terms and conditions as the corresponding MI Option to which it relates, as set forth in the Employee Matters Agreement.

(b) As of the MI Effective Time, each MI Option held by an employee of the MI Group that is outstanding as of the MI Effective Time shall be converted into an option to purchase shares of New MI Corp. (a "<u>New MI Corp. Option</u>"), as set forth in the Employee Matters Agreement. After the MI Effective Time, except as provided above in this <u>Section 3.10</u>, each New MI Corp. Option shall have the same terms and conditions as the corresponding MI Option to which it relates, as set forth in the Employee Matters Agreement.

(c) As of the MI Effective Time, each share of MI Restricted Stock shall be treated the same as all other outstanding shares of MI Corp. Common Stock in the transactions contemplated by the Investment Agreement and the Separation Agreement; <u>provided</u>, <u>however</u>, that each share of MVT Holding Common Stock and New MI Corp. Common Stock received in such transactions shall be subject to the same restrictions as the corresponding share of MI Restricted Stock to which such share relates and shall continue to be subject to the terms of the applicable MI Option Plan.

Section 3.11 <u>MVT Holding Class A Common Stock</u>. At 12:01 a.m. Eastern Standard Time on the first day following the Closing Date, each outstanding share of MVT Holding Class A Common Stock shall automatically be converted into one share of MVT Holding Common Stock.

Section 3.12 <u>Initiation of the Closing Transactions</u>. The parties hereto hereby acknowledge and agree that upon consummation of the MI Merger all conditions precedent to the Closing set forth in <u>Sections 7.1</u>, <u>7.2</u> and <u>7.3</u> shall be deemed to have been satisfied or waived in full and it is the agreement of the parties that once the MI Merger is consummated all other Closing Transactions shall be consummated as set forth in <u>Section 3.1</u>.

### ARTICLE IV

#### **REPRESENTATIONS AND WARRANTIES**

Section 4.1 <u>Representations and Warranties of Investor</u>. Investor represents and warrants to MVT Holding, MI Corp. and MVT Corp. as follows:

(a) <u>Organization, Standing and Power</u>. Investor is a limited partnership duly organized, validly existing and in good standing under the laws of the State of Delaware. Investor has not conducted any business prior to the date hereof and has no, and prior to the Closing Date will have no, assets, liabilities or obligations of any nature other than those incident to its formation and pursuant to this Agreement and

the Transactions. Investor and its Affiliates do not own beneficially or of record any shares of capital stock of MI Corp. As of the date hereof, Investor is a direct or indirect wholly-owned subsidiary of the Equity Fund.

# (b) Authority; No Conflicts.

(i) Investor has all requisite limited partnership power and authority to enter into this Agreement and the Transaction Agreements with respect to which Investor is contemplated thereby to be a party and to consummate the Transactions. The execution and delivery of this Agreement and the Transaction Agreements with respect to which Investor is contemplated thereby to be a party by Investor and the consummation by Investor of the Transactions have been duly authorized by all necessary limited partnership action on the part of Investor. This Agreement has been, and the Transaction Agreements with respect to which Investor is contemplated thereby to be a party will be, duly executed and delivered by Investor and, assuming the due authorization and valid execution and delivery of this Agreement or the applicable Transaction Agreement with respect to which Investor is contemplated thereby to be a party by the other parties hereto and thereto, as applicable, constitutes or will constitute a valid and binding agreement of Investor, enforceable against Investor in accordance with their respective terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, fraudulent conveyance, moratorium and similar Applicable Laws relating to or affecting creditors generally or by general equity principles (regardless of whether such enforceability is considered in a proceeding in equity or at law).

(ii) The execution and delivery of this Agreement by Investor does not, the execution and delivery by Investor of the Transaction Agreements with respect to which Investor is contemplated thereby to be a party will not, and the consummation of the Transactions will not, conflict with, or result in any breach or violation of, or constitute a default (with or without notice or lapse of time, or both) under, or give rise to a right of or result by its terms in the termination, amendment, cancellation or acceleration of any obligation or the loss of a benefit under, or the creation of any material pledges, claims, liens, charges, encumbrances and security interests of any kind or nature whatsoever (collectively, "Liens"), "put" or "call" right or other encumbrance on, or the loss of, any assets (any such conflict, breach, violation, default, right of termination, amendment, cancellation or acceleration, loss or creation, a "Violation") (with or without notice or lapse of time, or both) under: (A) any provision of the organizational documents of Investor or (B) except as, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on Investor, subject to obtaining or making Investor Necessary Consents, (1) any loan or credit agreement, note, instrument, mortgage, bond, indenture real estate or other lease or sublease, benefit plan, license, sublicense, memorandum of understanding, sales order, purchase order, open bid or other contract, agreement or obligation, in each case, including all amendments, modifications and supplements thereto and waivers and consents thereunder (a "Contract") to which Investor or any of its Subsidiaries is a party or by which any of them or any of their respective properties or assets is bound or (2) any permit, concession, franchise, license, judgment, order, decree, statute, law, ordinance, rule or regulation applicable to Investor or any Subsidiary of Investor or their respective properties or assets.

(iii) No consent, waiver, approval, order or authorization of, or registration, declaration or filing with, any supranational, national, federal, state, municipal, local

or foreign government, any instrumentality, subdivision, court, administrative agency, board, commission or other authority thereof, any arbitral tribunal, or any quasi-governmental or private body exercising any regulatory, taxing, importing or other governmental or quasi-governmental authority (a "<u>Governmental Entity</u>") or any other Person is required to be obtained or made by or with respect to Investor or any Subsidiary of Investor in connection with the execution and delivery of this Agreement and the Transaction Agreements with respect to which Investor is contemplated thereby to be a party by Investor or the consummation by Investor of the Transactions, except for those required under or in relation to (A) the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "<u>HSR Act</u>"), (B) state securities or "blue sky" laws or regulations, (C) the Securities Act of 1933, as amended (the "<u>Securities</u> <u>Act</u>"), (D) the Securities Exchange Act of 1934, as amended (the "<u>Exchange Act</u>"), (E) the rules and regulations of the NYSE and, if different, such national securities exchange designated by MVT Holding and Investor pursuant to <u>Section 6.15</u>. (F) antitrust or other competition laws of other jurisdictions, (G) applicable state or federal banking laws or regulations and (H) such consents, waivers, approvals, orders, authorizations, registrations, declarations and filings the failure of which to make or obtain, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on Investor. Consents, approvals, orders, authorizations, registrations, declarations and filings required under or in relation to any of the foregoing <u>clauses (A)</u> through (G) are referred to as the "Investor Necessary Consents."

(iv) Equity Fund is the "ultimate parent entity" (as defined in 16 C.F.R. Section 801.1(a)(3)) of the Investor.

#### (c) Information Supplied.

(i) None of the information supplied or to be supplied by Investor specifically for inclusion or incorporation by reference in (A) the Form S-4 will, at the time it becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, (B) the Proxy Statement will, on the date it is first mailed to MI Corp. shareholders and at the time of the MI Shareholders Meeting, respectively, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading and (C) the Form 10 will, at the time the Form 10 becomes effective, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading and (C) the Form 10 will, at the time the Form 10 becomes effective, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

(ii) Notwithstanding the foregoing provisions of this <u>Section 4.1(c)</u>, no representation or warranty is made by Investor with respect to statements made or incorporated by reference in the Form S-4, Form 10 or the Proxy Statement based on information supplied by or on behalf of MI Corp. or MVT Corp. for inclusion or incorporation by reference therein.

(d) <u>Litigation; Compliance with Laws</u>. (i) There is no suit, action, arbitration, proceeding, claim, charge, regulatory or accrediting agency investigation or other proceeding (an "<u>Action</u>") pending or, to the Knowledge of Investor, threatened

against Investor or any property or asset of Investor or any Affiliate of Investor which, individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect on Investor, nor is there any judgment, decree, injunction, rule or order of any Governmental Entity or arbitrator outstanding against Investor or any Affiliate of Investor which, individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect on Investor or any Affiliate of Investor which, individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect on Investor.

(ii) None of Investor or any of its Affiliates is in violation of, and Investor and its Subsidiaries have not received since Investor's inception any written notices of violations with respect to, any Applicable Laws, except for violations which, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on Investor.

(e) <u>Brokers or Finders</u>. No agent, broker, investment banker, financial advisor or other firm or Person is or will be entitled to any broker's or finder's fee or any other similar commission or fee in connection with any of the Transactions based upon arrangements made by or on behalf of Investor or any of its Affiliates, except Morgan Stanley & Co. Incorporated, whose fees and expenses shall be paid by Investor (or its Affiliates).

(f) Equity Financing. Concurrently with the execution of this Agreement, Investor has delivered true, correct and complete copies of (i) an executed commitment letter, dated the date hereof (the "Equity Commitment Letter"), from Warburg Pincus Private Equity IX, L.P. (the "Equity Fund") to MVT Corp. and MVT Holding to provide equity financing in an aggregate amount of \$625,000,000 to fund the Purchase Price (such equity financing from the Equity Fund and/or any Successor Fund in accordance with the terms hereof, the "Equity Financing"), and (ii) an executed copy of the Limited Guarantee. As of the date hereof, the Equity Commitment Letter and the Limited Guarantee in the form so delivered are in full force and effect and no event has occurred which, with or without notice, lapse of time or both, would constitute a default or breach on the part of Investor under any term or condition of the Equity Commitment Letter or the Limited Guarantee. There are no conditions precedent or other contractual contingencies to Investor's right to require the funding of the full amount of the Equity Financing, other than as set forth in or contemplated by the Equity Commitment Letter. As of the date of this Agreement, assuming the accuracy of the representations set forth in <u>Sections 4.2, 4.3</u> and 4.4 and the satisfaction of the conditions set forth in <u>Sections 7.1</u> and <u>7.2</u>. Investor does not have any reason to believe that any of the conditions to the Equity Financing will not be satisfied or that the Equity Financing will not be available to Investor on the Closing Date.

(g) <u>Acquisition for Investment</u>. Investor is acquiring the Shares being purchased by it for its own account for the purpose of investment and not with a view to or for sale in connection with any distribution thereof, and Investor has no present intention or plan to effect any distribution of shares of MVT Holding Class A Common Stock; <u>provided</u>, <u>however</u>; that the disposition of such Investor's property, including any shares of MVT Holding Class A Common Stock, shall at all times be and remain within its control, subject to the provisions of this Agreement and the

Transaction Agreements, including the restrictions on transfer set forth in the Shareholders Agreement.

(h) <u>No Investment Company</u>. Investor is not, and after giving effect to the Share Issuance will not be, required to register as an "investment company" as such term is defined in the Investment Company Act of 1940, as amended.

(i) <u>Ownership of Stock</u>. Neither Investor nor any Investor Tax Affiliate owns any shares, or options to acquire shares, of MI Common Stock or has any plan or intention to acquire any shares, or options to acquire shares, of capital stock of MVT Holding or any of its Subsidiaries (other than the Shares and shares acquired in a transaction that complies with Section 4.02 of the Tax Allocation Agreement).

Section 4.2 <u>Representations and Warranties of MI Corp.</u> Except as set forth in the MI Disclosure Schedule delivered to Investor prior to the execution of this Agreement (the "<u>MI Disclosure Schedule</u>") (each section of which, to the extent specified therein, qualifies the correspondingly numbered representation and warranty of MI Corp. herein and any disclosure in such section qualifies any other representation and warranty of MI Corp. or MVT Corp. contained herein to which its application or relevance is reasonably apparent on its face), MI Corp. represents and warrants to Investor as follows:

(a) <u>Organization</u>. (i) MI Corp. is a corporation duly organized, validly existing and in active status under the laws of the State of Wisconsin. MI Corp is duly registered as a bank holding company under the Bank Holding Company Act of 1956, as amended (the "<u>BHC Act</u>") and is a financial holding company pursuant to Section 4(l) of the BHC Act and meets the applicable requirements for qualification as such. Each of MI Corp. and the Subsidiaries of MI Corp. is duly qualified and in good standing or active status under the laws of its jurisdiction of incorporation or organization, and in each jurisdiction in which the property owned, leased or operated by the MVT Business or the nature of the MVT Business makes such qualification necessary, in each case other than in such jurisdictions where the failure to so qualify or to be in good standing, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on MVT Corp. or a Material Adverse Effect on MI Corp. True, correct and complete copies of the articles of incorporation and by-laws of MI Corp. in effect on the date hereof have been delivered or made available to Investor.

(ii) Merger Sub is a corporation duly organized, validly existing and in good standing under the laws of Wisconsin. Prior to the MI Merger, Merger Sub will not have conducted any operations and will have no assets or liabilities other than those incident to Merger Sub's formation and those pursuant to this Agreement and the Transactions. True, correct and complete copies of the articles of incorporation and by-laws of Merger Sub in effect on the date hereof have been delivered or made available to Investor.

## (b) Capital Structure.

(i) The authorized capital stock of MI Corp. consists of 700,000,000 shares of MI Common Stock and 5,000,000 shares of Preferred Stock (the "MI Preferred Stock") of which, as of this date hereof, 2,000,000 shares of the MI Preferred Stock have been designated Series A Convertible Preferred Stock. At the close of business on March 28, 2007 (the "Capitalization Date"), (A) 256,673,530 shares of MI Common Stock were issued and outstanding, all of which were validly issued, fully paid and nonassessable (subject to the personal liability which may be imposed on shareholders by former Section 180.0622(2)(b) of the WBCL for debts incurred prior to June 14, 2006 (for debts incurred on or after such date, Section 180.0622(2)(b) of the WBCL has been repealed) owing to employees for services performed, but not exceeding six months' service in any one case) and free of preemptive rights, (B) no shares of MI Preferred Stock were issued or outstanding, (C) 5,298,894 shares of MI Common Stock were held in the treasury of MI Corp.; (D) 17,644,536 shares of MI Common Stock were reserved for issuance pursuant to outstanding options (the "MI Options") to purchase shares of MI Common Stock pursuant to the plans and agreements listed in Section 4.2(b)(i) of the MI Disclosure Schedule (collectively, the "MI Option Plans"); (E) 824,682 shares of MI Common Stock were reserved for issuance pursuant to MI Corp.'s Amended and Restated 2000 Employee Stock Purchase Plan (the "MI Stock Purchase Plan"); (F) the issued and outstanding MI Common Stock included 601,620 restricted shares issued to employees of MI Corp. (the "MI Restricted Stock") pursuant to the plans and agreements listed in Section 4.2(b)(i) of the MI Disclosure Schedule (collectively, the "MI Restricted Stock Plans"); and (G) 10,800,000 shares of Common Stock were reserved for issuance pursuant to units of Common SPACES issued by MI Corp. and an Affiliate thereof (the "MI SPACES Plan"). Except as set forth in the preceding sentence and except for shares reserved for issuance pursuant to outstanding registration statements of MI Corp., and additional shares issuable upon anti-dilution adjustment of the MI SPACES Plan, as of the Capitalization Date, (x) no shares of capital stock or other voting securities of MI Corp. were issued, reserved for issuance or outstanding, and (y) there are no options, warrants, calls, rights, puts or Contracts to which MI Corp. or any of its Subsidiaries is a party or by which any of them is bound obligating MI Corp. or any of its Subsidiaries to issue, deliver, sell or redeem or otherwise acquire, or cause to be issued, delivered, sold or redeemed or otherwise acquired, any additional shares of capital stock or MI Voting Debt (or any securities convertible or exercisable or exchangeable for shares of capital stock of MI Corp. or any of its Subsidiaries or MI Voting Debt) of MI Corp. or any of its Subsidiaries or obligating MI Corp. or any of its Subsidiaries to grant, extend or enter into any such option, warrant, call, right, put or Contract. As of the Capitalization Date, the MI Stock Option Plans, MI Stock Purchase Plan and the MI Restricted Stock Plans are the only benefit plans of MI Corp. or its Subsidiaries under which any shares of capital stock of MI Corp. or any of its Subsidiaries are issuable or which provide for stock appreciation rights or other rights the value of which is determined in whole or in part by reference to the value of securities of MI Corp. or any of its Subsidiaries. Each share of MI Common Stock which may be issued pursuant to any MI Option Plan, MI Stock Purchase Plan, MI Restricted Stock Plan or MI SPACES Plan has been duly authorized and, if and when issued pursuant to the terms thereof, will be validly issued, fully paid and nonassessable and free of preemptive rights.

(ii) Except as set forth in clause (i) above, as of the Capitalization Date, no bonds, debentures, notes or other indebtedness of MI Corp. having the right to vote (or

convertible into or exchangeable for securities having the right to vote) on any matters on which shareholders of MI Corp. may vote ("<u>MI Voting Debt</u>") are issued or outstanding.

(iii) Other than the Transaction Agreements, as of the date hereof, there are no shareholder agreements, voting trusts or other Contracts to which MI Corp. is a party or by which it is bound relating to the voting or transfer of any shares of capital stock of MI Corp.

(iv) The authorized capital stock of Merger Sub consists of 1,000 shares of Merger Sub Common Stock, all of which are owned beneficially and of record by MVT Holding and are validly issued, fully paid and nonassessable.

#### (c) Authority; No Conflicts.

(i) MI Corp. has all requisite corporate power and authority to enter into this Agreement and the Transaction Agreements with respect to which MI Corp. is contemplated thereby to be a party and to consummate the Transactions, subject to obtaining the MI Transaction Approval and the Additional Required Approvals. The execution and delivery of this Agreement and the Transaction Agreements with respect to which MI Corp. is contemplated thereby to be a party by MI Corp. and the consummation by MI Corp. of the Transactions have been duly authorized by all necessary corporate action on the part of MI Corp., subject to obtaining the MI Transaction Approval and the Additional Required Approvals. The only votes of the holders of capital stock (at any time any shares of such stock are held by any Person other than a member of the MI Group or the MVT Group) or other securities of MI Corp. necessary in connection with the Transactions are (A) the approval of the Share Issuance by the affirmative vote of a majority of the shares of MI Common Stock represented at the MI Shareholders Meeting and entitled to vote (provided that a majority of the shares of MI Common Stock entitled to vote is represented in person or by proxy at such meeting) and (B) the adoption of the MI Merger by the affirmative vote of a majority of the outstanding shares of MI Common Stock, it being understood and agreed that notwithstanding the foregoing, the parties hereto have agreed that the approval and adoption of this Agreement and the Transactions, including the Share Issuance and the MI Merger, by the MI Vote (the "MI Transaction Approval") is a condition to their respective obligations hereunder to effect the Transactions. This Agreement has been, and the Transaction Agreements with respect to which MI Corp. is contemplated thereby to be a party will be, duly executed and delivered by MI Corp. and, assuming the due authorization and valid execution and delivery of this Agreement and the Transaction Agreements with respect to which MI Corp. is contemplated thereby to be a party by the other parties hereto and thereto, as applicable (other than Merger Sub, MVT Holding and MVT Corp.), constitute or will constitute valid and binding agreements of MI Corp., enforceable against MI Corp. in accordance with their respective terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, fraudulent conveyance, moratorium and similar Applicable Laws relating to or affecting creditors generally or by general equity principles (regardless of whether such enforceability is considered in a proceeding in equity or at law).

(ii) Merger Sub has all requisite corporate power and authority to enter into this Agreement and the Transaction Agreements with respect to which Merger Sub is contemplated thereby to be a party and to consummate the Transactions, subject to the adoption of this Agreement by MVT Holding as the sole shareholder of Merger Sub. The execution and

delivery of this Agreement and the Transaction Agreements with respect to which Merger Sub is contemplated thereby to be a party by Merger Sub and the consummation by Merger Sub of the Transactions have been duly authorized by all necessary corporate action on the part of Merger Sub, subject to the adoption of this Agreement by MVT Holding as the sole shareholder of Merger Sub. This Agreement has been, and the Transaction Agreements with respect to which Merger Sub is contemplated thereby to be a party will be, duly executed and delivered by Merger Sub and, assuming the due authorization and valid execution and delivery of this Agreement or the applicable Transaction Agreement with respect to which Merger Sub is contemplated thereby to be a party by the other parties hereto and thereto (other than MI Corp., MVT Holding and MVT Corp.), as applicable, constitutes or will constitute a valid and binding agreement of Merger Sub, enforceable against Merger Sub in accordance with their respective terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, fraudulent conveyance, moratorium and similar Applicable Laws relating to or affecting creditors generally or by general equity principles (regardless of whether such enforceability is considered in a proceeding in equity or at law).

(iii) The execution and delivery by MI Corp. and Merger Sub of this Agreement does not, the execution and delivery by MI Corp. and Merger Sub of the Transaction Agreements with respect to which MI Corp. or Merger Sub, as the case may be, is contemplated thereby to be a party will not, and the consummation by MI Corp. or Merger Sub, as the case may be, of the Transactions will not result in a Violation (with or without notice or lapse of time, or both) under: (A) any provision of the articles of incorporation or by-laws of MI Corp. or Merger Sub or (B) except as, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on MVT Corp. or a Material Adverse Effect on MI Corp., subject to obtaining or making the MI Necessary Consents or the MVT Necessary Consents, (1) any Contract to which MI Corp., MVT Corp. or any of their respective Subsidiaries is a party or by which any of their respective properties or assets is bound or (2) any permit, concession, franchise, license, judgment, order, decree, statute, law, ordinance, rule or regulation applicable to MI Corp., MVT Corp. or any Subsidiary of MI Corp. or MVT Corp. or their respective properties or assets.

(iv) No consent, approval, waiver, order or authorization of, or registration, declaration or filing with, any Governmental Entity or any other Person is required to be obtained or made by or with respect to MI Corp. in connection with the execution and delivery of this Agreement and the Transaction Agreements with respect to which MI Corp. is contemplated thereby to be a party by MI Corp. or the consummation by MI Corp. of the Transactions, except for those required under or in relation to (A) the HSR Act, (B) state securities or "blue sky" laws or regulations, (C) the Securities Act, (D) the Exchange Act, (E) filings under the WBCL and the WLLCL to effect the transactions contemplated by Section 2.01 of the Separation Agreement and <u>Article III</u> of this Agreement, (F) the rules and regulations of the NYSE and, if different, such national securities exchange designated by MVT Holding and Investor pursuant to <u>Section 6.15</u>, (G) applicable state or federal laws or regulating broker-dealers, investment advisors and insurance companies, (H) antitrust or other competition laws of other jurisdictions, (I) applicable state or federal banking laws or regulations, (J) obtaining the MI Transaction Approval and the Additional Required Approvals and (K) such consents, waivers, approvals, orders,

authorizations, registrations, declarations and filings the failure of which to make or obtain, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on MVT Corp. or a Material Adverse Effect on MI Corp. Consents, approvals, orders, authorizations, registrations, declarations and filings required under or in relation to any of the foregoing <u>clauses (A)</u> through <u>(J)</u> or set forth in <u>Section 4.2(c)(iv)</u> of the MI Disclosure Schedule are referred to as the "<u>MI Necessary Consents</u>."

(v) The Board of Directors of MI Corp., by resolution duly adopted at a meeting duly called and held and not subsequently rescinded or modified in any way, has duly (i) determined that this Agreement and the Transactions are advisable and in the best interests of MI Corp. and its shareholders and (ii) resolved to recommend to the shareholders of MI Corp. that the shareholders of MI Corp. approve the Transactions. No "fair price," "business combination," "moratorium," "control share acquisition" or other form of anti-takeover statute or regulation under Wisconsin law or any anti-takeover provision in the articles of incorporation or by-laws of MI Corp. or other similar organizational documents of MI Corp. is applicable to the Transactions.

# (d) Reports and Financial Statements.

(i) All forms, reports and other documents filed by MI Corp. and its Subsidiaries with the SEC since January 1, 2005 (collectively, including all exhibits thereto, the "<u>MI SEC Reports</u>") at the time they were filed (or, if amended or superseded by a filing prior to the date of this Agreement, then on the date of such filing), complied in all material respects with the requirements of the Securities Act, the Exchange Act, the Sarbanes-Oxley Act of 2002, as amended (the "<u>Sarbanes Act</u>") and the NYSE and the rules and regulations promulgated thereunder, as applicable, and none of such MI SEC Reports contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.

(ii) Each of the principal executive officer of MI Corp. and the principal financial officer of MI Corp. (or each former principal executive officer of MI Corp. and each former principal financial officer of MI Corp., as applicable) has made all certifications required by Rule 13a-14 or 15d-14 under the Exchange Act and Sections 302 and 906 of the Sarbanes Act with respect to the MI SEC Reports and the statements contained in such certifications are true, complete and correct. For purposes of this <u>Section 4.2(d)</u>, "principal executive officer" and "principal financial officer" shall have the meanings given to such terms in the Sarbanes Act.

(iii) With respect to MVT Corp. and its Subsidiaries, MI Corp. and its Subsidiaries have designed and maintain a system of internal controls over financial reporting (as defined in Rules 13a-15(f) and 15d-15(f) of the Exchange Act) sufficient to provide reasonable assurances regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP. MI Corp. (A) has designed and maintains disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) of the Exchange Act) to ensure that material information required to be disclosed by MI Corp. (with respect to MVT Corp. and its Subsidiaries) in the reports that MI Corp. files or submits under the

Exchange Act is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms and is accumulated and communicated to MI Corp.'s management as appropriate to allow timely decisions regarding required disclosure and to make the certifications of the principal executive officer and principal financial officer of MI Corp. required under the Exchange Act with respect to such reports and (B) has disclosed, based on its most recent evaluation of such disclosure controls and procedures prior to the date hereof to its auditors and the audit committee of its Board of Directors (x) any significant deficiencies and material weaknesses in the design or operation of MI Corp.'s internal controls over financial reporting (with respect to members of the MVT Group) that are reasonably likely to adversely affect in any material respect MI Corp.'s ability to record, process, summarize and report financial information (with respect to members of the MVT Group) and (y) any fraud, whether or not material, that involves management or other employees of members of the MVT Group who have a significant role in MI Corp.'s internal controls over financial reporting. MI Corp. has made available to Investor all available minutes of the audit committee of MI Corp.'s Board of Directors from January 1, 2006 until the date hereof.

### (e) Information Supplied.

(i) None of the information supplied or to be supplied by MI Corp. for inclusion or incorporation by reference in (A) the Form S-4 will, at the time it becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, (B) the Proxy Statement will, on the date it is first mailed to MI Corp. shareholders and at the time of the MI Shareholders Meeting, contain any untrue statement of a material fact or omit to state any material fact or omit to state any material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading and (C) the Form 10 will, at the time it becomes effective, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading and (C) the Form 10 will, at the time it becomes effective, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

(ii) Notwithstanding the foregoing provisions of this <u>Section 4.2(e)</u>, no representation or warranty is made by MI Corp. with respect to statements made or incorporated by reference in the Form S-4, the Form 10 or the Proxy Statement based on information supplied by or on behalf of Investor for inclusion or incorporation by reference therein.

(f) <u>Brokers or Finders</u>. No agent, broker, investment banker, financial advisor or other firm or Person is or will be entitled to any broker's or finder's fee or any other similar commission or fee in connection with any of the Transactions based upon arrangements made by or on behalf of MI Corp. or any of its Subsidiaries, except J.P. Morgan Securities Inc., whose fees and expenses shall be paid in accordance with <u>Section 6.6</u>.

(g) <u>Valuation Letter of MI Corp. Financial Advisor</u>. MI Corp. has received the view of J.P. Morgan Securities Inc., as of the date hereof, with respect to the valuation range of MVT Corp. as of such date. The Purchase Price falls within the range of values indicated by J.P. Morgan Securities Inc.

# (h) <u>Taxes</u>.

(i) All Tax Returns required to be filed with respect to each of MI Corp. and its Subsidiaries have been timely filed, or requests for extensions to file such Tax Returns have been timely filed, granted and have not expired, and all such Tax Returns are complete and correct, except to the extent that such failures to file, to have extensions granted that remain in effect or to be complete or correct, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on MVT Corp. All material Taxes due with respect to MI Corp. and its Subsidiaries have been paid or accrued in accordance with GAAP. Since the date of the most recent MI Filed SEC Reports, no Tax liability with respect to MI Corp. and its Subsidiaries has been incurred outside the ordinary course of business or otherwise inconsistent with past custom and practice. All material Tax Returns required to be filed by each of MVT Corp. and its Subsidiaries have been timely filed, or requests for extensions to file such Tax Returns have been timely filed, granted and have not expired, and all such Tax Returns are complete and correct in all material respects.

(ii) No deficiencies for any Taxes have been proposed, asserted or assessed in writing in respect of or against MI Corp. or any of its Subsidiaries that are not adequately reserved for in the financial statements of MI Corp. included in the MI Filed SEC Reports (and in the case of MVT Corp., the MVT Financial Statements), except for deficiencies that, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on MVT Corp. No material deficiencies for any Taxes have been proposed, asserted or assessed in writing against MVT Corp. or any of its Subsidiaries that are not adequately reserved for in the MVT Financial Statements. The applicable statutes of limitations have expired for all Tax periods through 2001 for all material Tax Returns of MI Corp. and each of its Subsidiaries. Since January 1, 2001, no written claim has been made to MI Corp. or its Subsidiaries by a Governmental Entity in a jurisdiction where MVT Corp. or any of its Subsidiaries is or may be subject to a material Tax liability in that jurisdiction.

(iii) None of MI Corp. or its Subsidiaries has taken any action, and to the Knowledge of MI Corp. there is no fact, agreement, plan or other circumstance, that could reasonably be expected to prevent (A) the MI Merger and the MI Conversion from together qualifying as a reorganization within the meaning of Section 368(a) of the Code or (B) the MI Contribution and the Share Distribution from qualifying as a reorganization under Section 368(a)(1)(D) of the Code and a distribution eligible for nonrecognition under Sections 355(a) and 361(c) of the Code.

(iv) None of MI Corp. or any of its Subsidiaries is a party to any Tax sharing or Tax indemnity agreements entered into after January 1, 2001 (other than the Tax Allocation Agreement and any agreements between or among MVT Corp. and its Subsidiaries) that could reasonably be expected to result in a material Tax liability to MVT Corp. or any of its Subsidiaries.

(v) Other than with respect to the Share Distribution, within the past five years, none of MI Corp. or any of its Subsidiaries has been a "distributing corporation" or a "controlled corporation" in a distribution intended to qualify under Section 355(a) of the Code.

(vi) None of MVT Corp. or any of its Subsidiaries is obligated to make any payments, or is a party to any Contract or MVT Plan that could obligate it to make any payments, that would not be deductible by reason of Section 162(m) or Section 280G of the Code.

(vii) None of MVT Corp. or any of its Subsidiaries has agreed to make, or is required to make, any material adjustment affecting any open taxable year or period under Section 481(a) of the Code or any similar provision of state, local or foreign law by reason of a change in accounting methods or otherwise.

(viii) Since January 1, 2001, none of MVT Corp. or any of its Subsidiaries has been required to reallocate, or is subject to an Internal Revenue Service ("<u>IRS</u>") or other Governmental Entity challenge that would require the reallocation of, gross income, deductions, credits or allowances, or of any item or element affecting taxable income, by reason of Section 482 of the Code.

(ix) Other than pursuant to the Tax Allocation Agreement, neither MVT Corp. nor any of its Subsidiaries has any material liability under Treasury Regulation Section 1.1502-6 (or any comparable or similar provision of federal, state, local or foreign Applicable Laws), as a transferee or successor, pursuant to any contractual obligation, or otherwise for any Taxes of any person other than MVT Corp. or any of its Subsidiaries.

(x) Neither MVT Corp., nor any of its Subsidiaries has engaged in one of the types of transactions the IRS has determined to be a tax avoidance transaction and identified by notice, regulation, or other form of published guidance as a "listed transaction," as set forth in Treasury Regulation Section 1.6011-4(b)(2).

(i) <u>Permits; Compliance with Laws</u>. Members of the MI Group (A) hold all permits, licenses, franchises, variances, exemptions, orders and approvals of all Governmental Entities which are necessary for the operation of the business of the MI Group as currently conducted (the "<u>MI Permits</u>"), and no suspension or cancellation of any of the MI Permits is pending or, to the Knowledge of MI Corp., threatened and (B) are in compliance in all respects with the terms of the MI Permits, except in each case where the failure to hold such MI Permits or the suspension or cancellation of any such MI Permits would not reasonably be expected to have a Material Adverse Effect on MI Corp. or a Material Adverse Effect on MVT Corp. None of the members of the MI Group is in violation of, and MI Corp. and its Subsidiaries have not received since January 1, 2005 any written notices of violations with respect to, any Applicable Laws, except where such violations would not reasonably be expected to have a Material Adverse Effect on MI Corp. or a Material Adverse Effect on MV Corp.

Section 4.3 <u>Representations and Warranties of MI Corp.</u>, <u>MVT Holding and MVT Corp.</u> Except as set forth in the MVT Disclosure Schedule delivered by MVT Corp. to Investor prior to the execution of this Agreement (the "<u>MVT Disclosure Schedule</u>") (each section of which, to the extent specified therein, qualifies the correspondingly numbered representation and warranty of MVT Corp. contained herein and any disclosure in such section

qualifies any other representation and warranty of MI Corp. or MVT Corp. contained herein to which its application or relevance is reasonably apparent on its face), each of MI Corp., MVT Holding and MVT Corp. represent and warrant to Investor as follows:

### (a) Organization, Standing and Power; Subsidiaries.

(i) MVT Corp. is a corporation duly incorporated, validly existing and in active status under the laws of State of Wisconsin. Each of the Subsidiaries of MVT Corp. is a corporation or other organization duly organized, validly existing and in good standing or active status (where applicable) under the laws of its jurisdiction of incorporation or organization, and MVT Corp. and each of its Subsidiaries has the requisite power and authority to own, lease and operate its properties and to carry on the MVT Business as now being conducted and as it will be conducted through the Closing Date, except where the failure to be so organized, existing and in good standing or to have such power and authority, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on MVT Corp., and each of MVT Corp. and its Subsidiaries is duly qualified and in good standing to do business in each jurisdiction in which the nature of the MVT Business or the ownership or leasing of its properties makes such qualification necessary, in each case other than in such jurisdictions where the failure to so qualify or to be in good standing, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on MVT Corp. True, correct and complete copies of the articles of incorporation and by-laws of MVT Corp. in effect on the date hereof have been delivered or made available to Investor.

(ii) <u>Section 4.3(a)(ii)</u> of the MVT Disclosure Schedule sets forth a list of all the Subsidiaries of MVT Corp. which as of the date of this Agreement would be Significant Subsidiaries of MVT Corp. (as defined in Rule 1-02(w) of Regulation S-X of the SEC) if the Share Distribution had occurred immediately prior to the date hereof (the "<u>MVT Significant Subsidiaries</u>"). All the outstanding shares of capital stock of, or other equity interests in, each such MVT Significant Subsidiary have been validly issued and are fully paid and nonassessable (except, in the case of any Significant Subsidiaries that are Wisconsin corporations or foreign corporations that are qualified to do business in Wisconsin, subject to the personal liability which may be imposed on shareholders by former Section 180.0622(2)(b) of the WBCL for debts incurred prior to June 14, 2006 (for debts incurred on or after such date, Section 180.0622(2)(b) of the WBCL has been repealed) owing to employees for services performed, but not exceeding six months' service in any one case) and are owned directly or indirectly by MVT Corp., free and clear of all material Liens and free of any other material restriction (including any restriction on the right to vote, sell or otherwise dispose of such capital stock or other equity interests, but excluding restrictions under the Securities Act). None of MVT Corp. or any MVT Significant Subsidiaries directly or indirectly owns any equity or similar interest in, or any interest convertible into or exchangeable or exercisable for any equity or similar interest in, any corporation, partnership, joint venture or other business association or entity (other than Subsidiaries of MVT Corp.) that is or would reasonably be expected to be material to MVT Corp. and its Subsidiaries taken as a whole.

(b) Capital Structure.

(i) The authorized capital stock of MVT Corp. consists of 1,000 shares of MVT Common Stock, all of which are, as of the date hereof, and at all times prior to the MVT Distribution Time will be, owned of record and beneficially by a member of the MI Group or the MVT Group free and clear of any Liens. There are no other shares of capital stock or other equity securities of MVT Corp. that are authorized or outstanding. Immediately following the MVT Distribution all of the outstanding shares of MVT Common Stock will be owned of record and beneficially by MVT Holding, and no shares of MVT Common Stock will be held by MVT Corp. in its treasury. All issued and outstanding shares of MVT Common Stock are duly authorized, validly issued, fully paid and nonassessable (subject to the personal liability which may be imposed on shareholders by former Section 180.0622(2)(b) of the WBCL for debts incurred prior to June 14, 2006 (for debts incurred on or after such date, Section 180.0622(2)(b) of the WBCL has been repealed) owing to employees for services performed, but not exceeding six months' service in any one case), and the shares of MVT Common Stock are not entitled to preemptive rights.

(ii) No bonds, debentures, notes or other indebtedness of MVT Corp. having the right to vote (or convertible into or exchangeable for securities having the right to vote) on any matters on which shareholders of MVT Corp. may vote ("<u>MVT Voting Debt</u>") are issued or outstanding.

(iii) Except as contemplated by this Agreement and the Transaction Agreements, there are no securities, options, warrants, calls, rights, commitments, agreements, arrangements or undertakings of any kind to which MVT Corp. or any of its Subsidiaries is a party or by which any of them is bound obligating MVT Corp. or any of its Subsidiaries to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of capital stock of MVT Corp. or any of its Subsidiaries, MVT Voting Debt, MVT Common Stock or other voting securities of MVT Corp. or any of its Subsidiaries to issue, grant, extend or enter into any such security, option, warrant, call, right, commitment, agreement, arrangement or undertaking. There are no, except as provided for in or permitted by this Agreement or the Transaction Agreements, outstanding obligations of MVT Corp. or any of its Subsidiaries to repurchase, redeem or otherwise acquire any shares of capital stock of MVT Corp. or any of its Subsidiaries.

(iv) Other than the Contracts contemplated to effect the Transactions, there are no shareholder agreements, voting trusts or other Contracts to which MVT Corp. is a party or by which it is bound relating to the voting or transfer of any shares of capital stock of MVT Corp.

# (c) Authority; No Conflicts.

(i) MVT Corp. has all requisite corporate power and authority to enter into this Agreement and the Transaction Agreements with respect to which MVT Corp. is contemplated thereby to be a party and to consummate the Transactions, subject to the MI Transaction Approval and the Additional Required Approvals. The execution and delivery of this Agreement and the Transaction Agreements with respect to which MVT Corp. is contemplated thereby to be a party by MVT Corp. and the consummation by MVT Corp. of the Transactions have been duly authorized by all necessary corporate action on the part of MVT

Corp., subject to the MI Transaction Approval and the Additional Required Approvals. This Agreement has been, and the Transaction Agreements with respect to which MVT Corp. is contemplated thereby to be a party will be, duly executed and delivered by MVT Corp. and, assuming the due authorization and valid execution and delivery of this Agreement or the applicable Transaction Agreements with respect to which MVT Corp. is contemplated thereby to be a party by the other parties hereto and thereto, as applicable (other than MI Corp., Merger Sub and MVT Holding), constitute or will constitute valid and binding agreements of MVT Corp., enforceable against MVT Corp. in accordance with their respective terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, fraudulent conveyance, moratorium and similar Applicable Laws relating to or affecting creditors generally or by general equity principles (regardless of whether such enforceability is considered in a proceeding in equity or at law).

(ii) The execution and delivery by MVT Corp. of this Agreement does not, the execution and delivery by MVT Corp. of the Transaction Agreements with respect to which MVT Corp. is contemplated thereby to be a party will not, and the consummation by MVT Corp. of the Transactions will not result in a Violation (with or without notice or lapse of time, or both) under: (A) any provision of the articles of incorporation or by-laws or similar organizational documents of MVT Corp. or any Subsidiary of MVT Corp. or (B) except as, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on MVT Corp., subject to obtaining or making the MVT Necessary Consents, (1) any Contract to which MVT Corp. or any of its Subsidiaries is a party or by which any of them or any of their respective properties or assets is bound or (2) any permit, concession, franchise, license, judgment, order, decree, statute, law, ordinance, rule or regulation applicable to MVT Corp. or any Subsidiary of MVT Corp. or their respective properties or assets.

(iii) The Board of Directors of MVT Corp., by resolutions duly adopted by unanimous written consent and not subsequently rescinded or modified in any way has duly (A) determined that this Agreement, the Transactions and the Transaction Agreements with respect to which MVT Corp. is contemplated thereby to be a party are advisable and in the best interests of MVT Corp. and its shareholder and (B) approved this Agreement, the Transactions contemplated hereby and by the Transaction Agreements.

(iv) No consent, approval, waiver, order or authorization of, or registration, declaration or filing with, any Governmental Entity or any other Person is required to be obtained or made by or with respect to MVT Corp. or any Subsidiary of MVT Corp. in connection with the execution and delivery by MVT Corp. of this Agreement and the Transaction Agreements with respect to which MVT Corp. is contemplated thereby to be a party or the consummation by MVT Corp. of the Transactions, except for those required under or in relation to (A) the HSR Act, (B) state securities or "blue sky" laws or regulations, (C) the Securities Act, (D) the Exchange Act, (E) the rules and regulations of the NYSE and, if different, such national securities exchange designated by MVT Holding and Investor pursuant to <u>Section 6.15</u>, (F) antitrust or other competition laws of other jurisdictions, (G) applicable state or federal laws or regulations and the rules and regulations of the NASD or any other applicable regulatory authority regulating broker dealers, investment advisors and insurance companies, (H) applicable state or federal banking laws or regulations, (I) obtaining the MI Transaction Approval and the Additional Required Approvals, and (J) such consents, approvals, waivers, orders,

authorizations, registrations, declarations and filings the failure of which to make or obtain, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on MVT Corp. Consents, waivers, approvals, orders, authorizations, registrations, declarations and filings required under or in relation to any of the foregoing <u>clauses (A)</u> through <u>(I)</u> or set forth in <u>Section 4.3(c)(iv)</u> of the MVT Disclosure Schedule are referred to as the "<u>MVT Necessary Consents</u>."

### (d) Reports and Financial Statements; No Undisclosed Liabilities

(i) As of the date hereof, neither MVT Corp. nor any of its Subsidiaries is required to file any form, report or other document with the SEC. Included in <u>Section 4.3(d)(i)</u> of the MVT Disclosure Schedule are the consolidated balance sheets of MVT Corp. and its Subsidiaries as of December 31, 2006 and 2005, and the related consolidated statements of income, cash flows and shareholders' equity for the fiscal years ended December 31, 2006 and 2005, in each case audited by MVT Corp.'s independent public accountants, whose report thereon is included therewith (such statements, together with the notes thereto, the "<u>MVT Financial Statements</u>"). The MVT Financial Statements (including the related notes thereto) fairly present, in all material respects, the consolidated financial position and consolidated results of operations and cash flows of MVT Corp. and its consolidated Subsidiaries as of the respective dates or for the respective periods set forth therein, all in conformity with GAAP consistently applied during the periods covered thereby except as otherwise noted therein.

(ii) Each of the principal executive officer of MVT Corp. and the principal financial officer of MVT Corp. have made back-up certifications with respect to the MVT Business to the principal executive officer of MI Corp. and the principal financial officer of MI Corp. necessary to allow such officers of MI Corp. to make all certifications required by Rule 13a 14 or 15d 14 under the Exchange Act and Sections 302 and 906 of the Sarbanes Act with respect to the Annual Report on Form 10-K of MI Corp. for the year ended December 31, 2006. For purposes of this <u>Section 4.3(d)</u>, "principal executive officer" and "principal financial officer" shall have the meanings given to such terms in the Sarbanes Act and applying such terms to MVT Corp. as though it were a separate reporting company under the Exchange Act.

(iii) Except as disclosed in the MI SEC Reports filed after January 1, 2007 and publicly available prior to the date hereof, other than the text of any "risk factors" and the text of any disclaimers with respect to forward-looking statements (such MI SEC Reports, other than such "risk factors" and disclaimers, the "<u>MI Filed SEC Reports</u>") or as set forth in the MVT Financial Statements, as of the date hereof none of MVT Corp. nor any of its Subsidiaries have any liabilities or obligations of any nature (whether absolute, accrued, indirect, direct or contingent) other than (A) liabilities or obligations incurred in the ordinary course of business or (B) liabilities or obligations that, individually or in the aggregate, do not exceed \$15 million.

## (e) Information Supplied.

(i) None of the information supplied or to be supplied by MVT Corp. specifically for inclusion or incorporation by reference in (A) the Form S-4 will, at the time it becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements

therein, in light of the circumstances under which they were made, not misleading, (B) the Proxy Statement will, on the date it is first mailed to MI Corp. shareholders and at the time of the MI Shareholders Meeting respectively, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. and (C) the Form 10 will, at the time the Form 10 becomes effective contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

(ii) Notwithstanding the foregoing provisions of this <u>Section 4.3(e)</u>, no representation or warranty is made by MVT Corp. with respect to statements made or incorporated by reference in the Form S-4, the Form 10 or the Proxy Statement based on information supplied by or on behalf of Investor for inclusion or incorporation by reference therein.

### (f) Litigation; Compliance with Laws.

(i) Except as set forth in the MI Filed SEC Reports, or in the MVT Financial Statements, there is no Action pending or, to the Knowledge of MVT Corp., threatened against MI Corp. or any Subsidiary of MI Corp. or any property or asset of MI Corp. or any Subsidiary of MI Corp. which, individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect on MVT Corp., nor is there any material judgment, decree, injunction, rule or order of any Governmental Entity or arbitrator outstanding against MI Corp. or any of its Subsidiaries with respect to the MVT Business.

(ii) Members of the MVT Group hold all material permits, licenses, franchises, variances, exemptions, orders and approvals of all Governmental Entities which are necessary for the operation of the MVT Business (the "<u>MVT Permits</u>"), and no suspension or cancellation of any of the MVT Permits is pending or, to the Knowledge of MVT Corp., threatened. Members of the MVT Group are in compliance in all material respects with the terms of the MVT Permits. None of MI Corp. or any of its Subsidiaries is, in any material respect, in violation of, and MI Corp. and its Subsidiaries have not received since January 1, 2005 any written notices of material violations with respect to, any Applicable Laws.

(g) <u>Absence of Certain Changes or Events</u>. Except (i) as specifically contemplated or permitted by this Agreement or the Transaction Agreements or (ii) as set forth in the MI Filed SEC Reports or in the MVT Financial Statements, since December 31, 2006 through the date hereof, (A) members of the MVT Group have conducted their business only in the ordinary course, consistent with past practice, and (B) there has not been any event, change, circumstance or development (including any damage, destruction or loss whether or not covered by insurance) which, individually or in the aggregate, has had, or would reasonably be expected to have, a Material Adverse Effect on MVT Corp. From December 31, 2006 through the date of this Agreement, except as set forth in the MI Filed SEC Reports or in the MVT Financial Statements, none of MI Corp. or any of its Subsidiaries has taken any action that, if taken without

the consent of Investor during the period from the date of this Agreement through the Closing Date, would constitute a breach of <u>Sections 5.1(e)</u>, <u>5.1(h)</u>, <u>5.1(j)</u> or <u>5.1(k)</u>.

(h) Environmental Matters. Except as, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on MVT Corp.: (i) the operations of members of the MVT Group have been and are in compliance with all applicable Environmental Laws and with all MVT Permits required by applicable Environmental Laws; (ii) there are no pending or, to the Knowledge of MVT Corp., threatened, Actions under or pursuant to Environmental Laws against MVT Corp., or other members of the MVT Group, or, to the Knowledge of MVT Corp., any other Person whose Environmental Liabilities or any member of the MVT Group has or may have retained or assumed by contract or operation of law, or involving any real property currently or, to the Knowledge of MVT Corp., Persons whose Environmental Liabilities any member of the MVT Group has or may have retained or assumed by contract or operation of law are not subject to any Environmental Liabilities and, to the Knowledge of MVT Corp., there are no facts, circumstances or conditions (including the presence, release or threatened release of Hazardous Materials at any location whether or not owned or operated by any member of the MVT Group) which would reasonably be expected to result in Environmental Liabilities for any member of the MVT Group, or, to the Knowledge of MVT Corp., any other Person whose Environmental Liabilities any member of the MVT Group has or may have retained or assumed by contract or operation of law are not subject to any Environmental Liabilities and, to the Knowledge of MVT Corp., there are no facts, circumstances or conditions (including the presence, release or threatened release of Hazardous Materials at any location whether or not owned or operated by any member of the MVT Group) which would reasonably be expected to result in Environmental Liabilities for any member of the MVT Group, or, to the Knowledge of MVT Corp., any other Person whose Environmental Liabilities any member of the MVT Group has or may have retained or assumed by contrac

As used in this Agreement, "Environmental Laws" means any and all foreign, federal, state, local or municipal laws, rules, orders, regulations, statutes, ordinances, codes, decisions, injunctions, orders, decrees, requirements of any Governmental Entity, any and all common law requirements, rules and bases of liability regulating, relating to or imposing liability or standards of conduct, in each case, concerning pollution, Hazardous Materials (as defined below) or protection of human health or the environment, and includes the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. Section 9601 <u>et seq</u>., the Hazardous Materials Transportation Act, 49 U.S.C. Section 1801 <u>et seq</u>., the Resource Conservation and Recovery Act, 42 U.S.C. Section 6901 <u>et seq</u>., the Clean Water Act, 33 U.S.C. Section 1251 <u>et seq</u>., the Clean Air Act, 33 U.S.C. Section 2601 <u>et seq</u>., the Toxic Substances Control Act, 15 U.S.C. Section 2601 <u>et seq</u>., the Occupational Safety and Health Act, 29 U.S.C. Section 651 <u>et seq</u>. (but solely as it relates to the exposure of Hazardous Materials) and the Oil Pollution Act of 1990, 33 U.S.C. Section 2701 <u>et seq</u>., as such laws have been amended or supplemented, and the regulations promulgated pursuant thereto, and all analogous state or local statutes. As used in this Agreement, "<u>Environmental Liabilities</u>" with respect to any Person means any and all liabilities of or relating to such Person or any of its Subsidiaries (including any entity which is, in whole or in part, a predecessor of such Person or any of such Subsidiaries), whether vested or unvested, contingent or fixed, actual or potential, known or unknown, which (i) arise under or relate to matters covered or regulated by, or for which liability is imposed under, Environmental Laws and (ii) relate to actions occurring or conditions existing on or prior to the Closing Date. As used in this Agreement, "<u>Hazardous Materials</u>" means all substances defined in, regulated under or for which liability is imposed by

Oils, Pollutants or Contaminants as defined in the National Oil and Hazardous Substances Pollution Contingency Plan, 40 C.F.R. § 300.5., asbestos, mold, polychlorinated biphenyls and radioactive materials.

(i) <u>Intellectual Property</u>. Except as set forth in the MI Filed SEC Reports and except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on MVT Corp.: (i) the members of the MVT Group own, or are licensed to use, all trademarks, service marks, trade dress, logos, trade names, service names, corporate names, domain names, brand identifiers, copyrights, proprietary designs, Computer Software, databases, trade secrets, know-how, patents, patent applications, inventions and similar rights ("<u>Intellectual Property</u>") used in the conduct of the MVT Business as currently conducted; (ii) to the Knowledge of MVT Corp., all Intellectual Property owned by the members of the MVT Group is valid and enforceable and the use of any Intellectual Property by the members of the MVT Group does not infringe upon, misappropriate or otherwise violate the Intellectual Property rights of any Person; (iii) to the Knowledge of MVT Corp., the use by the members of the MVT Group of Intellectual Property which is licensed to any member of the MVT Group acquired the right to use such Intellectual Property; (iv) to the Knowledge of MVT Corp., no Person is infringing, misappropriating or otherwise violating any right of any member of the MVT Group with respect to any Intellectual Property owned by and/or exclusively licensed to any member of the MVT Group; (v) there is no claim or proceeding pending or, to the Knowledge of MVT Corp., no Intellectual Property owned by any member of the MVT Group is being used by or enforced by any member of the MVT Group in a manner that would reasonably be expected to result in the abandonment, cancellation or unenforceability of such Intellectual Property.

(j) <u>Title to Properties; Assets/Services</u>. Each of MVT Corp. and its Subsidiaries has good and valid title to, or, in the case of leased properties and assets, valid leasehold interests in, all of its tangible properties and assets, except where the failure to have such good and valid title, or valid leasehold interest, would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on MVT Corp. The tangible property and assets of MVT Group, together with the services to be provided by the MI Group to the MVT Group pursuant to the Transaction Agreements, constitute (i) all of the tangible property and assets necessary to permit MVT Group to carry on the MVT Business after the Closing Date in all material respects as conducted by the MVT Group prior thereto and (ii) as of the date hereof, all of the tangible property and assets primarily used by the MVT Group in carrying on the MVT Business as of the date hereof.

# (k) Certain Contracts.

(i) Except as filed as exhibits to the MI Filed SEC Documents as of the date hereof, neither MVT Corp. nor any of its Subsidiaries is a party to or bound by any

Contract that (A) is a "material contract" (as such term is defined in Item 601(b)(10) of Regulation S-K promulgated by the SEC and applying such term to MVT Corp. as though it were a separate reporting company under the Exchange Act for purposes of this Section 4.3(k)), (B) limits or purports to limit the ability of MVT Corp. or any of its Subsidiaries to incur indebtedness or pay dividends, (C) is a loan agreement, credit agreement, note, bond, mortgage, indenture or other agreement or instrument pursuant to which any indebtedness of MVT Corp. or any of its Subsidiaries in an aggregate principal amount in excess of \$10 million is outstanding or may be incurred. (D) is a limited liability company agreement, partnership agreement, joint venture agreement or other similar agreement relating to any partnership or joint venture that is material to the MVT Business other than any such limited liability company agreement, partnership agreement, joint venture agreement or other similar agreement with respect to a wholly-owned Subsidiary of MVT Corp., (E) limits in any material respect the right of MVT Corp. or its Subsidiaries to engage or compete in any line of business that is material to MVT Corp. and its Subsidiaries as a whole, (F) contains "most favored nation" pricing provisions, that would reasonably be expected to be material to MVT Corp. or its Subsidiaries, (G) grants any exclusive rights, rights of first refusal, rights of first negotiation or similar rights to any person, in each case in any respect material to the business of MVT Corp. and its Subsidiaries, taken as a whole, (H) provides for the acquisition or disposition (pending as of the date hereof), directly or indirectly (by merger or otherwise), of assets or capital stock or other equity interests of another person for aggregate consideration under such Contract in excess of \$50 million, (I) is an acquisition or disposition Contract pursuant to which MVT Corp. or its Subsidiaries has continuing indemnification, "earn-out" or other contingent payment obligations, in each case, that could reasonably be expected to result in payments by MVT Corp. in excess of \$10 million, (J) is a Contract to provide services to any of MVT Corp.'s twenty largest customers, measured by net revenues during the fiscal year ended December 31, 2006 (the "Customer Contracts"), (K) calls for or could reasonably be expected to require aggregate payments by MVT Corp. or its Subsidiaries in excess of \$30 million over the remaining term of such Contract, or in excess of \$10 million in any twelve month period, or (L) is a Contract between or among any member or members of the MVT Group, on the one and, and any member or members of the MI Group, on the other hand, that is material to the business of the MVT Group as a whole. Each Contract of the type described in this Section 4.3(k)(i), whether or not listed in Section 4.3(k) of the MVT Disclosure Schedule, is referred to herein as a "MVT Material Contract."

(ii) Each MVT Material Contract is valid and in full force and effect (except those which are cancelled, rescinded or terminated after the date hereof in accordance with their terms), except where the failure to be in full force and effect would not, individually or in the aggregate, have a Material Adverse Effect on MVT Corp. To the Knowledge of MVT Corp., no Person is challenging the validity or enforceability of any MVT Material Contract, except such challenges which would not, individually or in the aggregate, have a Material Adverse Effect on MVT Corp. Neither MVT Corp. nor any of its Subsidiaries has Knowledge of, or has received written notice of, any violation of or default under (or any condition which with the passage of time or the giving of notice would cause such a violation of or default under) any MVT Material Contract or any other Contract to which it is a party or by which it or any of its properties or assets is bound, except for violations or defaults that would not, individually or in the aggregate, neither MVT Corp. nor any of its Subsidiaries has Knowledge of, or has received written notice of, neither MVT Corp. nor any of its Subsidiaries or assets is bound, except for violations or defaults that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on MVT Corp. As of the date hereof, neither MVT Corp. nor any of its Subsidiaries has Knowledge of, or has received written notice of, any customer's intention not to renew any Customer Contract. True and

complete copies of all MVT Material Contracts have been delivered or made available to Investor, which copies, in the case of Customer Contracts, may have been redacted to exclude confidential or competitively sensitive information.

## (l) Employee Benefits.

(i) Section 4.3(1)(i) of the MVT Disclosure Schedule contains a true and complete list of each MVT Plan. As of the date hereof, with respect to each MVT Plan, MVT has made available or delivered to Investor a true, correct and complete copy of such plan and the current summary plan description and any material modifications thereto, if any, and, if such plan will be maintained by the MVT Group following the Closing: (A) all related trust agreements, and insurance contracts and other funding vehicles; (B) the two most recent Annual Reports (Form 5500 Series) and accompanying schedules and exhibits, if any; (C) the two most recent annual financial reports, if any; (D) the two most recent actuarial reports, if any; and (E) the most recent determination letter from the IRS, if any. With respect to each MVT Plan, MVT Corp. and its ERISA Affiliates and other Subsidiaries have complied in all material respects with, and are now in compliance in all material respects with, to the extent applicable, all provisions of ERISA, the Code and all other Applicable Laws and regulations. Each MVT Plan has been operated and administered, in all material respects, in accordance with its terms. Except as set forth in Section 4.3(1)(i) of the MVT Disclosure Schedule, no MVT Plan is (or was, during the past six years) subject to Title IV or Section 302 of ERISA. No material liability under Title IV or Section 302 of ERISA has been incurred by MVT Corp. or any of its ERISA Affiliates or other member of the MVT Group that has not been satisfied in full, and no condition exists that presents a material risk to MVT Corp. or any of its ERISA Affiliates or other member of the MVT Group of incurring any such liability, other than liability for premiums due the Pension Benefit Guaranty Corporation. All premiums due the Pension Benefit Guaranty Corporation have been fully paid on a timely basis. No MVT Plan provides for post-employment welfare benefits, except as required under Applicable Laws or specified in Section 4.3(l)(i) of the MVT Disclosure Schedule. MI Corp. and its Affiliates have reserved the right to amend, terminate or modify at any time all plans or arrangements providing for retiree health or life insurance coverage, and with respect to any plan or arrangement that does not provide such benefits, there has not been, to the Knowledge of MVT Corp., any communication by MI Corp. or its Subsidiaries to MVT Employees which could reasonably be interpreted to promise or guarantee retiree health or life insurance or other retiree death benefits on a permanent basis. All MVT Plans subject to the Applicable Laws of any jurisdiction outside of the United States (1) have been maintained in all material respects in accordance with all applicable requirements, (2) if they are intended to qualify for special Tax treatment meet all requirements for such treatment in all material respects, and (3) if they are intended to be funded and/or book-reserved are fully funded and/or book-reserved, as appropriate, based upon reasonable actuarial assumptions.

(ii) As of the date hereof, each MVT Plan that is intended to be "qualified" within the meaning of Section 401(a) of the Code, and the trust maintained pursuant thereto, has been determined to be so qualified and exempt from federal income Taxation under Section 501 of the Code by the IRS, and to the Knowledge of MVT Corp. and MI Corp., nothing has occurred with respect to the operation of any such MVT Plan that would reasonably be expected to cause the loss of such qualification of exemption from Tax.

(iii) As of the date hereof, no member of the MVT Group has any material liability under or obligation to any Multiemployer Plan and no MVT Plan is a Multiemployer Plan. The MVT Plans provide benefits only to current and former employees of the MVT Group.

(iv) Except as contemplated by the Employee Matters Agreement or this Agreement, the consummation of the Transactions (either by themselves or in combination with another event) do not (A) entitle any current or former employee, director or officer of MVT Corp. or any of its Subsidiaries to any severance pay, unemployment compensation or any other payment or (B) accelerate the time of payment or vesting of benefits, or materially increase the amount of compensation, due any such current or former employee, director or officer of MVT Corp. or any of its Subsidiaries.

(v) MI Corp., MVT Corp. and their Subsidiaries and their ERISA Affiliates have paid all amounts that they are required to pay as contributions to the MVT Plans as of the last day of the most recent fiscal year of each of the MVT Plans; all benefits accrued under any funded or unfunded MVT Plan have been paid, accrued or otherwise adequately reserved for in accordance with GAAP as of December 31, 2006; and all monies withheld from employee paychecks with respect to MVT Plans have been or will be transferred to the appropriate MVT Plan in a timely manner as required by Applicable Law.

(m) <u>Labor Relations</u>. On the date hereof, neither MI Corp. or any of its Subsidiaries (in each case, with respect to any Persons employed by or otherwise performing services primarily for MVT Corp. or any of its Subsidiaries (the "<u>MVT Business Personnel</u>")), MVT Corp. nor any of its Subsidiaries is a party to any material collective bargaining Contract, work rules or practices, or any other labor-related agreements or arrangements with any labor union, labor organization or works council. As of the date hereof, (i) no labor organization or group of MVT Business Personnel has made a material pending demand for recognition or certification, (ii) there are no material representation or certification proceedings or petitions seeking a representation proceeding presently pending or, to the Knowledge of MVT Corp., threatened to be brought or filed, with the National Labor Relations Board or any other domestic or foreign labor relations tribunal or authority, and (iii) to the Knowledge of MVT Corp., there are no labor union organizing activities with respect to any MVT Business Personnel. Neither MI Corp., MVT Corp. nor any of their Subsidiaries has engaged in any material unfair labor practice with respect to any MVT Business Personnel, and there is no material unfair labor practice complaint or grievance against MI Corp., MVT Corp. or any of their Subsidiaries by the National Labor Relations Board or any comparable state agency pending or threatened in writing with respect to MVT Business Personnel. Between January 1, 2005 and the date hereof, there have been no actual, or to the Knowledge of MVT Corp., material threatened, strikes, work stoppages, slowdowns, lockouts, arbitrations, grievances or other labor disputes involving MVT Business Personnel.

(n) <u>Insurance</u>. As of the date hereof, MI Corp. maintains with respect to MVT Corp. insurance coverage with reputable insurers in such amounts and covering such risks as are in accordance with normal industry practice for companies engaged in

businesses similar to that of MVT Corp. (taking into account the cost and availability of such insurance). Section 4.3(n) of the MVT Disclosure Schedule sets forth, as of the date hereof, a complete and correct list of all material insurance policies owned or held by MI Corp. or its Subsidiaries with respect to MVT Corp. or its Subsidiaries. As of the date hereof, with respect to each such insurance policy, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on MVT Corp., (i) the policy is legal, valid, binding and enforceable in accordance with its terms and, except for policies that have expired under their terms in the ordinary course, is in full force and effect; (ii) neither MI Corp., MVT Corp. nor any Subsidiary is in material breach or default (including any such breach or default with respect to the payment of premiums or the giving of notice), and, to MI Corp.'s Knowledge, no event has occurred which, with notice or the lapse of time, would constitute such a breach or default, or permit termination or modification, under the policy; (iii) to the Knowledge of MI Corp., no insurer on the policy has been declared insolvent or placed in receivership, conservatorship or liquidation; and (iv) to the Knowledge of MI Corp., no notice of cancellation or termination has been received other than in connection with ordinary renewals.

(o) <u>Liens</u>. No Liens exist on any assets of any member of the MVT Group, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on MVT Corp.

## (p) Real Property.

(i) <u>Section 4.3(p)</u> of the MVT Disclosure Schedule contains as of the date hereof a list of all real property relating to the MVT Business that is (A) owned by MVT Corp. or its Subsidiaries or (B) leased or subleased by MVT Corp. or its Subsidiaries, as lessee or sublessee, except for such leases or subleases that provide for payments of less than \$50,000 in the aggregate in any given year (the "<u>MVT Leases</u>").

(ii) Each MVT Lease is valid and in full force and effect (except those which are cancelled, rescinded or terminated after the date hereof in accordance with their terms), except where the failure to be in full force and effect would not, individually or in the aggregate, have a Material Adverse Effect on MVT Corp. Neither MVT Corp. nor any of its Subsidiaries has Knowledge of, or has received written notice of, any violation of or default under (or any condition which with the passage of time or the giving of notice would cause such a violation of or default under) any MVT Lease, except for violations or defaults that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on MVT Corp. True and complete copies of all MVT Leases have been delivered or made available to Investor.

Section 4.4 <u>Representations and Warranties of MI Corp. and MVT Holding</u>. Each of MVT Holding and MI Corp. represent and warrant to Investor as follows:

(a) <u>Organization</u>. MVT Holding is a corporation duly organized, validly existing and in active status under the laws of Wisconsin. Prior to the MI Merger, MVT Holding will not have conducted any operations and will have no assets

or liabilities other than the capital stock of New MI Corp. and Merger Sub, those incident to MVT Holding's formation and the holding of the capital stock of New MI Corp. and those pursuant to this Agreement and the Transactions.

## (b) Capital Structure.

(i) On the date hereof, the authorized capital stock of MVT Holding consists of 1,000 shares of MVT Holding Common Stock, all of which are, as of the date hereof, and at all times prior to the MI Effective Time will be, owned of record and beneficially by MI Corp. free and clear of any Liens. As of the date hereof there are no and as of the MI Effective Time (except as permitted pursuant to this Agreement or pursuant to the Transaction Agreements) there will be no other shares of capital stock or other equity securities of MVT Holding that are outstanding. Immediately following the MI Effective Time, there will be outstanding a number of shares of MVT Holding Common Stock equal to the number of shares of MI Common Stock outstanding as of immediately prior to the MI Effective Time. All issued and outstanding shares of MVT Holding Common Stock are duly authorized, validly issued, fully paid and nonassessable, and the shares of MVT Holding Common Stock are not entitled to preemptive rights. There are no outstanding options, warrants, Contracts or other rights to acquire capital stock of MVT Holding.

(ii) No bonds, debentures, notes or other indebtedness of MVT Holding having the right to vote (or convertible into or exchangeable for securities having the right to vote) on any matters on which shareholders of MVT Holding may vote ("<u>MVT Holding Voting Debt</u>") are issued or outstanding.

(iii) Except as contemplated by this Agreement and the Transaction Agreements, there are no securities, options, warrants, calls, rights, commitments, agreements, arrangements or undertakings of any kind to which MVT Holding or any of its Subsidiaries is a party or by which any of them is bound obligating MVT Holding or any of its Subsidiaries to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of capital stock of MVT Holding or any of its Subsidiaries, MVT Holding Voting Debt, MVT Holding Common Stock or other voting securities of MVT Holding or any of its Subsidiaries to issue, grant, extend or enter into any such security, option, warrant, call, right, commitment, agreement, arrangement or undertaking. Except as contemplated by this Agreement and the Transaction Agreement there are no outstanding obligations of MVT Holding or any of its Subsidiaries to repurchase, redeem or otherwise acquire any shares of capital stock of MVT Holding or any of its Subsidiaries.

(iv) Other than the Contracts contemplated to effect the Transactions, there are no shareholder agreements, voting trusts or other Contracts to which MVT Holding is a party or by which it is bound relating to the voting or transfer of any shares of capital stock of MVT Holding.

# (c) Authority; No Conflicts.

(i) MVT Holding has all requisite corporate power and authority to enter into this Agreement and the Transaction Agreements with respect to which MVT Holding

is contemplated thereby to be a party and to consummate the Transactions, subject to the Additional Required Approvals. The execution and delivery of this Agreement and the Transaction Agreements with respect to which MVT Holding is contemplated thereby to be a party by MVT Holding and the consummation by MVT Holding of the Transactions have been duly authorized by all necessary corporate action on the part of MVT Holding, subject to the Additional Required Approvals. This Agreement has been, and the Transaction Agreements with respect to which MVT Holding is contemplated thereby to be a party will be, duly executed and delivered by MVT Holding and, assuming the due authorization and valid execution and delivery of this Agreement or the applicable Transaction Agreement with respect to which MVT Holding is contemplated thereby to be a party by the other parties hereto and thereto (other than MI Corp., Merger Sub and MVT Corp.), as applicable, constitutes or will constitute a valid and binding agreement of MVT Holding, enforceable against MVT Holding in accordance with their respective terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, fraudulent conveyance, moratorium and similar Applicable Laws relating to or affecting creditors generally or by general equity principles (regardless of whether such enforceability is considered in a proceeding in equity or at law). After the MI Effective Time, no approval of holders of MVT Holding capital stock will be required to effect the Transactions.

(ii) The execution and delivery of this Agreement by MVT Holding does not, the execution and delivery by MVT Holding of the Transaction Agreements with respect to which MVT Holding is contemplated thereby to be a party will not, and the consummation of the Transactions will not result in a Violation under: (A) any provision of the organizational documents of MVT Holding or (B) except as, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on MVT Corp., subject to obtaining or making MVT Holding Necessary Consents, any permit, concession, franchise, license, judgment, order, decree, statute, law, ordinance, rule or regulation applicable to MVT Holding.

(iii) No consent, waiver, approval, order or authorization of, or registration, declaration or filing with, any Governmental Entity or any other Person is required to be obtained or made by or with respect to MVT Holding in connection with the execution and delivery of this Agreement and the Transaction Agreements with respect to which MVT Holding is contemplated thereby to be a party by MVT Holding or the consummation by MVT Holding of the Transactions, except for those required under or in relation to (A) the HSR Act, (B) state securities or "blue sky" laws or regulations, (C) the Securities Act, (D) the Exchange Act, (E) filings under the WBCL and the WLLCL to effect the transactions contemplated by Section 2.01 of the Separation Agreement and Article III of this Agreement, (F) the rules and regulations of the NYSE and, if different, such national securities exchange designated by MVT Holding and Investor pursuant to Section 6.15, (G) applicable state or federal banking laws or regulations, (H) applicable state or federal laws or regulations, and the rules and regulations of the NASD or any other applicable regulatory authority (including the NASD), regulating broker dealers, investment advisors and insurance companies, (I) obtaining the MI Transaction Approval and the Additional Required Approvals, (J) antitrust or other competition laws of other jurisdictions, and (K) such consents, waivers, approvals, orders, authorizations, registrations, declarations and filings the failure of which to make or obtain, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on MVT Corp. Consents, waivers, approvals, orders, authorizations, registrations, declarations and filings required under or in

relation to any of the foregoing <u>clauses (A)</u> through <u>(J)</u> are referred to as the "<u>MVT Holding Necessary Consents</u>."

#### (d) Information Supplied.

(i) None of the information supplied or to be supplied by MVT Holding specifically for inclusion or incorporation by reference in (A) the Form S-4 will, at the time it becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, (B) the Proxy Statement will, on the date it is first mailed to MI Corp. shareholders or at the time of the MI Shareholders Meeting, respectively, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading and (C) the Form 10 will, at the time it becomes effective, contain any untrue statement of a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading and (C) the Form 10 will, at the time it becomes effective, contain any untrue statement of a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading and (C) the Form 10 will, at the time it becomes effective, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

(ii) Notwithstanding the foregoing provisions of this <u>Section 4.4(d)</u>, no representation or warranty is made by MVT Holding with respect to statements made or incorporated by reference in the Form S-4, the Form 10 or the Proxy Statement based on information supplied by or on behalf of MI Corp. or MVT Corp. for inclusion or incorporation by reference therein.

(e) <u>Litigation</u>. There is no Action pending or, to the Knowledge of MVT Holding, threatened against MVT Holding which, individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect on MVT Corp., nor is there any judgment, decree, injunction, rule or order of any Governmental Entity or arbitrator outstanding against MVT Holding which, individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect on MVT Corp.

(f) <u>Status of MVT Holding Shares</u>. Assuming the MI Transaction Approval has been obtained, the Shares and the shares of MVT Holding Common Stock being issued at the Closing will have been duly authorized by all necessary corporate action on the part of MI Corp. and MVT Holding, and such Shares and the shares of MVT Holding Common Stock will have been validly issued and, assuming payment therefor has been made, will be fully paid and nonassessable, and the issuance of such Shares and the shares of MVT Holding Common Stock will not be subject to preemptive rights.

#### ARTICLE V

# COVENANTS RELATING TO CONDUCT OF BUSINESS

Section 5.1 <u>Covenants of MI Corp. and MVT Corp</u>. During the period from the date of this Agreement and continuing until the Closing Date, MVT Holding, MI Corp. and

MVT Corp. each agrees that except as required or otherwise expressly contemplated or expressly permitted by this Agreement, the Transaction Agreements or <u>Section 5.1</u> of the MI Disclosure Schedule or the MVT Disclosure Schedule or to the extent that Investor shall otherwise consent in writing, which consent shall not be unreasonably withheld, delayed or conditioned (other than with respect to <u>Sections 5.1(b), (c), (d), (j)</u> and (<u>l)</u> in which case such consent may be withheld, delayed or conditioned in Investor's sole discretion):

(a) <u>Ordinary Course</u>. The members of the MI Group (solely with respect to the MVT Business) and the members of the MVT Group shall carry on their respective businesses in the ordinary course, in substantially the same manner as heretofore conducted (and the members of the MI Group shall allow the MVT Group to carry on the MVT Business with a level of autonomy that is consistent with the MI Group's past practice with respect to the MVT Business, subject to Applicable Laws), and shall use their reasonable best efforts to preserve intact their present business organizations, maintain their material rights, licenses and permits, keep available the services of their current officers and other key employees and preserve their relationships with customers, franchises and others having business dealings with them to the end that their ongoing businesses and goodwill shall not be materially impaired at the Closing Date.

(b) <u>Dividends; Changes in Share Capital</u>. MVT Holding, MI Corp. and MVT Corp. shall not, and MI Corp. and MVT Corp. shall not permit any Subsidiary of MVT Holding or MVT Corp. to, nor shall MVT Holding, MVT Corp. or any Subsidiary of MVT Holding or MVT Corp. propose to, declare, set aside or pay any dividends on or make other distributions (whether in cash, stock or property or any combination thereof) in respect of any capital stock of MVT Holding, MVT Corp. or any Subsidiaries of MVT Holding or MVT Corp., except (i) for the MVT Distribution, the MVT Dividend, the Share Distribution and the MI Contribution, and (ii) for the declaration and payment of cash dividends or distributions by any wholly-owned Subsidiary of MVT Holding or MVT Corp. to MVT Holding or MVT Corp., as applicable, except for quarterly cash dividends from the MVT Group to the MI Group in an amount not to exceed \$1,000,000 per quarter. Each of MVT Holding and MVT Corp. shall not, (A) split, combine or reclassify any of its capital stock, except for any such transaction by a wholly-owned Subsidiary of MVT Holding which remains a wholly-owned Subsidiary after consummation of such transaction or (B) amend the terms or change the period of exercisability of, purchase, repurchase, redeem or otherwise acquire any shares of its capital stock or any securities convertible into or exercisable or exchangeable for, or any rights, warrants, calls or options to acquire, any shares of its capital stock.

(c) <u>Issuance of Securities</u>. MVT Holding and MVT Corp. shall not, and MVT Holding and MVT Corp. shall not permit any of their respective Subsidiaries to, issue, deliver, sell, pledge, dispose of or otherwise encumber, or authorize or propose the issuance, delivery, sale, pledge, disposition or encumbrance of, (x) any shares of its capital stock of any class, (y) any MVT Voting Debt or MVT Holding Voting Debt or (z) any securities convertible into or exercisable or exchangeable for, or any rights, warrants, calls or options to acquire, any such shares or MVT Voting Debt or MVT

Holding Voting Debt, or enter into any commitment, arrangement, undertaking or agreement with respect to any of the foregoing, other than issuances, redemptions, deliveries and sales by a wholly owned Subsidiary of MVT Holding or MVT Corp. of capital stock of such Subsidiary to such Subsidiary's parent or another wholly owned Subsidiary of MVT Holding or MVT Corp.

(d) <u>Governing Documents</u>. Except to the extent specifically required to comply with its obligations hereunder or under the Transaction Agreements, MVT Holding and MVT Corp. shall not amend or propose to amend its articles of incorporation, by-laws or other governing documents. Except to the extent specifically required to comply with its obligations hereunder and under the Transactions Agreements, MI Corp. will not amend or propose to amend the articles of incorporation or by-laws of MI Corp. in any manner that would prevent or materially delay the consummation of the Transactions.

(e) <u>No Acquisitions</u>. Each of MI Corp. (solely with respect to the MVT Business), MVT Holding and MVT Corp. shall not, and shall not permit any Subsidiary of MVT Corp. or MVT Holding to, in a single transaction or series of related transactions, acquire or agree to acquire by merger or consolidation, or by purchasing an equity interest in or a portion of the assets of, or by any other manner, any business or any corporation, partnership, limited liability entity, joint venture, association or other business organization or division thereof or otherwise acquire or agree to acquire any assets, in each case if such acquisition would prevent or materially delay obtaining any consents, approvals or expirations of waiting periods from any Governmental Entity required for the consummation of the Transactions contemplated by this Agreement or the Transaction Agreements or otherwise would materially impede, frustrate or delay the consummation of the Transactions.

(f) <u>No Dispositions</u>. MVT Holding and MVT Corp. shall not, and shall not permit any of their Subsidiaries to, in a single transaction or a series of related transactions, sell (including sale-leaseback), lease, license or otherwise encumber or subject to any Lien or otherwise dispose of to any Person that is not a Subsidiary of MVT Corp., or agree to sell (or engage in a sale-leaseback), lease, license or otherwise encumber or subject to any Lien or otherwise dispose of to any Person that is not a Subsidiary of MVT Corp., any assets of MVT Corp. or any Subsidiary of MVT Holding or MVT Corp. (including capital stock of MVT Corp. and any MVT Subsidiary), other than (i) in the ordinary course consistent with past practice or (ii) sales (including sale-leasebacks), leasebacks), leases, licenses, encumbrances, Liens or other dispositions of assets with a fair market value of no more than \$3,000,000 in the aggregate.

(g) <u>Capital Expenditures</u>. Other than in connection with acquisitions permitted by <u>Section 5.1(e)</u>, MVT Holding and MVT Corp. shall not, and shall not permit any of their respective Subsidiaries to, commit to any capital expenditures that would reasonably be expected to be incurred after the Closing Date or incur any obligations or liabilities in connection with any capital expenditures that would reasonably be expected to be incurred after the Closing Date, in each case other than (i) any capital expenditure involving commitments of less than \$1,000,000 in the aggregate

or (ii) any capital expenditure set forth in the capital expenditure plan in effect on the date hereof and attached hereto in <u>Section 5.1</u> of the MVT Disclosure Schedule.

(h) <u>Settlement of Litigation</u>. MVT Holding, MVT Corp. and MI Corp. shall not, and MVT Corp. shall not permit any other member of the MVT Group to, settle any litigation, investigation, arbitration, proceeding or other claim if (i) MVT Corp., MVT Holding or any of their Subsidiaries would be required to pay following the Closing Date in excess of \$4,000,000 for any such settlement or \$15 million in the aggregate for all such settlements or (ii) if such settlement would obligate MVT Holding, MVT Corp. or any other member of the MVT Group to take any action or restrict MVT Holding, MVT Corp. or any other member of the MVT Group from taking any action at or after the Closing Date that would be material to MVT Corp. and its Subsidiaries taken as a whole.

(i) <u>Compensation</u>. MVT Corp. and its Subsidiaries, and MI Corp. and its Subsidiaries with respect to any MVT Employee, shall not (i) accelerate the vesting of, or the lapsing of restrictions with respect to, any stock options or other stock-based compensation, except (x) as required by Applicable Laws, or (y) in accordance with the existing terms of such awards or Contracts in existence on the date hereof, (ii) increase, or commit to increase, the amount of compensation or employee benefits of any current or former MVT Employee except for increases in compensation or employee benefits for individual MVT Employees (as opposed to increases applicable to MVT Employees generally) other than members of the Executive Committee of MVT in the ordinary course of business consistent with past practice, (iii) enter into any new, or amend any existing Contract with any MVT Employee or consultant of MVT Corp. or any other member of the MVT Group, regarding his or her employees who are not members of the Executive Committee of MVT); <u>provided</u>, <u>however</u>, that nothing in this clause (iii) shall limit or prevent MVT Corp. from hiring or extending an offer to employment to any person to serve as the Chief Financial Officer of MVT Corp. or entering into any Contract in connection therewith regarding his or her employment, compensation or benefits; <u>provided</u>, <u>further</u>, that MVT Corp. shall consult with Investor prior to hiring such person and in entering into such Contract, or (iv) transfer any employees primarily providing services to the MVT Business to a member of the MI Group or transfer any employees primarily providing services to business to a member of the MVT Group.

(j) <u>Tax Matters</u>. Except as required by Applicable Laws or as contemplated by this Agreement or the Tax Allocation Agreement, neither MVT Corp. nor any of its Subsidiaries shall (i) make, amend or change any material Tax election, (ii) make a request for a material Tax ruling (other than the Private Letter Ruling) or enter into a material closing agreement, (iii) settle or compromise any material Tax liability or material Tax claims, (iv) file any material amendments to any previously filed material Tax Returns or (v) surrender any right to claim any material amount of refund of any Taxes.

(k) <u>Accounting Methods</u>. Except as disclosed in the MI Corp. Filed SEC Reports or the MVT Financial Statements, as required by a Governmental Entity or as required by changes in GAAP as concurred by MVT Corp.'s independent public accountants, MVT Corp. shall not make any material change to its methods of accounting in effect at December 31, 2006 (including procedures with respect to revenue recognition, payments of accounts payable and collection of accounts receivable).

(1) <u>Related Party Agreements Indebtedness</u>. Except in connection with the Transactions contemplated hereby, MVT Holding, MI Corp. and MVT Corp. shall cause their respective Subsidiaries not to, enter into, waive any material rights under or amend in any material respect any material Contract between any member of the MVT Group, on the one hand, and any member of the MI Group, on the other hand, that, (x) pursuant to the Separation Agreement, will survive the Distribution Time; or (y) provides for indebtedness for borrowed money (including any amendment or waiver with respect to the MVT Notes); provided, however, that MVT Corp. and MI Corp. may amend the Continuing Business Agreements in the ordinary course of business. MVT Holding, MI Corp. and MVT Corp. shall not and shall cause their respective Subsidiaries to not (i) transfer any material liabilities (other than those liabilities that relate primarily to the MVT Business) from any member of the MI Group to any member of the MVT Group. MVT Holding and MVT Corp. shall not, and shall not permit any of their respective Subsidiaries to, incur any indebtedness for borrowed money, other than (A) trade payables incurred in the ordinary course of business, (B) draws on MVT Corp.'s revolving line of credit for settlement-type transactions, and (C) the Debt Financing.

(m) <u>No Related Actions</u>. MVT Holding and MI Corp. (as to MVT Corp. and the MVT Subsidiaries and/or the MVT Business) will not, and MVT Corp. will not, and will not permit any of its Subsidiaries to, agree or commit to do any of the actions specified in <u>Sections 5.1(b)</u> through <u>5.1(l)</u>.

Section 5.2 <u>Control of Other Party's Business</u>. Nothing contained in this Agreement shall give MI Corp. or MVT Corp., directly or indirectly, the right to control or direct Investor's business prior to the Closing. Nothing contained in this Agreement shall give Investor, directly or indirectly, the right to control or direct the operations of the MVT Business prior to the Closing.

Section 5.3 <u>Transfer of Investor Interests</u>. Between the date hereof and the Closing Date, Investor shall ensure that it remains wholly-owned by the Equity Fund, except that nothing herein shall prevent (i) interests in Investor from being assigned to or issued to any Successor Fund or (ii) a de minimis portion of the interests in Investor (no more than 1%) from being assigned to any Person or Persons; <u>provided</u>, <u>that</u>, such transfer would not be reasonably likely to prevent or impair or delay the consummation of the Transactions; <u>provided</u>, <u>further</u>, no such assignment shall relieve the Equity Fund of any of its obligations under the Limited Guarantee or the Equity Commitment Letter.

# ARTICLE VI

# ADDITIONAL AGREEMENTS

# Section 6.1 Preparation of Form S-4, Proxy Statement and Form 10; Shareholders Meeting.

(a) As promptly as practicable following the date hereof, MI Corp. shall or shall cause MVT Holding to prepare a proxy statement to be mailed to MI Corp.'s shareholders in connection with the MI Shareholders Meeting (such proxy statement, and any amendments or supplements thereto, the "Proxy Statement") and a registration statement on Form S-4 with respect to the shares of MVT Holding Common Stock to be issued in the MI Merger (the "Form S-4" and the prospectus of MVT Holding included in the Form S-4 the "MVT Holding Prospectus"), in each case in consultation with Investor, and shall or shall cause MVT Holding to file the Form S-4 and the Proxy Statement with the SEC. The Form S-4 and the Proxy Statement will comply as to form in all material respects with the requirements of the Exchange Act and the Securities Act, as applicable, and the rules and regulations of the SEC thereunder. MI Corp. shall use its reasonable best efforts to have the Proxy Statement cleared by the SEC as promptly as practicable after filing with the SEC and to have the Form S-4 declared effective by the SEC as promptly as practicable after filing with the SEC. MI Corp. shall use its reasonable best efforts to cause the Proxy Statement and the MVT Holding Prospectus to be mailed to MI Corp.'s shareholders as promptly as practicable after the Proxy Statement is cleared by the SEC and the Form S-4 is declared effective by the SEC. MI Corp. shall, as promptly as practicable after receipt thereof, provide to Investor copies of, consult with Investor and prepare written responses with respect to, any written comments received from the SEC with respect to the Form S-4 and the Proxy Statement and advise Investor of any oral comments with respect to the Form S-4 and the Proxy Statement received from the SEC. Notwithstanding any other provision herein to the contrary, no amendment or supplement to the Form S-4 or the Proxy Statement shall be made, and no correspondence filed with the SEC with respect thereto, without the approval of Investor, which approval shall not be unreasonably withheld, conditioned or delayed. MI Corp. will advise Investor, promptly after it receives notice thereof, of the time when the Proxy Statement has been cleared and the Form S-4 has been declared effective by the SEC, the issuance of any stop order with respect to the Proxy Statement or MVT Holding Prospectus or any request by the SEC for amendment of the Form S-4 or the Proxy Statement. If at any time prior to the Closing Date any information relating to Investor, MI Corp. or MVT Corp., or any of their respective Affiliates, officers or directors, should be discovered by Investor, MI Corp. or MVT Corp. which should be set forth in an amendment or supplement to the Form S-4 or the Proxy Statement so that any of such documents would not include any misstatement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, the party which discovers such information shall promptly notify the other parties hereto and, to the extent required by Applicable Laws, an appropriate amendment or supplement describing such information shall be promptly filed by MI Corp. with the SEC and disseminated to the shareholders of MI Corp. Investor shall

cooperate with MI Corp. in the preparation of the Proxy Statement, the Form 10 and the Form S-4 or any amendment or supplement thereto. MI Corp. shall prepare and shall cause New MI Corp. to file with the SEC as promptly as practicable after the date hereof the Form 10 and MI Corp. shall use its reasonable best efforts to cause the Form 10 to be declared effective and mailed to its shareholders as promptly as practicable.

(b) MI Corp. shall duly take all lawful action to call, give notice of, convene and hold a meeting of its shareholders as promptly as practicable after the date of this Agreement (including any adjournment or postponements thereof, the "<u>MI Shareholders Meeting</u>") for the purpose of obtaining the MI Transaction Approval by a majority of the outstanding shares of MI Common Stock (the "<u>MI Vote</u>"). Unless the Board of Directors of MI Corp. shall have made a Change in the MI Recommendation pursuant to <u>Section 6.5</u>, MI Corp. shall use reasonable best efforts to solicit the MI Transaction Approval by the MI Vote and the Board of Directors of MI Corp. shall declare that this Agreement and the Transactions (including the Share Issuance and the MI Merger) are advisable and in the best interests of MI Corp. and its shareholders and recommend to the shareholders of MI Corp. approval of this Agreement and the Transactions (including the Share Issuance and the MI Merger) by the shareholders of MI Corp. (the "<u>MI Recommendation</u>"). Notwithstanding anything herein to the contrary, if a Change in the MI Recommendation is made and within fifteen Business Days after Investor receives any written notice from MI Corp. of such Change in the MI Recommendation pursuant to <u>Section 6.5(d)(ii)</u> or (<u>iii)</u>, Investor delivers a written notice (a "<u>Force the MI Vote Notice</u>") to MI Corp. that Investor desires that this Agreement and the Transactions (including the Share Issuance and the MI Merger) be submitted to the shareholders of MI, then this Agreement and the Transactions (including the Share Issuance and the MI Merger) be submitted to the shareholders of MI, then this Agreement and the Transaction Approval and nothing contained herein shall be deemed to relieve MI of such obligation, unless this Agreement has previously been terminated pursuant to <u>Section 8.1(e)</u> or <u>8.1(h</u>).

(c) Subject to the terms and conditions of this Agreement, MI Corp., MVT Holding and MVT Corp. shall take or cause to be taken all actions necessary to obtain the Additional Required Approvals.

# Section 6.2 Governance Matters.

(a) MVT Holding and MI Corp. shall take all actions necessary to cause the Board of Directors of MVT Holding, effective at the Share Distribution Time and thereafter until changed in accordance with the organizational documents of MVT Holding, Applicable Laws and the Shareholders Agreement, to consist of eleven directors, consisting of (subject to <u>Section 6.4(c)</u>) (i) three directors designated by Investor, (ii) two directors who shall be officers of MVT Corp., one of whom shall be the President and Chief Executive Officer and one of whom shall be the Senior Vice President and Chief Operating Officer, (iii) one director who shall be designated by MI Corp. and shall initially be Dennis J. Kuester and (iv) five additional directors selected by MI Corp. after consulting with the President and Chief Executive Officer of MVT Corp. and with the consent of Investor (which shall not be unreasonably withheld or

delayed), each of whom shall qualify as Independent Directors and one of whom shall be a director of MI Corp.; provided, however, that if any of the individuals described in <u>clauses (i)</u> through (iv) are unable to begin serving as a director of MVT Holding at the Share Distribution Time, the MVT Holding Board of Directors shall not include such individual until he or she is able to serve as a director; provided, that he or she shall have agreed to begin to serve within a reasonable period of time thereafter; then such directors shall be Independent Directors selected pursuant to the foregoing clause (iv) in a manner which addresses the reason that the designee was originally prevented from being designated; provided, further, that if Investor or MI Corp. is prevented by Applicable Law or regulatory process from designating any of its designees pursuant to the foregoing clause (i), (iii) or (iv) (in the case of an MI Corp. director), as applicable, or if such designation would result in MVT Holding being an affiliate of MI Corp. for purposes of Sections 23A or 23B of the Federal Reserve Act, then such directors shall be Independent Directors designated pursuant to the foregoing clause (iv) in a manner which addresses the reason that the designee was originally prevented from being designated for prevention or prohibition. The term "Independent Director" means an individual who, as a member of the Board of Directors of MVT Holding following the Closing Date, would be independent of MVT Holding under the applicable rules of the NYSE or such other national securities exchange designated by MVT Holding and Investor pursuant to Section 6.15. The MVT Holding Board of Directors will have committees as contemplated by the Shareholders Agreement. The Chairman of the Board of MVT Holding shall be Dennis J. Kuester for a period of one year from the date hereof. If Dennis J. Kuester is unable to serve during such one-year period, and after such one-year period Frank Martire, shall, subject to the approval of the Board of Directors of MVT Holding, succeed Dennis J. Kuester as Chairman of the Board of MVT Holding. Until the Board of Directors of MVT Holding shall determine otherwise, the regular meetings of the Board of Directors shall be on the third Thursday of each February, April, June, August, October and December.

(b) The President and Chief Executive of MVT Corp. immediately prior to the Closing Date shall be the President and Chief Executive Officer of MVT Holding at the Share Distribution Time and thereafter until duly changed in accordance with the organizational documents of MVT Holding and Applicable Law.

Section 6.3 <u>Access to Information</u>. Upon reasonable notice, each of MI Corp. and MVT Corp. shall (and shall cause its Subsidiaries to), during the period prior to the earlier of the Closing Date or the date this Agreement is terminated, afford to Investor and to its respective officers, employees, accountants, counsel, financial advisors and other authorized representatives, reasonable access during normal business hours, to all the books, records, Contracts, properties, plants and personnel of the MVT Business and, during such period, MI Corp. or MVT Corp. shall (and shall cause its Subsidiaries to) furnish promptly to Investor (a) notice of each material report, schedule, registration statement and other document filed, published, announced or received by MVT Corp. during such period pursuant to the requirements of Federal or state securities laws, as applicable (other than documents which MI Corp. or MVT Corp. is not permitted to disclose under Applicable Laws) and (b) all information concerning MVT Corp. and its business, properties and personnel as Investor may reasonably request; provided, however, that MI Corp. or MVT Corp. may restrict the foregoing access to the

extent that (i) any Applicable Laws or Contracts requires MI Corp., MVT Corp. or their Subsidiaries to restrict or prohibit access to any such properties or information, (ii) disclosure of such information would violate confidentiality obligations to a third Person, (iii) disclosure of such information would be reasonably likely to result in significant competitive harm to MI Corp. or MVT Corp. if the Transactions were not consummated or (iv) in the case of MVT Corp. and its Subsidiaries the information is not related to the MVT Business. Investor will hold any such information obtained pursuant to this <u>Section 6.3</u> in confidence in accordance with, and will otherwise be subject to, the provisions of the Confidentiality Agreement dated January 29, 2007 between MI Corp. and Warburg Pincus LLC (as it may be amended or supplemented, the "<u>Confidentiality Agreement</u>"). The Confidentiality Agreement shall survive any termination of this Agreement. Any investigation by Investor shall not affect the representations and warranties contained herein or the conditions to the respective obligations of the parties to consummate the Transactions.

#### Section 6.4 Reasonable Best Efforts.

(a) Subject to the terms and conditions of this Agreement, each party will use its reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, and to assist and cooperate with the other parties in doing or causing to be done, all things necessary, proper or advisable under this Agreement and Applicable Laws to consummate the Transactions as soon as reasonably practicable after the date hereof, including (i) preparing and filing as promptly as practicable all documentation to effect all necessary applications, notices, petitions, filings, waivers and Tax ruling requests and to obtain as promptly as practicable all Investor Necessary Consents, MI Necessary Consents, MVT Necessary Consents and MVT Holding Necessary Consents and all other consents, waivers, licenses, orders, registrations, approvals, permits, rulings, authorizations and clearances necessary or advisable to be obtained from any third party and/or any Governmental Entity in order to consummate the Transactions (collectively, the "<u>Required Approvals</u>") and (ii) taking all reasonable steps as may be necessary to obtain all Required Approvals. In furtherance and not in limitation of the foregoing, each party hereto agrees to make (A) an appropriate filing of a Notification and Report Form pursuant to the HSR Act with respect to the Transactions as promptly as reasonably practicable, and, in each case, to supply as promptly as reasonably practicable any additional information and documentary material that may be requested pursuant to such Applicable Laws or by such authorities and to use its reasonable best efforts to cause the expiration or termination of the applicable waiting periods under the HSR Act and the receipt of the Required Approvals under such other Applicable Laws or from such authorities as soon as reasonably practicable.

(b) Investor, on the one hand, and MI Corp., MVT Corp. and MVT Holding, on the other hand, shall, in connection with the efforts referenced in <u>Section 6.4(a)</u> to obtain all Required Approvals, use its reasonable best efforts to (i) cooperate in all respects with each other in connection with any filing or submission and in connection with any investigation or other inquiry, including any proceeding initiated by a private party, (ii) promptly inform the other party of any communication received by

such party from, or given by such party to, the Antitrust Division of the Department of Justice (the "<u>DOJ</u>"), the Federal Trade Commission ("<u>FTC</u>"), the Board of Governors of the Federal Reserve System (the "<u>FRB</u>"), or any other Governmental Entity and of any material communication received or given in connection with any proceeding by a private party, in each case regarding any of the Transactions and (iii) permit the other party to review any communication given by it to, and consult with each other in advance of any meeting or conference with, the DOJ, the FTC, the FRB or any such other Governmental Entity or, in connection with any proceeding by a private party, with any other Person, and to the extent appropriate or permitted by the DOJ, the FTC, the FRB or such other applicable Governmental Entity or other Person, give the other party the opportunity to attend and participate in such meetings and conferences.

(c) In furtherance and not in limitation of the covenants of the parties contained in <u>Section 6.4(a)</u> and <u>Section 6.4(b)</u>, if any administrative or judicial action or proceeding, including any proceeding by a private party, is instituted (or threatened to be instituted) to prohibit or make illegal the Transactions, or if executive order, decree, injunction or administrative order is enacted, entered, promulgated or enforced by a Governmental Entity which would make any of the Transactions illegal or would otherwise prohibit or materially impair or delay the consummation of any of the Transactions, each of MI Corp., MVT Corp., MVT Holding and Investor shall cooperate in all respects with each other and use its respective reasonable best efforts, to contest and resist any such action or proceeding and to have vacated, lifted, reversed or overturned any decree, judgment, injunction or other order, whether temporary, preliminary or permanent, that is in effect and that prohibits, prevents or restricts consummation of any of the Transactions and to have such executive order, decree, injunction or administrative order repealed, rescinded or made inapplicable so as to permit consummation of the Transactions. Without limiting the generality of the foregoing, each of Investor, MI Corp., MVT Corp., and MVT Holding will agree to any restrictions or modifications, or take any action or enter into any settlement or other agreement or binding arrangement to sell, hold, separate or otherwise dispose of any assets, including the capital stock of any Subsidiary, suggested or requested by any Governmental Entity in order to facilitate the receipt of any Required Approval (including modifications to the Shareholders Agreement, the Administrative Services Agreement, or the Continuing Business Agreements), so long as such restrictions, modifications, sales, disposals or other actions would not, individually or in the aggregate, (i) reasonably be expected to have a Material Adverse Effect on MVT Corp. or MVT Holding, (ii) reasonably be expected to have a material adverse effect on MI Corp., (iii) materially detract from the expected financial benefits from the Transaction to the MI Group or (iv) in the case of Investor, result in (A) an increase in the Purchase Price or a decrease in the percentage of fully diluted shares of MVT Holding Common Stock represented by the Investor Share Number or (B) Investor or any Affiliate of Investor (other than members of the MVT Group) becoming subject to bank regulation or supervision (including being required to register as a bank holding company), other than any bank regulation that Investor or such Affiliates are, or become subject to, for reasons unrelated to the Transactions; it being understood that no modification to the provisions of Section 6.2 hereof or Sections 1.1 or 1.2 of the Shareholders Agreement

shall be considered to have any of the effects referred to in the foregoing clauses (i) through (iv).

(d) Notwithstanding the foregoing or any other provision of this Agreement, nothing in this <u>Section 6.4</u> shall (i) limit a party's right to terminate this Agreement pursuant to <u>Section 8.1(b)</u> or <u>Section 8.1(c)</u> so long as such party has complied with its obligations under this <u>Section 6.4</u>, (ii) limit MI Corp.'s right to make a Change in the MI Recommendation in accordance with <u>Section 6.5</u>, or (iii) limit MI Corp.'s ability to enter into any MI Acquisition Agreement so long as MI Corp. has previously terminated this Agreement pursuant to <u>Section 8.1(h)</u> and paid Investor the Termination Fee in accordance with the terms of <u>Section 8.2(c)</u>.

(e) Except as set forth in Section 6.4(c) after giving effect to the limitations set forth therein, nothing in this Section 6.4 shall require any of Investor, MI Corp. or MVT Corp., MVT Holding or any of their respective Subsidiaries to take any action or enter into any settlement or other agreement or binding arrangement that requires such Person to sell, hold separate or otherwise dispose of any businesses, product lines or assets of MI Corp., MVT Corp., MVT Holding Investor or any of their Subsidiaries including the capital stock of any such Subsidiary. This Section 6.4 shall not be deemed to address the obligations of the parties with respect to the Private Letter Ruling, which are addressed in Section 6.10, the Debt Financing, which are addressed in Section 6.13 or, with respect to the obligations of MI Corp. and its Subsidiaries, the Surplus and Solvency Opinions, which are addressed in Section 6.17.

#### Section 6.5 MI Acquisition Proposal; Change in Recommendation.

(a) From and after the date of this Agreement until the earlier of the Closing Date or the termination of this Agreement in accordance with <u>Article VIII</u>, MI Corp. agrees that neither it, any of its Subsidiaries nor any of the executive officers or directors of MI Corp. or of MVT Corp. shall, and MI Corp. shall not authorize any of its or its Subsidiaries respective officers, directors, employees, agents or representatives to (and shall instruct J.P. Morgan Securities Inc., in its capacity as MI Corp.'s investment banker, not to), directly or indirectly, (A) solicit, initiate or knowingly encourage (including by way of furnishing non-public information) the making of any inquiry, offer or proposal which constitutes or that would reasonably be expected to lead to, any MI Acquisition Proposal or MVT Acquisition Proposal, (B) enter into any letter of intent, agreement in principle, merger agreement, acquisition agreement, or other agreement providing for any MI Acquisition Proposal (each, an "<u>MI Acquisition</u> <u>Agreement</u>") or any MVT Acquisition Proposal or (C) participate in any discussions or negotiations regarding any MI Acquisition Proposal or any MVT Acquisition Proposal or MVT Acquisition Proposal from any Person, MI Corp. may (x) furnish information (including non-public information) with respect to MI Corp. and its Subsidiaries, including MVT Corp., to any such Person pursuant to a confidentiality agreement, which in the case of an MVT Acquisition Proposal, shall contain terms no less restrictive on such Person than those in the Confidentiality Agreement are to

Investor and (y) participate in negotiations with such Person regarding such MI Acquisition Proposal or MVT Acquisition Proposal, if, (1) in each case, in the good faith judgment of the Board of Directors of MI Corp. after consultation with its outside legal counsel, failure to take such action would be inconsistent with the fiduciary duties of the Board of Directors of MI Corp. to its shareholders under Applicable Laws and (2) in the case of an MVT Acquisition Proposal only, in the good faith judgment of the Board of Directors of MI Corp., after consultation with its outside legal counsel and financial advisor(s), such MVT Acquisition Proposal is or is reasonably likely to lead to a Superior MVT Acquisition Proposal. MI Corp. shall (i) promptly (and in any case within 48 hours) notify Investor if MI Corp. has received any MVT Acquisition Proposal and (ii) keep Investor reasonably informed on a prompt basis as to any material developments regarding any such MVT Acquisition Proposal. Such notification shall include, to the extent then known, the identity of the parties and a copy of any MVT Acquisition Proposal and the material documents submitted therewith and, if conveyed orally, a description of this <u>Section 6.5</u> solely because of a failure to comply with the preceding two sentences of this <u>Section 6.5(a)</u>, so long as it shall have materially complied with those two sentences. Notwithstanding anything contained herein to the contrary, nothing herein shall prohibit MI Corp. from, following receipt of an MI Acquisition Proposal or an MVT Acquisition Proposal, so and the material terms thereof and the conditions to consummation; <u>provided</u>, <u>however</u>, that nothing in this sentence shall relieve MI Corp. of its obligations to notify Investor and keep Investor reasonably informed of any such MVT Acquisition Proposal and the material terms thereof and the conditions to consummation; <u>provided</u>, <u>however</u>, that nothing in this sentence shall relieve MI Corp. of its obligations to notify Investor and keep Investor

(b) For purposes of this Agreement, (i) "<u>MI Acquisition Proposal</u>" means any inquiry, proposal or offer (other than an MVT Acquisition Proposal) from any Person with respect to (A) a merger, reorganization, share exchange, consolidation, business combination, recapitalization, liquidation, dissolution or similar transaction involving MI Corp. or any of its Subsidiaries that, if consummated, would result in any Person (or the shareholders of such Person in the aggregate) beneficially owning securities representing 25% or more of the total voting power of MI Corp. then outstanding, (B) any direct or indirect purchase or sale, lease, exchange, transfer or other disposition of 25% or more of the consolidated assets (including stock of MI Corp.'s Subsidiaries) of MI Corp. and its Subsidiaries, taken as a whole, or (C) any direct or indirect purchase or sale of, or tender or exchange offer for, or similar transaction with respect to, the equity securities of MI Corp. that, if consummated, would result in any Person (or the shareholders of such Person in the aggregate) beneficially owning securities representing 25% or more of MI Corp. (or of the surviving parent entity in such transaction) then outstanding, including in the case of each of <u>clauses (A)</u> through (<u>C</u>), any single or multi-step transaction or series of related transactions (other than an inquiry, proposal or offer made by Investor or any Affiliate thereof); <u>provided</u>, <u>however</u>, that no such inquiry, proposal or offer shall be considered to be an MI Acquisition Proposal if it relates to a transaction that the Board of Directors of MI Corp. reasonably determines in good faith, after consultation

with its outside legal counsel, is not reasonably likely to prevent or materially impair, modify or delay the consummation of the Transactions, (ii) "<u>MVT</u> <u>Acquisition Proposal</u>" means any inquiry, proposal or offer from any Person solely with respect to (A) a merger, reorganization, share exchange, consolidation, business combination, recapitalization, liquidation, dissolution or similar transaction solely involving MVT Corp. or any of its Subsidiaries that, if consummated, would result in any Person (or the shareholders of such Person in the aggregate) beneficially owning securities representing 25% or more of the total voting power of MVT Corp. then outstanding, (B) any direct or indirect purchase or sale, lease, exchange, transfer or other disposition of 25% or more of the consolidated assets (including stock of MVT Corp.'s Subsidiaries) of MVT Corp. and its Subsidiaries, taken as a whole, or (C) any direct or indirect purchase or sale of, or similar transaction with respect to, the equity securities of MVT Corp. that, if consummated, would result in any Person (or the shareholders of such Person in the aggregate) beneficially owning securities representing 25% or more of the total voting power of MVT Corp. (or of the surviving parent entity in such transaction) then outstanding, including in the case of each of clauses (A) through (C), any single or multistep transaction or series of related transactions (other than an inquiry, proposal or offer made by Investor or any Affiliate thereof), and (iii) "<u>Superior</u> <u>MVT Acquisition Proposal</u>" means any unsolicited bona fide written MVT Acquisition Proposal on terms that, in the good faith judgment of the Board of Directors of MI Corp. after consultation with its outside legal counsel and financial advisor(s) and after taking into account all legal, financial, regulatory and other aspects of the proposal, including the financing terms thereof, is (A) reasonably capable of being consummated and (B) superior from a financial point of view to the shareholders of

(c) Neither the Board of Directors of MI Corp. nor any committee thereof shall (A) withdraw, modify, qualify or amend the MI Recommendation in any manner adverse to Investor, (B) fail to publicly reaffirm the MI Recommendation within five Business Days after receiving a written request to do so from Investor in response to any public statement or disclosure by MI Corp. or any of its Affiliates that could reasonably be interpreted to have any of the effects set forth in clause (A) or (C) of this sentence, if such request is made by Investor within five Business Days after such public statement or disclosure, (C) approve, recommend, agree to or accept, any MI Acquisition Proposal or (D) approve, recommend, agree to or accept, any MVT Acquisition Proposal (each of the actions in <u>clauses (A)</u> through (<u>D</u>), a "<u>Change in Recommendation</u>"); provided, however, that at any time prior to the receipt of the MI Transaction Approval (or in the case of clause (C) above, at any time subject to compliance with the next proviso of this <u>Section 6.5.(c)</u>), the Board of Directors of MI Corp. may make a Change in the MI Recommendation if (x) MI Corp. has complied with this <u>Section 6.5.(y</u>) in the good faith judgment of the Board of Directors of MI Corp., after consultation with the outside legal counsel of MI Corp. to its shareholders under Applicable Laws and (z) in the case of a Change in the MI Recommendation Proposal, the Board of Directors of MI Corp., after consultation with its outside legal

counsel and financial advisors, has also determined in its good faith judgment that such MVT Acquisition Proposal is a Superior MVT Acquisition Proposal; <u>provided</u>, <u>further</u>, that at any time, the Board of Directors of MI Corp. may take any of the actions described in <u>clause (C)</u> with respect to any MI Acquisition Proposal if (1) MI Corp. has complied with this <u>Section 6.5</u>, (2) in the good faith judgment of the Board of Directors of MI Corp., after consulting with its outside legal counsel, the failure to take such action would be inconsistent with the fiduciary duties of the Board of Directors of MI Corp. to its shareholders under Applicable Laws and (3) prior to or concurrently with authorizing or permitting MI Corp. or any of its Subsidiaries to enter into any MI Acquisition Agreement, MI Corp. has terminated this Agreement pursuant to Section 8.1(h) and has paid the Termination Fee pursuant to <u>Section 8.2(c)</u>.

#### (d) Notices.

(i) The Board of Directors of MI Corp. shall not make any Change in the MI Recommendation in response to an MVT Acquisition Proposal unless (A) MI Corp. and its Subsidiaries shall not have breached this <u>Section 6.5</u>; (B) MI Corp. shall have provided prior written notice to Investor, at least five Business Days in advance, of its intention to take such action, which notice shall specify the material terms and conditions of such MVT Acquisition Proposal (including the identity of the Person making such proposal) and shall enclose a copy of the current version of the proposed transaction agreement, if any, with the Person making such proposal; and (C) prior to effecting such Change in the MI Recommendation, MI Corp. shall, and shall cause its financial and legal advisors to, during such five Business Day period, negotiate in good faith with Investor (to the extent Investor desires to negotiate) to make such adjustments to the terms and conditions of this Agreement as may be proposed by Investor; <u>provided</u>, <u>that</u> in the event of any material revision to any such MVT Acquisition Proposal, MI Corp. shall be required to deliver a new written notice to Investor and to again comply with the requirements of clause (B) with respect to such new written notice (except that the notice period shall be two Business Days instead of five Business Days).

(ii) In any case, and in addition to the notice requirements of Section 6.5(d)(i), if applicable, MI Corp. shall notify Investor in writing that a Change in the MI Recommendation has occurred no later than twenty four hours after a Change in the MI Recommendation has occurred (a "Change in Recommendation Notice"), which Change in Recommendation Notice shall set forth the principal reasons for the Change in the MI Recommendation. No later than fifteen Business Days after Investor receives such written notice, Investor shall provide the Force the MI Vote Notice to MI Corp. in accordance with Section 6.1(b) if it desires that the Transactions (including the Share Issuance and the MI Merger) be submitted to the shareholders of MI Corp. For the avoidance of doubt, the failure to provide such verbal and/or written notice shall in no way affect Investor's right to terminate this Agreement pursuant to Section 8.1(e).

(iii) If, at any time after MI Corp. has provided a Change in Recommendation Notice pursuant to <u>Section 6.5(d)(ii)</u> and this Agreement has not been terminated pursuant to Article VIII, the Board of Directors of MI Corp. shall make a further change in its recommendation to the shareholders of MI Corp. as to approval of this Agreement and the Transactions by the shareholders of MI Corp. based on subsequent material

developments, then (A) MI Corp. shall deliver a new Change in Recommendation Notice to Investor in accordance with <u>Section 6.5(d)(ii)</u>, (B) any prior Force the MI Vote Notice shall be of no further force or effect (other than that MI Corp. shall have no right to terminate this Agreement pursuant to <u>Section 8.1(h)</u> until the fifteen Business Day period referred to in clause (C) of this sentence has elapsed), and (C) Investor shall have a new fifteen Business Day period from the time of delivery of the notice pursuant to clause (A) in which to provide a Force the MI Vote Notice to MI Corp. in accordance with <u>Sections 6.1(b)</u> and <u>6.5(d)(ii)</u>.

(iv) For the avoidance of doubt, no notice shall be required to be made pursuant to this <u>Section 6.5(d)</u> at any time after this Agreement has been terminated pursuant to <u>Section 8.1(h)</u>.

(e) Nothing contained in this <u>Section 6.5</u> shall prohibit MI Corp. from complying with Rules 14d-9 or 14e-2 promulgated under the Exchange Act with respect to an MI Acquisition Proposal or from making any disclosure to the shareholders of MI Corp. with respect to an MI Acquisition Proposal or an MVT Acquisition Proposal, if, in the good faith judgment of the Board of Directors of MI Corp., after consultation with outside legal counsel, failure to make such disclosure would be inconsistent with the fiduciary duties of the Board of Directors of MI Corp. to its shareholders under Applicable Laws; <u>provided</u>, <u>however</u>, that compliance with such rules shall not in any way limit or modify the effect that any action taken pursuant to such rules has under any other provision of this Agreement, including <u>clause (B)</u> of <u>Section 6.5(c)</u>.

Section 6.6 Fees and Expenses. Without in any way limiting Sections 8.2(b) and 8.2(c), if the Closing does not occur, (i) all Expenses incurred by Investor shall be paid by Investor and (ii) all Expenses incurred by the MI Group or the MVT Group shall be paid by MI Corp. or MVT Corp., as the case may be. If the Closing occurs, all Expenses incurred by Investor on or prior to the Closing Date shall be paid by MVT Holding or MVT Corp. within three Business Days of the Closing Date by wire transfer of immediately available funds to an account specified in writing by Investor (and no MI Group member shall have any obligation with respect thereto), all Expenses incurred by MVT Group shall be paid by MVT Corp. and all Expenses incurred by the MI Group shall be paid by MI Corp. All Expenses of any party incurred after the Closing Date shall be paid by such party. As used in this Agreement, "Expenses," means all out-of-pocket expenses (including applicable filing and registration fees and all fees and expenses of counsel, accountants, investment bankers, printers, experts and consultants to a party hereto and its Affiliates) incurred by a party hereto or on its behalf in connection with or related to the authorization, preparation, negotiation, execution and performance of this Agreement, the Transaction Agreements and the Transactions and the transactions contemplated under the Transactions. All fees or expenses charged by Persons providing the Debt Financing to MVT Holding, MVT Corp. and their respective Subsidiaries, any interest expenses of the Debt Financing and any reasonable and documented out-of-pocket expenses of MI Corp., MVT Corp., MVT Holding or any of their respective Subsidiaries and, if the Closing occurs, Investor or the Equity Fund (including applicable filing and registration fees and all fees and expenses of AII Corp., MVT Corp., MVT Holding or any of their respective Subsidiaries and, if the Closing occurs, Investor or the Equity Fund (including applicable filing and registration fees and all fees a

hereto and its Affiliates), incurred in connection with the Debt Financing shall be borne by MVT Corp. or MVT Holding).

Section 6.7 <u>Public Announcements</u>. The parties shall each use reasonable best efforts to develop a joint communications plan and each party shall use reasonable best efforts (a) to ensure that all press releases and other public statements with respect to the Transactions shall be consistent with such joint communications plan and (b) unless otherwise required by Applicable Laws or by obligations pursuant to any listing agreement with or rules of any securities exchange or automated quotation system, to consult with each other before issuing any press release or, to the extent practicable, otherwise making any public statement with respect to this Agreement or the Transactions.

Section 6.8 <u>Takeover Statutes</u>. If any "fair price," "moratorium," "control share acquisition" or other form of antitakeover statute or regulation shall become applicable to the Transactions, each of MI Corp., Merger Sub, MVT Corp. or MVT Holding and their respective Boards of Directors shall use its reasonable best efforts to grant such approvals and take such actions as are reasonably necessary so that the Transactions may be consummated as promptly as practicable on the terms contemplated hereby and otherwise act to eliminate or minimize the effects of such statute or regulation on the Transactions.

Section 6.9 <u>Advice of Changes</u>. Each of Investor, MI Corp., MVT Corp. and MVT Holding shall as promptly as reasonably practicable after becoming aware thereof advise the others of any change or event (a) having, or which would, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on MVT Corp., or (b) which has resulted, or which, insofar as can reasonably be foreseen, would result, in any of the conditions set forth in <u>Article VII</u> not being satisfied; <u>provided</u>, <u>however</u>, that no such notification shall alter or affect in any manner the representations, warranties, covenants or agreements of the parties or the conditions to the obligations of the parties under this Agreement.

# Section 6.10 Private Letter Ruling; Tax-Free Reorganization Treatment; Pre-Distribution Tax Returns.

(a) MI Corp. shall use its reasonable best efforts to obtain the Private Letter Ruling as soon as practicable after the date hereof. MI Corp. shall (i) use its reasonable best efforts to allow Investor to participate in all meetings and material telephone calls with the IRS with respect to the Private Letter Ruling, (ii) use its reasonable best efforts to allow Investor a reasonable period of time (consistent with IRS requests as to timing) to review and comment on any material written submissions related to the Private Letter Ruling and incorporate any reasonable comments provided by Investor with respect thereto, (iii) provide to Investor a copy of any written submission to the IRS or any written material received from the IRS with respect to the Private Letter Ruling, and (iv) keep Investor fully informed of the status of IRS review and any material issue that arises during the course of such review and consult with Investor concerning such status and any such issue. MI Corp. and Investor agree that each will consider in good faith any reasonable modifications to the structure of the Transactions, such as adding new steps or altering the timing of steps set forth in the recitals and <u>Section 3.1</u> of this Agreement that will facilitate the receipt of the Private

Letter Rulings, so long as none of the Parties is materially prejudiced thereby and the anticipated Tax consequences of the Transactions set forth in the recitals are not affected.

(b) Neither MI Corp. nor any of its Subsidiaries shall take or cause to be taken any action, on or before the Closing Date, that (i) would result in any failure to obtain the Private Letter Ruling or (ii) could be reasonably expected to prevent the MI Merger and the MI Conversion from qualifying as a reorganization under Section 368(a) of the Code or the MI Contribution and the Share Distribution from qualifying as a reorganization under Section 368(a)(1)(D) of the Code and a distribution eligible for nonrecognition under Sections 355(a) and 361(c) of the Code. Investor will not, and will cause each Investor Tax Affiliate not to, (x) acquire any shares of MI Common Stock prior to the Closing Date or (y) acquire any shares of capital stock of MVT Holding or any of its Subsidiaries until the second anniversary of the Closing, other than the Shares and shares acquired in a transaction that complies with Section 4.02 of the Tax Allocation Agreement. For the avoidance of doubt, no member of the MI Group or the MVT Holding Group shall be considered an Affiliate of Investor for purposes of the preceding sentence.

(c) From the date hereof to the Share Distribution Time, MI Corp. shall timely file or cause to be timely filed all Tax Returns with respect to the MVT Group on a basis consistent with past practice (unless there has been a change in Applicable Laws).

(d) To the extent MVT Corp. or MVT Holding pays any of the Expenses as contemplated by the proviso in the first sentence of <u>Section 6.6</u> hereof, such payment shall be treated as an adjustment to the Purchase Price for U.S. federal income tax purposes.

Section 6.11 <u>Obligations under Separation Agreement</u>. Each of MI Corp., MVT Corp. and MVT Holding will prior to the Closing Date (a) perform its respective obligations and covenants under the Separation Agreement in accordance with the terms of the Separation Agreement and (b) enforce and preserve its respective rights under the Separation Agreement to the full extent permitted under the Separation Agreement.

Section 6.12 <u>Employee Benefits Matters</u>. MVT Holding shall provide employees of MVT Corp. and its Subsidiaries with benefits as set forth in the Employee Matters Agreement pursuant to and in accordance with the terms thereof.

#### Section 6.13 Debt Financing.

(a) MI Corp. and MVT Holding have received an executed commitment letter dated the date hereof pursuant to which J.P. Morgan Securities Inc. and Morgan Stanley Senior Funding, Inc. have each agreed to provide debt financing in the amount of \$1,750,000,000 (one billion seven hundred fifty million dollars) (the "Debt Financing") to MVT Holding and/or one or more of its Subsidiaries (the "Debt Commitment Letter"). MI Corp., MVT Corp., MVT Holdings and Investor each agrees

to use its reasonable best efforts to provide, and to cause its respective Subsidiaries and its and their respective officers, employees, independent auditors, counsel and other representatives to provide, all timely cooperation reasonably required in connection with the arrangement of the Debt Financing provided for in the Debt Commitment Letter; <u>provided</u>, <u>however</u>, that no member of the MI Group or any of their respective officers or employees shall be required to execute any document in connection with the Debt Financing and none of MVT Corp., MVT Holding or any of their respective Subsidiaries or any of their respective officers or employees shall be reflective at any time before the time that will be immediately prior to the Share Distribution Time unless an earlier time would be necessary in order to effect the Debt Financing in which case the applicable document shall be effective at such earlier time and such document shall expressly provide that if the Share Distribution does not occur, such document and each of its provisions shall be of no force or effect ab initio.

(b) MVT Corp., MVT Holding and MI Corp. shall use, and shall cause their respective Subsidiaries to use, their respective reasonable best efforts to enforce the provisions of the Debt Commitment Letter.

(c) If any portion of the Debt Financing becomes unavailable on the terms and conditions contemplated in the Debt Commitment Letter, Investor may arrange to obtain alternative financing from alternative sources following the occurrence of such event, in an amount sufficient (when taken together with the aggregate proceeds contemplated by the Equity Commitment Letter and the portion, if any, of the Debt Financing that remains available under the Debt Commitment Letter on the terms and conditions contemplated therein) to consummate the Transactions (the "<u>Alternative Debt Financing</u>"), and Investor, MI Corp. and its Subsidiaries (including MVT Corp.) shall use their respective reasonable best efforts to provide all timely cooperation reasonably required in connection with the arrangement of the Alternative Debt Financing in accordance with the provisions of <u>Section 6.13(a)</u>, mutatis mutandis; <u>provided</u>, <u>however</u>, that any such Alternative Debt Financing shall be on terms and subject to conditions reasonably acceptable to MVT Corp.

Section 6.14 Shareholders Agreement; Stock Purchase Right Agreement; Continuing Business Agreements.

(a) MVT Holding and Investor shall take all necessary action to, immediately prior to the Closing, execute and deliver the shareholders agreement in the form of <u>Exhibit E</u> (the "<u>Shareholders Agreement</u>").

(b) MVT Holding and Investor shall take all necessary action to, immediately prior to the Closing, execute and deliver the Stock Purchase Right Agreement in the form of Exhibit H.

(c) MI Corp., MVT Holding and MVT Corp., shall take all necessary action to cause any Continuing Business Agreements that have not been executed as of the date hereof to be executed and delivered prior to the Closing Date.

Section 6.15 Listing. Each of MI Corp. and MVT Holding shall use its reasonable best efforts to cause (a) the shares of MVT Holding Common Stock to be issued in the MI Merger and the shares of MVT Holding Common Stock to be reserved for issuance upon exercise of MVT Options and (b) the shares of New MI Corp. Common Stock to be distributed in the Share Distribution and the shares of New MI Corp. Common Stock to be reserved for issuance upon the exercise of New MI Corp. Options, to be approved for listing on the NYSE or on such other national securities exchange as MVT Holding and Investor may mutually agree, subject to official notice of issuance, prior to the Closing Date.

Section 6.16 <u>Investor Activity</u>. During the period from the date of this Agreement through the Closing Date, Investor shall not engage in any activity of any nature except as provided in or contemplated by this Agreement.

Section 6.17 Valuation Firm. MI Corp. has engaged American Appraisal Associates to act as a valuation or appraisal firm in connection with the Transactions (American Appraisal Associates or any valuation or appraisal firm engaged by MI Corp. in replacement of American Appraisal Associates with the consent of Investor (which shall not be unreasonably withheld or delayed), the "Valuation Firm") MI Corp. shall use its reasonable best efforts to obtain from the Valuation Firm an opinion addressed to the Board of Directors of MI Corp., MVT Corp. and MVT Holding dated as specified by MI Corp. after the date hereof and on or prior to the Closing Date to the effect that (a) with respect to the MVT Distribution (i) MI LLC will be able to pay its debts as they mature or become due (ii) the fair value of MI LLC's total assets will not be less than the sum of its total liabilities, (iii) MI LLC's assets will not be unreasonably small in relation to its business or the MVT Distribution and (iv) MI LLC will have property remaining that does not constitute unreasonably small capital; (b) with respect to the MI LLC Contribution and the MI Cash Contribution (i) MVT Holding will be able to pay its debts as they mature or become due (ii) the fair value of MVT Holding's total assets will not be less than the sum of its total liabilities, (iii) MVT Holding's assets will not be unreasonably small in relation to its business or the MI LLC Contribution and the MI Cash Contribution and (iv) MVT Holding will have property remaining that does not constitute unreasonably small capital; (c) with respect to the MVT Dividend (i) MVT Corp. will be able to pay its debts as they mature or become due (ii) the fair value of MVT Corp.'s total assets will not be less than the sum of its total liabilities, (iii) MVT Corp.'s assets will not be unreasonably small in relation to its business or the MVT Dividend and (iv) MVT Corp. will have property remaining that does not constitute unreasonably small capital; (d) with respect to the Share Distribution (i) MVT Holding will be able to pay its debts as they mature or become due (ii) the fair value of MVT Holding's total assets will not be less than the sum of its total liabilities, (iii) MVT Holding's assets will not be unreasonably small in relation to its business or the Share Distribution and (iv) MVT Holding will have property remaining that does not constitute unreasonably small capital; and (e) with respect to consummation of the Closing Transactions, each of MVT Holding, MVT Corp., New MI Corp. and MI LLC will (i) be able to pay each of their respective debts as they mature or become due (ii) the fair value and present fair saleable value of each of their respective assets will not be less than the sum of each of their respective aggregate liabilities; (iii) each of their respective assets will not be unreasonably small in relation to each of their respective businesses and (iv) each will have property remaining that does not constitute unreasonably small capital (the opinions in clauses (a), (b), (c), (d) and (e) collectively, the "Surplus and Solvency Opinions"). For the avoidance of doubt, MI Corp. and its Subsidiaries shall have no obligation

to make any capital contribution or loan to any member of the MVT Group or take any other action not expressly required by this Agreement in connection with this <u>Section 6.17</u>.

Section 6.18 <u>Merger Sub Activity</u>. During the period from the date of this Agreement through the Closing Date, neither Merger Sub nor MVT Holding shall engage in any activity of any nature except as provided in or contemplated by this Agreement.

Section 6.19 <u>Sole Shareholder Approval</u>. Promptly (and in any event no later than 48 hours) after the execution of this Agreement, MVT Holding, as the sole shareholder of Merger Sub, shall vote to adopt this Agreement in accordance with the WBCL.

Section 6.20 <u>Affiliates</u>. Not less than 30 days prior to the Closing Date, MI Corp. shall deliver to Investor and MVT Holding a letter identifying all Persons who, in the judgment of MI Corp., may be deemed at the Closing Date, "affiliates" of MVT Holding for purposes of Rule 145 under the Securities Act and applicable SEC rules and regulations, and such list shall be updated as necessary to reflect changes from the date of delivery thereof. MI Corp. shall use its reasonable best efforts to cause each person identified on such list to deliver to MVT Holding not less than 15 days prior to the Closing Date, a written agreement mutually agreeable to Investor and MI Corp.

Section 6.21 Non-Competition; Non-Solicitation.

#### (a) Non-Competition.

(i) MI Corp. agrees that for a period of 36 months following the Closing Date New MI Corp. shall not and shall cause its Subsidiaries not to engage, within the United States, in competition with members of MVT Group in any business with financial services firms and related businesses that MVT Corp. is engaged in as of the date hereof in any substantial respect (collectively, the "<u>Restricted Activities</u>"), <u>provided</u>, <u>that</u>, the Restricted Activities shall not include: (A) any business engaged in by members of MI Group as of the date hereof, (B) any activities engaged in solely to service the operations of the MI Group and (C) any other traditional banking services which shall include credit services, deposit taking, lending services, trust products or services, cash management services, wealth management products or services and financing commercial and residential real estates;

(ii) <u>Section 6.21(a)(i)</u> shall be deemed not to be breached as a result of (A) the ownership by New MI Corp. or any of its Subsidiaries of: (1) less than an aggregate of 5% of any class of capital stock of a Person engaged, directly or indirectly, in Restricted Activities; <u>provided</u>, <u>however</u>, that such capital stock is listed or quoted on a national securities exchange or the Nasdaq National Market, (2) less than 15% in value of any instrument of indebtedness of a Person engaged, directly or indirectly, in Restricted Activities; or (3) any securities held in a fiduciary capacity, (B) New MI Corp. or any of its Subsidiaries acquiring control of any Person or business that for the fiscal year immediately preceding such acquisition derived less than 10% of its revenues from Restricted Activities, (C) New MI Corp. or any of its Subsidiaries acquiring control of any Person or business that 10% of its revenues but less than 35% of its revenues from Restricted Activities so long as New MI Corp. and its Subsidiaries shall use its

reasonable best efforts to divest such operations as promptly as practicable and in any event within 12 months after the consummation of such acquisition of control, or (D) New MI Corp. or any of its Subsidiaries owning an interest acquired as a creditor in bankruptcy or otherwise than by a voluntary investment decision in a Person or business that for the fiscal year immediately preceding the acquisition of such interest by New MI Corp. or any of its Subsidiaries derived 10% or more of its revenues from Restricted Activities, so long as New MI Corp. or its applicable Subsidiary shall use its reasonable best efforts to divest such interest as promptly as practicable and in any event within 12 months after the acquisition of such interest; provided, however, that New MI Corp. or any of its Subsidiaries shall not be required to divest any such interest acquired pursuant to this clause (D) if its fair market value at the time of such acquisition is less than \$1,000,000. For the avoidance of doubt, New MI Corp. or any of its Subsidiaries may actively operate, manage and control any Person or business any of them acquire in accordance with clause (B) and/or clause (C).

(iii) MI Corp. and Investor agree that the covenants included in this <u>Section 6.21(a)</u> are reasonable in their geographic and temporal coverage, and that neither MI Corp. nor Investor shall raise any issue of geographic or temporal reasonableness in any proceeding to enforce such covenant; <u>provided</u>, <u>however</u>, that if the provisions of this <u>Section 6.21(a)</u> should ever be deemed to exceed the time or geographic limitations or any other limitations permitted by Applicable Law in any jurisdiction, then such provisions shall be deemed reformed in such jurisdiction to the minimum extent required by Applicable Law to cure such problem. Notwithstanding any other provision of this <u>Section 6.21(a)</u>, and that Investor shall be entitled to seek equitable relief, including the remedy of specific performance, with respect to any breach or attempted breach of such covenants.

(b) <u>MI Corp. Non-Solicitation</u>. MI Corp. agrees that for a period of six months following the Closing Date neither New MI Corp. nor any of its Subsidiaries will encourage any MVT Employee listed in <u>Section 6.21(b)</u> of the MVT Disclosure Schedule (such employees, the "<u>Executive Committee</u>") to terminate his or her employment with any member of the MVT Group, or solicit such individual for employment outside the MVT Group which would end or diminish such employee's service to the MVT Group.

(c) <u>MVT Holding Non-Solicitation</u>. MVT Holding and MVT Corp. each agrees that for a period of six months following the Closing Date neither MVT Holding nor any of its Subsidiaries will encourage any employee listed in <u>Section 6.21(c)</u> of the MI Disclosure Schedule to terminate his or her employment with any member of the MI Group or solicit such individual for employment outside the MI Group which would end or diminish that employee's services to the MI Group.

Section 6.22 <u>Transaction Agreements</u>. From the date hereof until the earlier of the Closing Date and the termination of this Agreement in accordance with <u>Article VIII</u>, without the written consent of Investor, none of MI Corp., MVT Corp., MVT Holding and Merger Sub shall (i) make, propose or agree to any amendment or modification of any of the Transaction Agreements to which any of them is a party or (ii) waive compliance with any obligations of any

party thereunder; provided, however, that MVT Corp. and MI Corp. may amend the Continuing Business Agreements in the ordinary course of business.

#### ARTICLE VII

# **CONDITIONS PRECEDENT**

Section 7.1 <u>Conditions to Each Party's Obligation to Effect the Transactions</u>. The respective obligations of Investor, MI Corp., MVT Corp. and MVT Holding to effect the Closing Transactions are subject to the satisfaction or waiver on or prior to the Closing Date of the following conditions:

(a) Shareholder Approval. MI Corp. shall have obtained the MI Transaction Approval.

(b) <u>No Injunctions or Restraints, Illegality</u>. No Applicable Laws shall have been adopted, promulgated or enforced by any Governmental Entity, and no temporary restraining order, preliminary or permanent injunction or other order issued by a court or other Governmental Entity of competent jurisdiction (an "<u>Injunction</u>") shall be in effect, having the effect of making the Transactions illegal or otherwise prohibiting consummation of the Transactions.

(c) <u>No Pending Governmental Actions</u>. No proceeding initiated by any Governmental Entity seeking an Injunction having the effect of making the Transactions illegal or otherwise prohibiting consummation of the Transactions shall be pending.

(d) <u>HSR Act</u>. The waiting period (and any extension thereof) applicable to the Transactions under the HSR Act shall have been terminated or shall have expired.

(e) <u>Effectiveness of the Form S-4</u>, Form 10 and the Proxy Statement. The Form S-4 shall have been declared effective by the SEC in accordance with the Securities Act, the Form 10 shall have been declared effective by the SEC and the Proxy Statement shall have been cleared by the SEC and none shall be the subject of any stop order or proceedings seeking a stop order.

(f) Private Letter Ruling and Tax Opinion.

(i) MVT Holding shall have received a private letter ruling from the IRS in form and substance reasonably satisfactory to each of MI Corp. and Investor to the effect that, on the basis of the facts, representations and assumptions set forth in the written request for such ruling which are consistent with the state of facts existing at the Share Distribution Time, (a) the MI Merger and the MI Conversion qualify as a reorganization under Section 368(a) of the Code and (b) the MI Contribution and the Share Distribution qualify as a reorganization under Section 368(a)(1)(D) of the Code and a distribution eligible for nonrecognition under Sections 355(a) and 361(c) of the Code (the "<u>Private Letter Ruling</u>") and such ruling shall, as of the

Closing Date, remain in full force and effect and shall not have been modified or amended in any respect adversely affecting the Tax consequences set forth therein.

(ii) MI Corp. and MVT Holding shall have received a reasoned opinion letter from Sidley Austin LLP (or other law firm of national standing) addressed to MI Corp. and MVT Holding, dated the Closing Date, such opinion letter to be in form and substance reasonably satisfactory to MI Corp. and Investor (including, for the avoidance of doubt, with respect to any underlying assumptions or representations), to the effect that, on the basis of facts, representations and assumptions set forth in such opinion which are consistent with the state of facts existing at the Share Distribution Time, the Share Distribution will (i) be treated as satisfying the business purpose requirement described in Treasury Regulation Section 1.355-2(b)(1); (ii) not be treated as being used principally as a device for the distribution of earnings and profits of the distributing corporation or the controlled corporation or both under Section 355(a)(1)(B) of the Code; and (iii) not be taxable as a result of the application of Section 355(e) of the Code. In rendering the opinions referred to in the preceding sentence, Sidley Austin LLP (or such other law firm of national standing) may rely as to matters of fact upon the representations contained herein and representations from MI Corp., MVT Corp., Investor and other parties (including, but not limited to, any major shareholders of MI Corp. and/or MVT Corp.) as Sidley Austin LLP (or such other law firm of national standing) may reasonably request for purposes of rendering such opinions.

(g) <u>Listing</u>. The (i) shares of MVT Holding Common Stock to be issued in the MI Merger and to be reserved for issuance upon exercise of MVT Options and (ii) the shares of New MI Corp. Common Stock to be distributed in the Share Distribution and the shares of New MI Corp. Common Stock to be reserved for issuance upon the exercise of New MI Corp. Options, shall each have been approved for listing on the NYSE or such other national securities exchange designated by MVT Holding and Investor pursuant to <u>Section 6.15</u>, subject to official notice of issuance.

(h) <u>Surplus and Solvency Opinions</u>. The Surplus and Solvency Opinions shall have been delivered in accordance with <u>Section 6.17</u> and such opinions shall not have been withdrawn, modified or rescinded.

(i) Banking Approval. The state and federal banking approvals set forth in Section 7.1(i) of the MVT Disclosure Schedule shall have been obtained.

(j) <u>Debt Financing</u>. MVT Holding shall have received the proceeds of the Debt Financing pursuant to the terms of the Debt Financing Commitment or, if the Investor has secured Alternative Debt Financing on terms reasonably acceptable to MVT Corp. pursuant to <u>Section 6.13(b)</u>, MVT Holding and/or one or more of its wholly-owned Subsidiaries shall have received the proceeds of the Alternative Debt Financing.

(k) <u>MVT Dividend and MI Cash Contribution</u>. Following the consummation of the Debt Financing (or Alternative Debt Financing, if applicable), MVT Corp., shall have funds sufficient to pay the MVT Dividend and to permit MVT Holding to make the MI Cash Contribution.

Section 7.2 <u>Additional Conditions to Obligations of Investor</u>. The obligations of Investor to effect the Closing Transactions are subject to the satisfaction or waiver by Investor on or prior to the Closing Date of the following additional conditions:

(a) <u>Representations and Warranties</u>. Each of the representations and warranties of MI Corp., MVT Corp. and MVT Holding (i) in <u>Section 4.2(a)</u> (Organization), <u>Sections 4.2(c)(i)</u> and (<u>v</u>) (Authority; No Conflicts), <u>Section 4.2(f)</u> (Brokers or Finders), <u>Section 4.3(a)</u> (Organization), <u>Section 4.3(c)(i)</u> and (<u>iii</u>) (Authority; No Conflicts), <u>Section 4.4(a)</u> (Organization), <u>Section 4.4(c)(i)</u> (Authority; No Conflicts), <u>Section 4.4(b)</u> (Capital Structure) and <u>Section 4.4(f)</u> (Status of MVT Holding Shares) shall be true and correct as of the date of this Agreement and as of the Closing Date as though made on and as of such dates (except to the extent that such representations and warranties speak only as of the date hereof or as of another date or dates in which case, only as of such date(s)), (ii) in <u>Section 4.2(b)(i)</u> (Capital Structure) shall be true and correct in all material respects as of the date of this Agreement and as of the Closing Date as though made on and as of such date(s)) and (iii) that is otherwise made in this Agreement shall be true and correct as of the date of this Agreement and as of the Closing Date as though made on and as of such date(s)) and (iii) that is otherwise made in this Agreement shall be true and correct as of the date of this Agreement and as of the Closing Date as though made on and as of such date(s)), in each case in this clause (iii), except for changes therein specifically permitted by this Agreement or any transaction contemplated by this Agreement or the Transaction Agreements and, in each case in this clause (iii), except where the failure of such representations or warranties to be true and correct (without giving effect to any limitation as to "material Adverse Effect" set forth in such representations and warranties) doe

(b) <u>Performance of Obligations of MI Corp., MVT Corp. and MVT Holding</u>. MI Corp., MVT Corp. and MVT Holding shall have each performed or complied with all agreements and covenants required to be performed by it under this Agreement at or prior to the Closing Date that are qualified as to materiality or Material Adverse Effect and shall have performed or complied in all material respects with all other agreements and covenants required to be performed by it under this Agreement at or prior to the Closing Date that are qualified as to be performed by it under this Agreement at or prior to the Closing Date that are not so qualified, and Investor shall have received a certificate executed by an executive officer of MI Corp., MVT Corp. and MVT Holding to such effect.

(c) <u>Transaction Agreements</u>. The Transaction Agreements contemplated to be in effect at such time shall be in full force and effect and each of MI Corp., MVT Corp. and MVT Holding shall have performed or complied with, in all material respects, the obligations required to be performed or complied with by it under the Transaction Agreements at or prior to the Closing Date.

(d) <u>Material Adverse Effect</u>. Since the date of this Agreement, there shall not have been any Material Adverse Effect on MVT Corp or MVT Holding.

Section 7.3 <u>Additional Conditions to Obligations of MI Corp.</u>, <u>MVT Corp. and MVT Holding</u>. The obligations of MI Corp., MVT Corp. and MVT Holding to effect the Closing Transactions are subject to the satisfaction or waiver by MI Corp. on or prior to the Closing Date of the following additional conditions:

(a) <u>Representations and Warranties</u>. Each of the representations and warranties of Investor (i) in <u>Section 4.1(a)</u> (Organization), <u>Section 4.1(b)(i)</u> (Authority; No Conflicts) and <u>Section 4.1(e) (Brokers or Finders</u>) shall be true and correct as of the date of this Agreement and as of the Closing Date as though made on and as of such dates (except to the extent that such representations and warranties speak only as of the date hereof or as of another date or dates in which case, only as of such date(s)), and (ii) that is otherwise made in this Agreement shall be true and correct as of the date of this Agreement and as of the Closing Date as though made on and as of such dates (except to the extent that such representations and warranties speak only as of the date of this Agreement and as of the Closing Date as though made on and as of such dates (except to the extent that such representations and warranties speak only as of the date hereof or as of another date or dates in which case, only as of such date(s)), in each case in this clause (ii), except for changes therein specifically permitted by this Agreement or any transaction contemplated by this Agreement or the Transaction Agreements and, in each case, except where the failure of such representations or warranties to be true and correct (without giving effect to any limitation as to "materiality" or "Material Adverse Effect" set forth in such representations and warranties) does not have and would not have, individually or in the aggregate, a Material Adverse Effect on Investor. MI Corp. and MVT Corp. shall have received a certificate of Investor executed by an executive officer of Investor to such effect.

(b) <u>Performance of Obligations of Investor</u>. Investor shall have performed or complied with all agreements and covenants required to be performed by it under this Agreement at or prior to the Closing Date that are qualified as to materiality or Material Adverse Effect and shall have performed or complied in all material respects with all other agreements and covenants required to be performed by it under this Agreement at or prior to the Closing Date that are qualified as the performed by it under this Agreement at or prior to the Closing Date that are not so qualified, and MI Corp., MVT Corp. and MVT Holding shall have received a certificate of Investor executed by an executive officer of Investor to such effect.

(c) <u>Transaction Agreements</u>. The Transaction Agreements contemplated to be in effect at such time shall be in full force and effect and Investor shall have performed or complied with, in all material respects, the obligations required to be performed or complied with by it under the Transaction Agreements at or prior to the Closing Date.

(d) Equity Financing. The proceeds of the Equity Financing shall have been deposited with MVT Holding in accordance with Section 2.2.

#### ARTICLE VIII

# TERMINATION AND AMENDMENT

Section 8.1 <u>Termination</u>. This Agreement may be terminated at any time prior to the Closing Date, by action taken or authorized by the Board of Directors of the terminating party or parties, and except as provided below, whether before or after the MI Transaction Approval:

(a) by the mutual written consent of MI Corp. and Investor;

(b) by either MI Corp. or Investor if the Closing Date shall not have occurred on or before April 4, 2008 (the "<u>Termination Date</u>"); <u>provided</u>, <u>however</u>, that the right to terminate this Agreement under this <u>Section 8.1(b)</u> shall not be available to any party that has breached in any material respect any of its obligations under this Agreement that has been the cause of, or resulted in, the failure of the Closing Date to occur on or before the Termination Date;

(c) by either MI Corp. or Investor if any Governmental Entity shall have issued an order, decree or ruling or taken any other action (which action such party shall have used its reasonable best efforts to resist, resolve or lift, as applicable, in accordance with <u>Section 6.4</u>) permanently restraining, enjoining or otherwise prohibiting the Transactions, and such order, decree, ruling or other action shall have become final and nonappealable;

(d) by either MI Corp. or Investor if at the MI Shareholders Meeting the MI Transaction Approval shall not have been obtained;

(e) by Investor, if (i) (A) the Board of Directors of MI Corp. shall have failed to make the MI Recommendation or (B) the Board of Directors of MI Corp. or a committee thereof shall have made a Change in the MI Recommendation (or publicly announced its intention to take any such action referred to in <u>clause (A)</u> or <u>(B)</u>) and Investor has not delivered a Force the MI Vote Notice in accordance with <u>Section 6.1(b)</u> that remains in effect in accordance with <u>Section 6.5(d)(iii)</u>, or (ii) MI Corp. shall have breached its obligations under this Agreement by reason of a failure to call and hold the MI Shareholders Meeting in accordance with <u>Section 6.1(b)</u>;

(f) by MI Corp., if Investor shall have breached or failed to perform any of its representations, warranties, covenants or other agreements contained in this Agreement, such that the conditions set forth in <u>Section 7.3(a)</u> or <u>Section 7.3(b)</u> are not capable of being satisfied on or before the Termination Date;

(g) by Investor, if MVT Holding, MI Corp. or MVT Corp. shall have breached or failed to perform any of its representations, warranties, covenants or other agreements contained in this Agreement, such that the conditions set forth in <u>Section 7.2(a)</u> or <u>Section 7.2(b)</u> are not capable of being satisfied on or before the Termination Date; or

(h) by MI Corp., if (i) (A) the Board of Directors of MI Corp. authorizes MI Corp. to enter into a MI Acquisition Agreement and (B) MI Corp. has paid to Investor the Termination Fee pursuant to <u>Section 8.2(c)</u> or (ii) (x) the Board of Directors of MI Corp. makes a Change in Recommendation in accordance with <u>Section 6.5</u>, (y) Investor has not delivered a Force the MI Vote Notice in accordance with <u>Section 6.5(d)(iii)</u> and the fifteen Business Day period for delivery of a Force the MI Vote Notice pursuant to <u>Section 6.5(d)(ii)</u> or (<u>iii</u>) has elapsed and (z) MI Corp. has paid to Investor the Termination Fee pursuant to <u>Section 8.2(c)</u>.

#### Section 8.2 Effect of Termination.

(a) In the event of termination of this Agreement by either MI Corp. or Investor as provided in <u>Section 8.1</u>, this Agreement shall forthwith become void and there shall be no liability or obligation on the part of Investor, MI Corp., Merger Sub, MVT Holding or MVT Corp. or their respective Subsidiaries, officers or directors under this Agreement, except that the second and third sentences of <u>Section 6.3</u> (regarding the Confidentiality Agreement), <u>Section 6.6</u> (Fees and Expenses), this <u>Section 8.2</u> and <u>Article IX</u> shall survive such termination and <u>provided</u> that nothing herein shall relieve any party from liability for its willful breach of this Agreement.

(b) If Investor shall terminate this Agreement pursuant to Section 8.1(e), MI Corp. shall pay Investor an amount equal to \$75,000,000.

(c) If MI Corp. shall terminate this Agreement pursuant to Section 8.1(h), MI Corp. shall pay Investor an amount equal to \$75,000,000.

(d) If (i) (A) either party shall terminate this Agreement pursuant to <u>Section 8.1(b)</u> or (B) Investor shall terminate this Agreement pursuant to <u>Section 8.1(g)</u> and (ii) at any time after the date of this Agreement and before any such termination, a bona fide MI Acquisition Proposal or MVT Acquisition Proposal shall have been publicly announced, become publicly known or otherwise been communicated to the senior management, the Board of Directors or stockholders of MI Corp. (whether or not conditional), then if within twelve months after such termination, MI Corp. or any of its Subsidiaries enters into a definitive agreement with respect to a MI Acquisition Proposal or MVT Acquisition Proposal, or consummates any MI Acquisition Proposal or MVT Acquisition Proposal, MI Corp. shall pay Investor an amount equal to \$75,000,000.

(e) If (i) either party shall terminate this Agreement pursuant to <u>Section 8.1(d)</u>, MI Corp. shall pay Investor an amount equal to \$20,000,000 and (ii) at any time after the date of this Agreement and before the MI Shareholder Meeting, a bona fide MI Acquisition Proposal or MVT Acquisition Proposal (other than by Investor or any Affiliate thereof) shall have been publicly announced, become publicly known or otherwise been communicated to the senior management, the Board of Directors or stockholders of MI Corp. (whether or not conditional), then if within twelve months after such termination, MI Corp. or any of its Subsidiaries enters into a definitive agreement with respect to a MI Acquisition Proposal or MVT Acquisition Proposal, or

consummates any MI Acquisition Proposal or MVT Acquisition Proposal, MI Corp. shall pay Investor an amount equal to \$55,000,000 in addition to the amount paid to Investor pursuant to <u>clause (i)</u> above.

(f) If this Agreement is terminated by MI Corp. pursuant to <u>Section 8.1(f)</u> on the basis of an intentional and material breach by Investor, Investor shall pay to MI Corp. an amount equal to \$75,000,000.

(g) If this Agreement is terminated by Investor pursuant to <u>Section 8.1(g)</u> on the basis of an intentional and material breach by MI Corp., MI Corp. shall pay to Investor an amount equal to \$75,000,000.

(h) Any amount payable under <u>clause (b)</u> through (g) of this <u>Section 8.2</u> is referred to herein as the "<u>Termination Fee</u>." The Parties hereby acknowledge and agree that the amounts payable pursuant to <u>clauses (b)</u> through (g) are not cumulative, that the Termination Fee shall be paid only under one subsection of this <u>Section 8.2</u> and that in no event shall the Termination Fee exceed \$75,000,000. All payments to be made pursuant to this <u>Section 8.2</u> shall be made by the applicable party as promptly as reasonably practicable (and, in any event, within three Business Days) following the date of termination of this Agreement pursuant to this <u>Article VIII</u> by wire transfer of immediately available funds to an account specified in writing by the party to receive such payments; <u>provided</u>, <u>however</u>, that any payment to be made pursuant to <u>Section 8.2(c)</u> shall be made to Investor prior to the termination of this Agreement pursuant to <u>Section 8.1(h)</u> by wire transfer of immediately available funds to an account specified in writing by Investor.

(i) The parties acknowledge and agree that, notwithstanding any other provision of this Agreement, (i) in no event shall the Investor, on the one hand, or any member of the MI Group or the MVT Group, on the other hand, be liable for, or seek to recover against the other party, any losses or damages with respect to this Agreement in excess of \$75,000,000 (the "<u>Cap</u>"), (ii) the payment of the Cap by the Investor or the Equity Fund (in the case of the Equity Fund pursuant to the Limited Guarantee), on the one hand, or any member of the MI Group or the MVT Group, on the other hand, pursuant to this <u>Section 8.2</u> shall be the sole and exclusive remedy of such party against the other party and such other party's respective stockholders, partners, members, directors, officers, employees or agents for any losses or damages suffered such party as a result of the failure of the Transactions to be consummated or the termination of this Agreement, and (iii) upon payment of the Cap by the Investor or the Equity Fund pursuant to the Limited Guarantee), on the one hand, or any member of the MI Group or the MVT Group, on the other hand, pursuant to the Limited Guarantee), on the one hand, or any member of the MI Group or the MVT Group, on the other hand, pursuant to the Limited Guarantee), on the one hand, or any member of the MI Group or the MVT Group, on the other hand, pursuant to this <u>Section 8.2</u> none of Investor or the Equity Fund, on the one hand, or any member of the MI Group or the MVT Group, on the other hand, or any of their respective stockholders, partners, members, directors, officers, employees or agents, as the case may be, shall have any further liability or obligation relating to or arising out of this Agreement or the Transactions. Notwithstanding the foregoing, the parties acknowledge that if any party fails to pay promptly any amount due pursuant to and in accordance with this <u>Section 8.2</u>, and, in order to obtain such payment, the other party

commences a suit which results in a final, binding and nonappealable judgment against the breaching party for the fee set forth in this <u>Section 8.2</u>, the breaching party shall pay to the other party its reasonable attorneys' fees and expenses in connection with such suit; <u>provided</u>, <u>that</u>, the amounts payable by either party pursuant to this sentence shall in no event exceed \$2,000,000. The parties acknowledge that the agreements contained in this <u>Section 8.2</u> are an integral part of the Transactions, and that, without these agreements, the parties would not enter into this Agreement.

# ARTICLE IX

# **GENERAL PROVISIONS**

Section 9.1 <u>Non-Survival of Representations and Warranties</u>. The representations and warranties in this Agreement shall not survive the Closing Date. This <u>Section 9.1</u> shall not limit any covenant or agreement of the Parties which by its terms contemplates performance after the Closing Date.

Section 9.2 <u>Notices</u>. All notices and other communications hereunder shall be in writing and shall be deemed duly given (a) on the date of delivery if delivered personally, (b) upon confirmation of receipt if delivered by telecopy or telefacsimile, (c) on the first Business Day following the date of dispatch if delivered by a recognized next-day courier service, or (d) on the date received if delivered by registered or certified mail, return receipt requested, postage prepaid. All notices hereunder shall be delivered as set forth below, or pursuant to such other instructions as may be designated in writing by the party to receive such notice:

(a) if to MI Corp. or Merger Sub or, prior to the Closing Date, to MVT Corp. or MVT Holding to:

Marshall & Ilsley Corporation 770 N. Water Street Milwaukee, Wisconsin 53202 Fax: (414) 765-7899 Attention: Dennis J. Kuester Chairman and Chief Executive Officer Randall J. Erickson Senior Vice President, General Counsel and Corporate Secretary

with a copy to:

Sidley Austin LLP One S. Dearborn Street Chicago, Illinois 60603 Fax: (312) 853-7036 Attention: Imad I. Qasim, Esq. Pran Jha, Esq.

(b) if to MVT Corp. or MVT Holding, to:

Metavante Holding Company 4900 West Brown Deer Rd. Milwaukee, Wisconsin 53223 Fax: (414) 362-1705 Attention: Frank Martire Chief Executive Officer Norrie J. Daroga Executive Vice President, Chief Risk Officer & Secretary with a copy to:

Quarles & Brady LLP 411 East Wisconsin Avenue Milwaukee, Wisconsin 53202-4497 Fax: (414) 978-8786 Attention: Patrick M. Ryan, Esq.

(c) if to Investor, to:

WPM, L.P. c/o Warburg Pincus & Co. 466 Lexington Avenue New York, New York 10017 Fax: (212) 878-9351 Attention: James Neary

with a copy to:

Wachtell, Lipton, Rosen & Katz 51 West 52nd Street New York, New York 10019 Fax: (212) 403-2000 Attention: Andrew R. Brownstein, Esq. Igor Kirman, Esq.

or to such other persons or addresses as may be designated in writing by the party to receive such notice as provided above.

Section 9.3 <u>Amendment</u>. Subject to Applicable Law, this Agreement may be amended by the parties hereto, by action taken or authorized by their respective Boards of Directors, at any time before or after approval of the matters presented in connection with this Agreement and the Transactions by the shareholders of MI Corp., but, after any such approval, no amendment shall be made which by law or in accordance with the rules of any relevant stock

exchange requires further approval by such shareholders without such further approval. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto.

Section 9.4 <u>Extension; Waiver</u>. Subject to Applicable Law, at any time prior to the Closing Date, the parties hereto, by action taken or authorized by their respective Boards of Directors, may, to the extent legally allowed, (a) extend the time for the performance of any of the obligations or other acts of the other parties hereto, (b) waive any inaccuracies in the representations and warranties of other parties contained herein or in any document delivered pursuant hereto and (c) waive compliance with any of the agreements or conditions of other parties contained herein or in any document delivered pursuant hereto. Any agreement on the part of a party hereto to any such extension or waiver shall be valid only if set forth in a written instrument signed on behalf of such party. The failure of any party to this Agreement to assert any of its rights under this Agreement or otherwise shall not constitute a waiver of those rights.

Section 9.5 <u>Interpretation</u>. When a reference is made in this Agreement to Sections, Exhibits or Schedules, such reference shall be to a Section of or Exhibit or Schedule to this Agreement unless otherwise indicated. The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation."

Section 9.6 <u>Counterparts</u>. This Agreement may be executed in multiple counterparts, all of which shall be considered one and the same agreement and shall become effective when two or more counterparts have been signed by each of the parties and delivered to the other parties, it being understood that the parties need not sign the same counterpart.

#### Section 9.7 Entire Agreement; No Third Party Beneficiaries.

(a) This Agreement, the Confidentiality Agreement, the Transaction Agreements and the exhibits and schedules hereto and thereto and the other agreements and instruments of the parties delivered in connection herewith and therewith constitute the entire agreement and supersede all prior agreements, understandings, representations and warranties, both written and oral, among the parties with respect to the subject matter hereof and thereof.

(b) This Agreement shall be binding upon and inure solely to the benefit of each party hereto, and nothing in this Agreement, express or implied, is intended to or shall confer upon any other Person any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.

Section 9.8 <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of Wisconsin (without giving effect to choice of law principles thereof).

Section 9.9 <u>Severability</u>. If any provision of this Agreement or the application thereof to any Person or circumstance is determined by a court of competent jurisdiction to be invalid, void or unenforceable, the remaining provisions hereof, or the application of such

provision to Persons or circumstances other than those as to which it has been held invalid or unenforceable, shall remain in full force and effect and shall in no way be affected, impaired or invalidated thereby, so long as the economic or legal substance of the Transactions is not affected in any manner materially adverse to any Party. Upon any such determination, the parties shall negotiate in good faith in an effort to agree upon a suitable and equitable substitute provision to effect the original intent of the Parties.

Section 9.10 <u>Assignment</u>. This Agreement shall not be assignable by any party without the prior written consent of the other parties; <u>provided</u>, <u>however</u>, that Investor may assign all of its rights hereunder to the Equity Fund or any Successor Fund or any entity wholly owned (other than de minimis (no more than 1%) ownership by any other Person or Persons, provided, that, such assignment would not be reasonably likely to prevent or impair or delay the consummation of the Transactions) by the Equity Fund and/or any Successor Fund with the prior written consent of MI Corp., which consent will not be unreasonably withheld, conditioned or delayed; <u>provided</u>, <u>further</u>, that (i) MI Corp. shall not be required to give its consent to any such assignment that would delay or impair the Closing, (ii) notwithstanding any such assignment, Investor shall remain liable to perform all of its obligations hereunder, (iii) the obligations of MI Corp., MVT Corp. and MVT Holding hereunder shall be subject to delivery by such assignee, on or prior to the Closing Date, of a certificate signed on its behalf containing substantially similar representations and warranties to those contained in <u>Section 4.1</u> and (iv) Investor shall take all actions requested by MI Corp. in order to ensure that the Limited Guarantee and the Equity Commitment Letter remain in full force and effect after giving effect to such assignment. Subject to the foregoing, this Agreement shall be binding upon, inure to the benefit of, and be enforceable by the parties and their respective successors and assigns. If Investor requests that a portion of its rights hereunder be permitted to be assigned to any Person, MI Corp. will, subject to its consent rights, endeavor in good faith to accommodate such request.

Section 9.11 <u>Submission to Jurisdiction; Waivers</u>. (a) Each of Investor, MI Corp., MVT Corp. and MVT Holding irrevocably agrees that any legal action or proceeding with respect to this Agreement, the Transactions, any provision hereof, the breach, performance, validity or invalidity hereof or for recognition and enforcement of any judgment in respect hereof brought by another party hereto or its successors or permitted assigns may be brought and determined in any federal or state court located in the State of Wisconsin, and each of Investor, MI Corp., Merger Sub, MVT Corp. and MVT Holding hereby irrevocably submits with regard to any such action or proceeding for itself and in respect to its property, generally and unconditionally, to the exclusive jurisdiction of the aforesaid courts.

(b) Each of Investor, MI Corp., Merger Sub, MVT Corp. and MVT Holding hereby irrevocably waives, and agrees not to assert, by way of motion, as a defense, counterclaim or otherwise, in any action or proceeding with respect to this Agreement, the Transactions, any provision hereof or the breach, performance, enforcement, validity or invalidity hereof, (i) any claim that it is not personally subject to the jurisdiction of the above-named courts for any reason other than the failure to lawfully serve process, (ii) that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise) and (iii) to the fullest extent permitted

by Applicable Laws, that (A) the suit, action or proceeding in any such court is brought in an inconvenient forum, (B) the venue of such suit, action or proceeding is improper and (C) this Agreement, or the subject matter hereof, may not be enforced in or by such courts. EACH PARTY FURTHER ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (a) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, TO IT THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (b) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (c) EACH PARTY MAKES THIS WAIVER VOLUNTARILY AND (d) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS <u>SECTION 9.11</u>.

Section 9.12 <u>Enforcement</u>. The parties agree that irreparable damage would occur in the event that any of the provisions of <u>Section 6.21</u> of this Agreement, the third to last sentence of <u>Section 6.3</u>, or the Confidentiality Agreement were not performed in accordance with their specific terms. It is accordingly agreed that the parties shall be entitled to pursue specific performance of the terms of <u>Section 6.21</u> of this Agreement, the third to last sentence of <u>Section 6.3</u>, or the Confidentiality Agreement of the terms of <u>Section 6.21</u> of this Agreement, the third to last sentence of <u>Section 6.21</u> of this Agreement, the being in addition to any other remedy to which they are entitled at law or in equity. Other than <u>Section 6.21</u> of this Agreement, the third to last sentence of <u>Section 6.21</u> of this Agreement, the third to last sentence of <u>Section 6.21</u> of this Agreement, the parties shall not be entitled to pursue specific performance or other equitable relief with respect to any provision of this Agreement.

Section 9.13 <u>Disclosure Schedule</u>. The mere inclusion of an item in the relevant Disclosure Schedule as an exception to a representation, warranty or covenant shall not be deemed an admission by a party that such item represents a material exception or material fact, event or circumstance or that such item has had or would have a Material Adverse Effect with respect to MI Corp., Merger Sub, MVT Corp. or any Subsidiary of the foregoing, as applicable.

Section 9.14 <u>Mutual Drafting</u>. This Agreement shall be deemed to be the joint work product of Investor, MI Corp., Merger Sub, MVT Corp. and MVT Holding and any rule of construction that a document shall be interpreted or construed against a drafter of such document shall not be applicable.

# [REMAINDER OF PAGE INTENTIONALLY LEFT BLANK.]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective officers thereunto duly authorized, all as of the date first written above.

# MARSHALL & ILSLEY CORPORATION

By:	/s/ Mark F. Furlong	
Name:	Mark F. Furlong	
Title:	0	
METAVANTE CORPORATION		
P		
By:	5	
Name:		
Title:	Senior Executive Vice President	
METAVANTE HOLDING COMPANY		
METAVALUTE HOLDING COMPANY		
By:	/s/ Gregory A Smith	
Name:	Gregory A. Smith	
Title:		
MONTANA MERGER SUB INC.		
By:	/s/ Gregory A Smith	
Name:	Gregory A. Smith	
Title:	President	
WPM, L.P.		
By: WPM GP, LLC, its general partner		
Dy. mini or, DDo, no general paralet		
By:	/s/ James Neary	
Name:	James Neary	
Title:	Managing Director	

# <u>Exhibit G</u>

# Calculation of Shares

See Attached.

#### **Investor Share Number Calculation**

For purposes of the Parties' determination pursuant to Section 2.3, the "*Investor Share Number*" shall be calculated using the spreadsheet formula indicated as part of this Exhibit G. The variables used in the formula shall be determined and defined in the following order and manner:

*First*, the Parties shall determine the "*M&I Pre-Distribution Stock Price*", which shall mean the average of the closing prices per share of M&I Common Stock over the three trading days ending on the third trading day before (i) the Share Distribution Time or (ii) if earlier, the date on which New MI Corp. Common stock begins to trade "ex-distribution". Solely for purposes of illustration, the M&I Pre-Distribution Stock Price in Exhibit G shall be \$50.00 per share.

Second, the Parties shall calculate the "Investor Fraction" by dividing (i) the "Investor Ownership Percentage", which shall be 25%, by (ii) the "Metavante Ownership Percentage", which shall be 75%.

*Third*, the Parties shall calculate the "*New Metavante Basic Shares*," which shall equal the then-current number of issued and outstanding shares of MI Corp. *divided* by the "*Reverse Split Adjustment Factor*". For the purposes of this Agreement, the reverse split adjustment factor shall equal three.

*Fourth*, the Parties shall determine the "*New Metavante Restricted Shares*", which shall equal the sum of the following amount, in each case adjusted as provided for in the Employee Matters Agreement: (i) the total number of "restricted shares" of MI Common Stock not associated with deferred stock compensation plans, (ii) the total number of "restricted shares" associated with deferred stock compensation plans and attributable only to MI Corp. employees, and (iii) the total number of "restricted shares" associated with deferred stock compensation plans and attributable only to MVT Corp. employees.

*Fifth*, the Parties shall calculate the Investor Share Number by *multiplying* the Investor Fraction by the sum of (i) New Metavante Basic Shares and (ii) New Metavante Restricted Shares.

## **Illustrative Calculation of Investor Share Number**

		Step #
M&I Pre-Distribution Stock Price	\$ 50.00	1
Investor Ownership Percentage	25%	2
Metavante Ownership Percentage	75%	2
Investor Fraction	33%	2
Current issued and outstanding M&I shares	255.719948	3
Reverse Split Adjustment Factor	3.0	3
New Metavante Basic Shares	85.239983	3
New Metavante Restricted Shares	0.321031	4
Investor Share Number	28.52033783	5

# Illustrative of Investor Share Number \$ millions

				Check	Step #
M&I Pre-Distribution Stock Price			\$50.00		1
Implied Metavante Equity Value			\$2,500		2
Investment Amount			\$ 625	\$ 625	2
Investor Ownership Percentage			25%		2
Metavante Ownership Percentage			75%		3
Investor Fraction			33%		3
Allocation Percentage			25%		4
Current issued and outstanding M&I shares			255.7		5
Reverse Split Adjustment Factor			3.0		5
New Metavante Basic Shares			85.2		5
New Metavante Restricted Shares			0.3		6
Investor Share Number			28.5		7
New Metavante Management Stock Options			10.5		8
New Metavante Management Stock Options Weighte	ed Average Exercis	se Price	\$16.28		9
Director Stock Options			1.0		10
Director Stock Options Weighted Average Exercise F	Price		\$14.69		11
Investor Option Number			3.9		12
Investor Option Weighted Average Exercise Price			\$16.14		12
New Metavante Stock Options			15.4		13
Implied Metavante Share Price			\$21.23		14
Treasury Stock Method Dilution of New Metavante S	Stock Options		3.7		15
New Metavante Implied Fully Diluted Shares			117.8	\$2,500	16
Memo items			Comments		
Adjustment factor		1,000,000			
Pre-spin options		4.5			
M&I Pre-Distribution Stock Price	\$	50.00			
Montana mgmt pre-closing money-at-work	\$	224	Pre-spin options x M&I Pre-Distril	oution Stock P	rice
Implied Metavante Share Price	\$	21.23			
Post-spin options	Ψ	10.5			
Montana mgmt post-closing money at work	\$	224	New Metavante Stock Options x Ir	nnlied Metava	nto Sharo
Montana mynit post-ciosing money at work	ψ	224	Price	iipiieu wietava	ine Share
Martin and the state of the sta					
Montana management intrinsic value	¢	FD 1			
Pre-spin	\$	52.1	(Share price – exercise price) x # o		
Post-spin	\$	52.1	(Share price – exercise price) x # o	ptions	
Overhang		13.10%	Total dilutive securities / total New Diluted Shares	v Metavante Fu	ılly

Investor Share Number—WP Warrants	
New Metavante Basic Shares	85.2
New Metavante Restricted Shares	0.3
Investor Fraction	33%
Investor Share Number	28.5
Investor PF Ownership CHECK	
Investor Common Share Number	28.5
PF MVT Basic Shares + Restricted Shares	114.1
% Investor Ownership	25.0000%
Investor warrants	3.9
PF MVT Options + Warrants	15.4
% Investor Ownership	25.0000%
Investor Share Number	28.5
Investor Net Warrants (Treasury Method)	0.9
Total Net Investor Shares	29.4
New Metavante Implied Fully Diluted Shares	117.8
% Investor Ownership	25.0000%

# METAVANTE TECHNOLOGIES, INC.

SHAREHOLDERS AGREEMENT

Dated as of November 1, 2007

# Table of Contents

# ARTICLE I

Page

# GOVERNANCE

1.1	Composition of the Board of Directors	2
1.2	Committees	4
1.3	Articles of Incorporation and By-laws	4
1.4	Approval Rights	4
1.5	Venture Capital Qualifying Investment	5
1.6	Termination of Article I	5

# ARTICLE II

# REGISTRATION RIGHTS

2.1	Demand Registrations	6
2.2	Piggyback Registrations	9
2.3	Registration Procedures	10
2.4	Registration Expenses	13
2.5	Participation in Underwritten Registrations	13
2.6	Rule 144; Legended Securities; etc	14
2.7	Holdback	15

# ARTICLE III

# TRANSFERS; STANDSTILL PROVISIONS; PREEMPTIVE RIGHTS

3.1	Investor Group Transfer Restrictions	15
3.2	Standstill Provisions	16
3.3	Anti-Takeover Provisions	17
3.4	Buyout Transactions	18
3.5	Preemptive Rights	18
	ARTICLE IV	
	INDEMNIFICATION	
4.1	Indemnification	21
	ARTICLE V	
	DEFINITIONS	
5.1	Defined Terms	24
	i	

5.2 Terms Generally

# ARTICLE VI

# MISCELLANEOUS

6.1	Term	30
6.2	No Inconsistent Agreements	30
6.3	Legend	30
6.4	Amendments and Waivers	31
6.5	Successors and Assigns	31
6.6	Severability	32
6.7	Counterparts	32
6.8	Descriptive Headings	32
6.9	Governing Law	32
6.10	Consent to Jurisdiction	32
6.11	Waiver of Jury Trial	32
6.12	Enforcement; Attorneys' Fees	33
6.13	No Third Party Beneficiaries	33
6.14	Notices	33
6.15	Entire Agreement	34
	ii	

Page

SHAREHOLDERS AGREEMENT, dated as of November 1, 2007 (as it may be amended from time to time, this "<u>Agreement</u>"), among (<u>i</u>) Metavante Technologies, Inc., a Wisconsin corporation (the "<u>Company</u>"), (<u>ii</u>) WPM, L.P., a Delaware limited partnership ("<u>Investor</u>"), and (<u>iii</u>) any other Shareholder that may become a party to this Agreement after the date and pursuant to the terms hereof.

## $\underline{W} \underline{I} \underline{T} \underline{N} \underline{E} \underline{S} \underline{S} \underline{E} \underline{T} \underline{H}$ :

WHEREAS, pursuant to an Investment Agreement, dated as of April 3, 2007 (the "<u>Investment Agreement</u>"), among the Company, Marshall & Ilsley Corporation, a Wisconsin corporation ("<u>MI Corp.</u>"), New M&I Corporation, a Wisconsin corporation, Metavante Corporation, a Wisconsin corporation, and Investor, Investor has agreed to acquire, on the terms and subject to the conditions set forth in such agreement, (<u>i</u>) newly issued shares of the Class A common stock, par value \$0.01 per share (the "<u>Class A Common Stock</u>") of the Company and (<u>ii</u>) certain purchase rights with respect to shares of Common Stock pursuant to the Stock Purchase Right Agreement, dated as of the date hereof, between the Company and Investor ("<u>Purchase Rights</u>") (such transaction, the "<u>Investment</u>");

WHEREAS, as of the date hereof, Investor will own 29,732,214 shares of Class A Common Stock;

WHEREAS, at 12:01 a.m. Eastern Standard Time on the first day following the date hereof, each outstanding share of Class A Common Stock held by Investor shall automatically convert into a share of Common Stock;

WHEREAS, it is a condition to the consummation of the transactions contemplated by the Investment Agreement that the Company execute and deliver this Agreement; and

WHEREAS, each of the parties hereto wishes to set forth in this Agreement certain terms and conditions regarding the Investment and the ownership of shares of Common Stock, including certain registration rights applicable to such shares, restrictions on the transfer of such shares, restrictions on certain actions relating to the Company, and the management of the Company and its subsidiaries.

NOW, THEREFORE, in consideration of the mutual agreements contained herein, the parties hereto hereby agree as follows:

### ARTICLE I

#### GOVERNANCE

1.1 Composition of the Board of Directors. (a) The by-laws of the Company shall provide that so long as this Article I is in effect the Board of Directors of the Company (the "Board") shall consist of eleven directors, such directors to be nominated and elected in accordance with this Agreement and the provisions of the by-laws of the Company. As of the Closing Date, the directors shall consist of (i) three directors designated by Investor (such designees and any persons nominated pursuant to Section 1.1(b) and elected as directors and any persons designated as replacement directors for such designees or their replacements pursuant to Section 1.1(c), the "Investor Designees"), (ii) two directors who shall be officers of the Company, one of whom shall be the President and Chief Executive Officer of the Company and one of whom shall be the Senior Vice President and Chief Operating Officer of the Company, (iii) one director who shall be designated by MI Corp. and shall initially be Dennis J. Kuester (such designee and any person designated as a replacement director for such designee or their replacement pursuant to Section 1.1(d), the "MI Designee"), and (iv) five additional directors designated pursuant to Section 6.2 of the Investment Agreement, each of whom shall qualify as Independent Directors and one of whom shall also be a director of MI Corp. (such designees, any persons nominated and elected as directors or designated as replacement directors for such designees or their replacements pursuant to Section 1.1(d), the "Initial Unaffiliated Directors"); provided, however, that if Investor or MI Corp. is prevented by Applicable Law or regulatory process from designating any of its designees pursuant to the foregoing clause (i), (iii) or (iv) (in the case of a MI Corp. director), as applicable, or if such designation is otherwise prohibited by Section 6.2(a) of the Investment Agreement (because such designation would result in the Company being an affiliate of New MI Corp. for purposes of Section 23A or 23B of the Federal Reserve Act), then such directors shall be Independent Directors selected pursuant to the foregoing clause (iv) in a manner which addresses the reason that the designee was originally prevented from being designated. The Chairman of the Board of the Company shall be Dennis J. Kuester for a period of one year from the date hereof. If Dennis J. Kuester is unable to serve as Chairman of the Board during such one-year period, and after such one-year period, the President and Chief Executive Officer of the Company shall, subject to the approval of the Board, succeed Dennis J. Kuester as the Chairman of the Board. In connection with the 2008 annual meeting of the Company, the Company shall take all actions necessary to provide that the Investor Designees are nominated for re-election to the Board at such annual meeting and the remaining directors shall be nominated in accordance with the provisions of this Agreement and the by-laws of the Company.

(b) Following the 2008 annual meeting of shareholders of the Company: (i) so long as the Investor Percentage Interest equals or exceeds 17.5%, Investor shall

have the right to nominate three directors; (ii) if the Investor Percentage Interest is less than 17.5% but equals or exceeds 7.5% Investor shall have the right to nominate two directors; (iii) if the Investor Percentage Interest is less than 7.5% but the fair market value, as determined by the Board in good faith, of the Voting Securities Beneficially Owned by the Investor Group equals or exceeds \$150 million, Investor shall have the right to nominate one director; and (iv) if the Investor Percentage Interest is less than 7.5% and the fair market value, as determined by the Board in good faith, of the Voting Securities Beneficially Owned by the Investor Shall have the right to nominate any directors. Such nominees shall, subject to Applicable Law, be the Company's nominees to serve on the Board and the Company shall solicit proxies for them to the same extent as it does for any of its other nominees to the Board. Following the 2008 annual meeting of shareholders of the Company, the remaining directors of the Board shall be nominated in accordance with this Agreement and the provisions of the by-laws of the Company.

(c) Subject to Section 1.1(b), the remaining Investor Designees then in office shall have the right to designate any replacement for an Investor Designee upon the death, resignation, retirement, disqualification or removal from office of such director; <u>provided</u>, that if an Investor Designee is removed for cause by the shareholders, the remaining Investor Designee shall not designate the person who was removed as such replacement Investor Designee.

(d) Until the 2008 annual meeting of shareholders of the Company, (<u>i</u>) the remaining MI Designees then in office shall have the right to designate any replacement for a MI Designee upon the death, resignation, retirement, disqualification or removal from office of such director; <u>provided</u>, that if an MI Designee is removed for cause by the shareholders, the remaining MI Designees shall not designate the person who was removed as such replacement MI Designee and (<u>ii</u>) the Initial Unaffiliated Directors by majority vote or consent of those Initial Unaffiliated Directors then in office of such director; <u>provided</u>, that if an Initial Unaffiliated Director upon the death, resignation, retirement, disqualification or removal from office of such director; <u>provided</u>, that if an Initial Unaffiliated Director is removed for cause by the shareholders, the remaining Initial Unaffiliated Directors shall not designate the person who was removed as such replacement Initial Unaffiliated Director.

(e) For purposes of constituting the initial Board as of the Closing Date upon consummation of the Transactions, no Investor Designee shall be deemed not to be an Independent Director because of the ownership of Common Stock by Investor or because of the rights of Investor under this Agreement.

(f) Until the Board shall determine otherwise, the regular meetings of the Board shall be held on the third Thursday of each February, April, June, August, October and December.



# 1.2 Committees.

(a) The Board shall have the following committees: an Audit Committee, a Compensation Committee and a Nominating and Corporate Governance Committee (as such terms are defined in the Company's by-laws). Each of the foregoing committees shall have three members.

(b) All the members of each of the Audit Committee, Compensation Committee and Nominating and Corporate Governance Committee shall qualify as Independent Directors. To the extent permitted by Applicable Law and the rules of the New York Stock Exchange, at least one member of the Compensation Committee (who shall be the Chairman of the Compensation Committee), Nominating and Corporate Governance Committee and the Audit Committee shall be an Investor Designee.

1.3 <u>Articles of Incorporation and By-laws</u>. The Company and Investor shall take or cause to be taken all lawful action necessary to ensure at all times as of and following the Closing Date that the articles of incorporation and by-laws of the Company are not inconsistent with the provisions of this Agreement or the transactions contemplated hereby.

1.4 <u>Approval Rights</u>. In addition to any other approval required, during any time that the restrictions of Section 3.1(a) and Section 3.1(b) are in effect, the Company shall not, and shall cause its subsidiaries not to, take any of the following actions without the approval of the Board by Supermajority Vote:

(i) entering into a merger, reorganization, share exchange, consolidation, business combination, recapitalization, liquidation, dissolution or similar transaction involving the Company that if consummated, would result in a Change of Control; <u>provided</u>, <u>however</u>, that for the purposes of this clause (i) of Section 1.4, the words "a majority of" and "all or substantially all of" in the definition of "Change of Control" shall be replaced by the words "twenty percent of";

(ii) acquiring (including by merger, business combination, reorganization or other similar transaction), in a single transaction or a series of related transactions, any business or assets for consideration having a value (valuing any non-cash consideration at fair market value as determined by the Board in good faith) in excess of \$300 million;

(iii) making or committing to make any capital expenditure or series of related capital expenditures in excess of \$300 million;

(iv) disposing of (including by merger, business combination, reorganization or other similar transaction), in a single transaction or a series of

related transactions, any business or assets for consideration having a value (valuing non-cash consideration at fair market value as determined by the Board in good faith) in excess of \$100 million; and

(v) (A) incurring any indebtedness for borrowed money or issuing any debt securities (other than indebtedness or debt securities owed or issued solely between or among the Company and/or one or more wholly owned Subsidiaries), or (B) guaranteeing any indebtedness for borrowed money of any other Person if the amount of such incurred or guaranteed indebtedness exceeds \$300 million.

1.5 <u>Venture Capital Qualifying Investment</u>. (a) Investor represents and warrants that Investor Fund is a "venture capital operating company" within the meaning of Department of Labor "plan asset" regulations ("<u>VCOC</u>"). Investor agrees to notify the Company promptly if Investor Fund ceases to be a VCOC or if, in Investor's good faith judgment, the provisions set forth in <u>Section 1.5(b)</u> are no longer required in order for the ownership of Common Stock to qualify as a venture capital investment within the meaning of Department of Labor "plan asset" regulations.

(b) The Company hereby agrees that, subject to Applicable Law and existing contractual restrictions and provided that Investor Fund executes a confidentiality agreement in form reasonably satisfactory to the Company covering Investor Fund and its representatives which governs the confidentiality and use of any information received by Investor Fund or its representatives from the Company pursuant to this <u>Section 1.5</u>, it shall (i) furnish Investor Fund with such financial and operating data and other information with respect to the business and properties of the Company as the Company prepares and compiles for its directors in the ordinary course and as Investor Fund may from time to time reasonably request, (ii) permit Investor Fund to discuss the affairs, finances and accounts of the Company, and to make proposals and furnish advice with respect thereto, with the principal officers of the Company within thirty days after the end of each fiscal quarter of the Company, and (iii) invite a representative of Investor Fund to attend all meetings of the Board in a nonvoting observer capacity if none of the Investor Designees is a member of the Board and, in this respect, shall give such representative copies of all notices, minutes, consents and other material that it provides to the directors and such representative shall be entitled to participate in discussions of matters brought to the Board. The provisions of this <u>Section 1.5(b)</u> shall terminate on the earlier of (i) the date of termination of this Article I pursuant to <u>Section 1.5(b)</u> are no longer required in order for the ownership of Common Stock to qualify as a venture capital investment within the meaning of Department of Labor "plan asset" regulations.

1.6 Termination of Article I. Subject to Section 6.1, this Article I (other than Section 1.3) shall terminate and be of no further force or effect on the earlier of (i) the

date on which the Investor Percentage Interest is less than 7.5% and the fair market value, as determined by the Board in good faith, of the Voting Securities Beneficially Owned by the Investor Group is less than \$150 million and (<u>ii</u>) the tenth anniversary of the Closing Date.

### ARTICLE II

### REGISTRATION RIGHTS

#### 2.1 Demand Registrations.

(a) <u>Requests for Registration</u>. At any time following the first anniversary of the Closing Date, Investor may request in writing, on behalf of Investor Group, that the Company effect the registration of all or any part of the Registrable Securities held by Investor Group (a "<u>Registration Request</u>"), <u>provided</u> that, prior to the second anniversary of the Closing Date, the number of shares of Common Stock to be sold by Investor Group pursuant to a Registration Request shall be limited to an amount that will not cause the Investor Percentage Interest to be less than 25%. Promptly after its receipt of any Registration Request, the Company will give written notice of such request to all other Shareholders, and will use its reasonable best efforts to register, in accordance with the provisions of this Agreement, all Registrable Securities that have been requested to be registered in the Registration Request or by any other Shareholders by written notice to the Company given within fifteen Business Days after the date the Company has given such Shareholders notice of the Registration Request. The Company will pay all Registration Expenses incurred in connection with any registration pursuant to this Section 2.1. Any registration requested by Investor pursuant to Section 2.1(a) or 2.1(c) is referred to in this Agreement as a "Demand Registration".

(b) <u>Limitation on Demand Registrations</u>. Investor will be entitled to initiate no more than four Demand Registrations (including Short-Form Registrations permitted pursuant to Section 2.1(c)). No request for registration will count for the purposes of the limitations in this Section 2.1(b) if (<u>i</u>) Investor determines in good faith to withdraw the proposed registration prior to the effectiveness of the Registration Statement relating to such request due to marketing conditions or regulatory reasons relating to the Company, (<u>ii</u>) the Registration Statement relating to such request is not declared effective within 180 days of the date such Registration Request prior to such Registration Statement being declared effective, (<u>iii</u>) prior to the sale of at least 90% of the Registrable Securities included in the applicable registration relating to such request, such registration is adversely affected by any stop order, injunction or other order or requirement of the Commission or other governmental agency or court for any reason and the Company fails to have such stop order, injunction or other order or requirement removed, withdrawn or resolved to Investor's reasonable satisfaction within thirty days of the date of such order, (<u>iv</u>) more

than 10% of the Registrable Securities requested by Investor to be included in the registration are not so included pursuant to Section 2.1(f), or ( $\underline{v}$ ) the conditions to closing specified in the underwriting agreement or purchase agreement entered into in connection with the registration relating to such request are not satisfied (other than as a result of a material default or breach thereunder by Investor). Notwithstanding the foregoing, the Company will pay all Registration Expenses in connection with any request for registration pursuant to Section 2.1(a) regardless of whether or not such request counts toward the limitation set forth above.

(c) <u>Short-Form Registrations</u>. The Company will use its reasonable best efforts to qualify for registration on Form S-3 or any comparable or successor form or forms or any similar short-form registration ("<u>Short-Form Registrations</u>"), and, if requested by Investor and available to the Company, such Short-Form Registration will be a "shelf" registration statement providing for the registration of, and the sale on a continuous or delayed basis of the Registrable Securities, pursuant to Rule 415. In no event shall the Company be obligated to effect any shelf registration other than pursuant to a Short-Form Registration. The Company will pay all Registration Expenses incurred in connection with any Short-Form Registration.

(d) <u>Restrictions on Demand Registrations</u>. If the filing, initial effectiveness or continued use of a registration statement, including a shelf registration statement pursuant to Rule 415, with respect to a Demand Registration would (<u>i</u>) require the Company to make a public disclosure of material non-public information, which disclosure in the good faith judgment of the Board (A) would be required to be made in any Registration Statement so that such Registration Statement would not be materially misleading, (B) would not be required to be made at such time but for the filing, effectiveness or continued use of such Registration Statement and (C) would in the good faith judgment of the Board reasonably be expected to have a material adverse effect on the Company or its business if made at such time, or (<u>ii</u>) would in the good faith and judgment of the Board reasonably be expected to have a material adverse effect on the Company or its business or on the Company sability to effect a planned or proposed acquisition, disposition, financing, reorganization, recapitalization or similar transaction, then the Company may upon giving prompt written notice of such action to the participants in such registration (each of whom hereby agrees to maintain the confidentiality of all information disclosed to such participants) delay the filing or initial effectiveness of, or suspend use of, such Registration Statement, <u>provided</u>, that the Company shall not be permitted to do so (<u>x</u>) more than three times during any twelve-month period or (<u>y</u>) for periods exceeding, in the aggregate, one hundred twenty-five days during any twelve-month period. In the event the Company exercises its rights under the preceding sentence, such Shareholders agree to suspend, promptly upon their receipt of the notice referred to above, their use of any prospectus relating to such registration in connection with any sale or offer to sell Registrable Securities. If the Company so postpones the filing of a prospectus or the effectiveness of a Regis

entitled to withdraw such request and, if such request is withdrawn, such registration request will not count for the purposes of the limitation set forth in Section 2.1(b). The Company will pay all Registration Expenses incurred in connection with any such aborted registration or prospectus.

### (e) Selection of Underwriters.

(i) If Investor intends that the Registrable Securities covered by its Registration Request shall be distributed by means of an underwritten offering, Investor will so advise the Company as a part of the Registration Request, and the Company will include such information in the notice sent by the Company to the other Shareholders with respect to such Registration Request. In such event, the lead underwriter to administer the offering will be chosen by Investor subject to the prior written consent, not to be unreasonably withheld or delayed, of the Company.

(ii) If the offering is underwritten, the right of any Shareholder to registration pursuant to this Section 2.1 will be conditioned upon such Shareholder's participation in such underwriting and the inclusion of such Shareholder's Registrable Securities in the underwriting, and each such Shareholder will (together with the Company and the other Shareholders distributing their securities through such underwriting) enter into an underwriting agreement in customary form with the underwriter or underwriters selected for such underwriting. If any Shareholder disapproves of the terms of the underwriting, such Shareholder may elect to withdraw therefrom by written notice to the Company, the managing underwriter and Investor.

(f) <u>Priority on Demand Registrations</u>. The Company will not include in any underwritten registration pursuant to this Section 2.1 any securities that are not Registrable Securities, without the prior written consent of Investor. If the managing underwriter advises the Company that in its reasonable opinion the number of Registrable Securities (and, if permitted hereunder, other securities requested to be included in such offering) exceeds the number of securities that can be sold in such offering without adversely affecting the marketability of the offering (including an adverse effect on the per share offering price), the Company will include in such offering only such number of securities that in the reasonable opinion of such underwriters can be sold without adversely affecting the marketability of the offering price), which securities will be so included in the following order of priority: (<u>i</u>) first, Registrable Securities of Investor Group and (<u>ii</u>) second, Registrable Securities of any other Shareholders who have delivered written requests for registration pursuant to Section 2.1(a), *pro rata* on the basis of the aggregate number of Registrable Securities owned by each such Shareholder and (<u>iii</u>) any other securities of the Company that have been requested to be so included, subject to the terms of this Agreement.

(g) <u>Effective Registration Statement</u>. A registration requested pursuant to Section 2.1(a) shall not be deemed to have been effected unless it is declared effective by the Commission and remains effective for the period specified in Section 2.3(b).

#### 2.2 Piggyback Registrations.

(a) <u>Right to Piggyback</u>. Whenever the Company proposes to register any of its securities, other than a registration pursuant to Section 2.1 or a Special Registration, and the registration form to be filed may be used for the registration or qualification for distribution of Registrable Securities, the Company will give prompt written notice (and in any event no later than fifteen Business Days prior to the filing of a Registration Statement with respect to such registration) to all Shareholders of its intention to effect such a registration and, subject to Section 2.2(d), will include in such registration all Registrable Securities with respect to which the Company has received written requests for inclusion therein within ten Business Days after the date of the Company's notice (a "<u>Piggyback Registration</u>"). Any Shareholder that has made such a written request may withdraw its Registrable Securities from such Piggyback Registration by giving written notice to the Company and the managing underwriter, if any, on or before the tenth Business Day prior to the planned effective date of such Piggyback Registration. The Company may terminate or withdraw any registration under this Section 2.2 prior to the effectiveness of such registration, whether or not any Shareholder has elected to include Registrable Securities in such registration, and except for the obligation to pay Registration Expenses pursuant to Section 2.2(c) the Company will have no liability to any Shareholder in connection with such termination or withdrawal.

(b) <u>Underwritten Registration</u>. If the registration referred to in Section 2.2(a) is proposed to be underwritten, the Company will so advise the Shareholders as a part of the written notice given pursuant to Section 2.2(a). In such event, the right of any Shareholder to registration pursuant to this Section 2.2 will be conditioned upon such Shareholder's participation in such underwriting and the inclusion of such Shareholder's Registrable Securities in the underwriting, and each such Shareholder will (together with the Company and the other Shareholders and other holders of securities distributing their securities through such underwriting) enter into an underwriting agreement in customary form with the underwriter or underwriters selected for such underwriting by the Company. If any Shareholder disapproves of the terms of the underwriting, such Shareholder may elect to withdraw therefrom by written notice to the Company, the managing underwriter and Investor.

(c) <u>Piggyback Registration Expenses</u>. The Company will pay all Registration Expenses in connection with any Piggyback Registration, whether or not any registration or prospectus becomes effective or final.

(d) Priority on Primary Registrations. If a Piggyback Registration relates to an underwritten primary offering on behalf of the Company, and the managing

underwriters advise the Company that in their reasonable opinion the number of securities requested to be included in such registration exceeds the number which can be sold without adversely affecting the marketability of such offering (including an adverse effect on the per share offering price), the Company will include in such registration or prospectus only such number of securities that in the reasonable opinion of such underwriters can be sold without adversely affecting the marketability of the offering (including an adverse effect on the per share offering price), which securities will be so included in the following order of priority: (i) <u>first</u>, the securities the Company proposes to sell, (ii) <u>second</u>, Registrable Securities of any Shareholders who have requested registration of Registrable Securities pursuant to Sections 2.1 or 2.2, *pro rata* on the basis of the aggregate number of such securities or shares owned by each such Shareholder and (<u>iii) third</u>, any other securities of the Company that have been requested to be so included, subject to the terms of this Agreement.

2.3 <u>Registration Procedures</u>. Subject to Section 2.1(d), whenever the Shareholders of Registrable Securities have requested that any Registrable Securities be registered pursuant to Sections 2.1 or 2.2 of this Agreement, the Company will use its commercially reasonable efforts to effect the registration and sale of such Registrable Securities as soon as reasonably practicable in accordance with the intended method of disposition thereof and pursuant thereto. The Company shall use its reasonable best efforts to as expeditiously as possible:

(a) prepare and file with the Commission a Registration Statement with respect to such Registrable Securities, make all required filings with the National Association of Securities Dealers and thereafter use its reasonable best efforts to cause such Registration Statement to become effective as soon as reasonably practicable, <u>provided</u> that before filing a Registration Statement or any amendments or supplements thereto, the Company will, in the case of a Demand Registration, furnish to Shareholders' Counsel copies of all such documents proposed to be filed, which documents will be subject to review of such counsel at the Company's expense;

(b) prepare and file with the Commission such amendments and supplements to such Registration Statement as may be necessary to keep such Registration Statement effective for a period of either (<u>i</u>) not less than (A) three months, (B) if such Registration Statement relates to an underwritten offering, such longer period as a prospectus is required by law to be delivered in connection with sales of Registrable Securities by an underwriter or dealer or (C) two years in the case of shelf registration statements (or in each case such shorter period ending on the date that the securities covered by such shelf registration statement cease to constitute Registrable Securities) or (<u>ii</u>) such shorter period as will terminate when all of the securities covered by such Registration Statement have been disposed of in accordance with the intended methods of disposition by the seller or sellers thereof set forth in such Registration Statement (but in any event not before the expiration of any longer period required under the Securities Act), and comply

with the provisions of the Securities Act with respect to the disposition of all securities covered by such Registration Statement until such time as all of such securities have been disposed of in accordance with the intended methods of disposition by the seller or sellers thereof set forth in such Registration Statement;

(c) furnish to each seller of Registrable Securities such number of copies, without charge, of such Registration Statement, each amendment and supplement thereto, including each preliminary prospectus, final prospectus, any other prospectus (including any prospectus filed under Rule 424, Rule 430A or Rule 430B under the Securities Act and any "issuer free writing prospectus" as such term is defined under Rule 433 promulgated under the Securities Act), all exhibits and other documents filed therewith and such other documents as such seller may reasonably request including in order to facilitate the disposition of the Registrable Securities owned by such seller;

(d) register or qualify such Registrable Securities under such other securities or blue sky laws of such jurisdictions as any seller reasonably requests and do any and all other acts and things that may be reasonably necessary or reasonably advisable to enable such seller to consummate the disposition in such jurisdictions of the Registrable Securities owned by such seller (<u>provided</u> that the Company will not be required to (<u>i</u>) qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify but for this subsection, (<u>ii</u>) subject itself to taxation in any such jurisdiction or (<u>iii</u>) consent to general service of process in any such jurisdiction);

(e) notify each seller of such Registrable Securities and Shareholders' Counsel, at any time when a prospectus relating thereto is required to be delivered under the Securities Act, upon discovery that, or upon the discovery of the happening of any event as a result of which, the prospectus contains an untrue statement of a material fact or omits any fact necessary to make the statements therein not misleading in the light of the circumstances under which they were made, and, as soon as reasonably practicable, prepare and furnish to such seller a reasonable number of copies of a supplement or amendment to such prospectus so that, as thereafter delivered to the purchasers of such Registrable Securities, such prospectus will not contain an untrue statement of a material fact or omit to state any fact necessary to make the statements therein not misleading in the light of the circumstances under which they were made;

(f) notify each seller of any Registrable Securities covered by such Registration Statement and Shareholders' Counsel (<u>i</u>) when such Registration Statement or the prospectus or any prospectus supplement or post-effective amendment has been filed and, with respect to such Registration Statement or any post-effective amendment, when the same has become effective, (<u>ii</u>) of any request by the Commission for amendments or supplements to such Registration Statement or to amend or to supplement such prospectus or for additional information, and (<u>iii</u>) of the issuance by the Commission of any stop order suspending the effectiveness of such Registration Statement or the initiation of any proceedings for any of such purposes;

(g) cause all such Registrable Securities to be listed on each securities exchange on which similar securities issued by the Company are then listed or, if no similar securities issued by the Company are then listed on any securities exchange, use its reasonable best efforts to cause all such Registrable Securities to be listed on the New York Stock Exchange or the NASDAQ stock market, as determined by the Company;

(h) provide a transfer agent and registrar for all such Registrable Securities not later than the effective date of such Registration Statement;

(i) enter into such customary agreements (including underwriting agreements and, subject to Section 2.7, lock-up agreements in customary form, and including provisions with respect to indemnification and contribution in customary form) and take all such other customary actions as Investor, the selling Shareholders or the underwriters, if any, reasonably request in order to expedite or facilitate the disposition of such Registrable Securities (including, without limitation, making members of senior management of the Company available to participate in "road show" and other customary marketing activities);

(j) make available for inspection by any seller of Registrable Securities and Shareholders' Counsel, any underwriter participating in any disposition pursuant to such Registration Statement and any attorney, accountant or other agent retained by any such seller or underwriter, all financial and other records, pertinent corporate documents and documents relating to the business of the Company, and cause the Company's officers, directors, employees and independent accountants to supply all information reasonably requested by any such seller, underwriter, attorney, accountant or agent in connection with such Registration Statement, <u>provided</u> that it shall be a condition to such inspection and receipt of such information that the inspecting Person (<u>i</u>) enter into a confidentiality agreement in form and substance reasonably satisfactory to the Company and (<u>ii</u>) agree to minimize the disruption to the Company's business in connection with the foregoing;

(k) timely provide to its security holders earning statements satisfying the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder;

(l) in the event of the issuance of any stop order suspending the effectiveness of a Registration Statement, or of any order suspending or preventing the use of any related prospectus or ceasing trading of any securities included in such Registration Statement for sale in any jurisdiction, use every reasonable effort to promptly obtain the withdrawal of such order;

(m) obtain one or more comfort letters, addressed to the underwriters, if any, dated the effective date of such Registration Statement and the date of the closing under the underwriting agreement for such offering, signed by the Company's independent public accountants in customary form and covering such matters of the type customarily covered by comfort letters as such underwriters shall reasonably request; and

(n) provide legal opinions of the Company's counsel, addressed to the underwriters, if any, dated the date of the closing under the underwriting agreement, with respect to the Registration Statement, each amendment and supplement thereto (including the preliminary prospectus) and such other documents relating thereto as the underwriter shall reasonably request in customary form and covering such matters of the type customarily covered by legal opinions of such nature.

As a condition to registering Registrable Securities, the Company may require each Shareholder of Registrable Securities as to which any registration is being effected to furnish the Company with such information regarding such Shareholder and pertinent to the disclosure requirements relating to the registration and the distribution of such securities as the Company may from time to time reasonably request in writing.

#### 2.4 Registration Expenses.

(a) Except as otherwise provided in this Agreement, all expenses incidental to the Company's performance of or compliance with this Agreement, including, without limitation, all registration and filing fees, fees and expenses of compliance with securities or blue sky laws, word processing, duplicating and printing expenses, messenger and delivery expenses, and fees and disbursements of counsel for the Company and all independent certified public accountants, underwriters and other Persons retained by the Company (all such expenses, "<u>Registration Expenses</u>"), will be borne by the Company. The Company will, in any event, pay its internal expenses (including, without limitation, all salaries and expenses of its officers and employees performing legal or accounting duties), the expenses of any annual audit or quarterly review, the expenses of any liability insurance and the expenses and fees for listing the securities to be registered on each securities exchange on which similar securities issued by the Company are then listed or on the New York Stock Exchange or NASDAQ. All Selling Expenses will be borne by the holders of the securities so registered *pro rata* on the basis of the amount of proceeds from the sale of their shares so registered.

(b) In connection with each Demand Registration and each Piggyback Registration in which members of Investor Group participate, the Company will reimburse Investor for the reasonable fees and disbursements of one counsel ("<u>Shareholders' Counsel</u>").

### 2.5 Participation in Underwritten Registrations.

(a) No Shareholder may participate in any registration hereunder that is underwritten unless such Shareholder (<u>i</u>) agrees to sell its Registrable Securities on the basis provided in any underwriting arrangements approved by Investor (including, without limitation, pursuant to the terms of any over-allotment or "green shoe" option requested by the managing underwriter(s), <u>provided</u> that no Shareholder will be required to sell more than the number of Registrable Securities that such Shareholder has

requested the Company to include in any registration), (<u>ii</u>) completes and executes all questionnaires, powers of attorney, indemnities, underwriting agreements and other documents reasonably required under the terms of such underwriting arrangements, and (<u>iii</u>) cooperates with the Company's reasonable requests in connection with such registration or qualification (it being understood that the Company's failure to perform its obligations hereunder, which failure is caused by such Shareholder's failure to cooperate with such reasonable requests, will not constitute a breach by the Company of this Agreement). Notwithstanding the foregoing, no Shareholder will be required to agree to any indemnification obligations on the part of such Shareholder that are materially greater than its obligations pursuant to Section 4.1(b).

(b) Each Shareholder that is participating in any registration hereunder agrees that, upon receipt of any notice from the Company of the happening of any event of the kind described in Section 2.3(f), such Shareholder will forthwith discontinue the disposition of its Registrable Securities pursuant to the Registration Statement until such Shareholder receives copies of a supplemented or amended prospectus as contemplated by such Section 2.3(f). In the event the Company gives any such notice, the applicable time period mentioned in Section 2.3(b) during which a Registration Statement is to remain effective will be extended by the number of days during the period from and including the date of the giving of such notice pursuant to this Section 2.5(b) to and including the date when each seller of a Registrable Security covered by such Registration Statement will have received the copies of the supplemented or amended prospectus contemplated by Section 2.3(f).

### 2.6 Rule 144; Legended Securities; etc.

(a) The Company will use its reasonable best efforts to timely file all reports and other documents required to be filed by it under the Securities Act and the Exchange Act and the rules and regulations adopted by the Commission thereunder (or, if the Company is not required to file such reports, it will, upon the request of any Shareholder, make publicly available such information as necessary to permit sales pursuant to Rule 144), and will take such further action as any Shareholder may reasonably request, all to the extent required from time to time to enable such Shareholder to sell shares of Registrable Securities without registration under the Securities Act within the limitation of the exemptions provided by Rule 144. Upon the request of any Shareholder, the Company will deliver to such Shareholder a written statement as to whether it has complied with such information requirements.

(b) The Company will not issue new certificates for shares of Registrable Securities without a legend restricting further transfer unless (<u>i</u>) such shares have been sold to the public pursuant to an effective Registration Statement under the Securities Act or Rule 144, or (<u>ii</u>) (<u>x</u>) otherwise permitted under the Securities Act, (<u>y</u>) the Shareholder of such shares shall have delivered to the Company an opinion of counsel, which opinion

and counsel shall be reasonably satisfactory to the Company, to such effect, and (<u>z</u>) the Shareholder of such shares expressly requests the issuance of such certificates in writing.

2.7 <u>Holdback</u>. In consideration for the Company agreeing to its obligations under this Agreement, each Shareholder agrees in connection with any registration of the Company's securities (whether or not such Shareholder is participating in such registration) upon the request of the Company and the underwriters managing any underwritten offering of the Company's securities, not to effect (other than pursuant to such registration) any public sale or distribution of Registrable Securities, including, but not limited to, any sale pursuant to Rule 144 or Rule 144A, or make any short sale of, loan, grant any option for the purchase of, or otherwise dispose of any Registrable Securities, any other equity securities of the Company or any securities convertible into or exchangeable or exercisable for any equity securities of the Company without the prior written consent of the Company or such underwriters, as the case may be, during the Holdback Period, <u>provided</u> that nothing herein will prevent any Shareholder that is otherwise in compliance with applicable securities laws, so long as such distributees agree to be so bound. With respect to such underwritten offering of Registrable Securities covered by a registration pursuant to Sections 2.1 or 2.2, the Company further agrees not to effect (other than pursuant to such registration) covering any, of its equity securities, or any securities convertible into or exchangeable or exercisable for such securities, during the Holdback Period with respect to such underwritten offering, if required by the managing underwriter, <u>provided</u> that notwithstanding anything to the contrary herein, the Company's obligations under this Section 2.7 shall not apply during any twelve-month period for more than an aggregate of ninety days.

# ARTICLE III

#### TRANSFERS; STANDSTILL PROVISIONS; PREEMPTIVE RIGHTS

3.1 <u>Investor Group Transfer Restrictions</u>. (a) Prior to the first anniversary of the Closing Date, no member of Investor Group will, directly or indirectly, sell, transfer, make any short sale of, loan, grant any option for the purchase of or otherwise dispose of any shares of Common Stock (it being understood that transfers of, or other transactions with respect to ownership interests in the Investor Fund or ownership interests in other members of the Investor Group the purpose of which is not to transfer shares of Common Stock shall not be considered to be direct or indirect transfers of shares of Common Stock) except (i) to other members of Investor Group who agree in writing to be bound by the terms of this Agreement, (ii) pursuant to the terms of a Buyout Transaction, (iii) in connection with a bona

fide pledge to, or similar arrangement in connection with a bona fide borrowing from, a financial institution, or (<u>iv</u>) in a transaction approved by a majority of the directors of the Company who qualify as Independent Directors who are not Investor Designees.

(b) Following the first anniversary of the Closing Date and prior to the second anniversary of the Closing Date, no member of Investor Group will, directly or indirectly, sell, transfer, make any short sale of, loan, grant any option for the purchase of or otherwise dispose of any shares of Common Stock (it being understood that transfers of, or other transactions with respect to ownership interests in the Investor Fund or ownership interests in other members of the Investor Group the purpose of which is not to transfer shares of Common Stock shall not be considered to be direct or indirect transfers of shares of Common Stock) except (i) to other members of Investor Group who agree in writing to be bound by the terms of this Agreement, (ii) pursuant to the terms of a Buyout Transaction, (iii) if following the closing of such transfer, the Investor Percentage Interest would not be less than 17.5%, (iv) in connection with a bona fide pledge to, or similar arrangement in connection with a bona fide borrowing from, a financial institution or (v) in a transaction approved by a majority of the directors of the Company who qualify as Independent Directors who are not Investor Designees.

(c) Any transfer or attempted transfer of shares of Common Stock in violation of this Section 3.1 shall, to the fullest extent permitted by law, be null and void <u>ab initio</u>, and the Company shall not, and shall instruct its transfer agent and other third parties not to, record or recognize any such purported transaction on the share register of the Company.

(d) Investor acknowledges that this Section 3.1 may be enforced by the Company at the direction of a majority of the Independent Directors who are not Investor Designees.

(e) This Section 3.1 shall terminate and be of no further force or effect on the second anniversary of the Closing Date, <u>provided</u> that such termination shall not relieve any party of liability for such party's breach of this Article III prior to such termination.

3.2 <u>Standstill Provisions</u>. (a) Unless specifically requested in writing in advance by the Company's Board of Directors, Investor will not and will cause each Investor Group member not to (and Investor will not and will cause each Investor Group member not to at any time assist or encourage others to):

(i) acquire or agree, offer, seek or propose to acquire, directly or indirectly, alone or in concert with any other Person, by purchase or otherwise, any (A) ownership of any of the material assets or businesses of the Company or any subsidiary thereof, or any rights or options to acquire such ownership (including from any third party), or (B) ownership, including, but not limited to, beneficial ownership as defined in Rule 13d-3 under the Exchange Act, of any

securities of the Company or any subsidiary thereof, or any rights or options to acquire such ownership (including from any third party), if such ownership would result in an Investor Percentage Interest in excess of 40%;

(ii) solicit proxies (as such terms are defined in Rule 14a-1 under the Exchange Act), whether or not such solicitation is exempt under Rule 14a-2 under the Exchange Act, with respect to any matter from holders of any shares of stock of the Company or any securities convertible into or exchangeable for or exercisable (whether currently or upon the occurrence of any contingency) for the purchase of such stock, or make any communication exempted from the definition of solicitation by Rule 14a-1(l)(2)(iv) under the Exchange Act;

(iii) initiate, or induce or attempt to induce any other Person, entity or group (as defined in Section 13(d)(3) of the Exchange Act) to initiate, any shareholder proposal or tender offer for any securities of the Company or any subsidiary thereof, any change of control of the Company or any subsidiary thereof or the convening of a shareholders' meeting of the Company or any subsidiary thereof;

(iv) enter into any discussions, negotiations, arrangements or understandings with any other Person with respect to any matter described in the foregoing subparagraphs (i) through (iii);

(v) request the Company (or its directors, officers, employees or agents), directly or indirectly, to amend or waive any provision of this Section 3.2(a); or

(vi) take any action with respect to any of the matters described in this Section 3.2(a) that requires public disclosure.

(b) The provisions of Section 3.2(a) shall not apply in respect of any action taken by the Investor Designees in their capacity as members of the Board.

(c) The provisions of Section 3.2(a) shall terminate on earliest of (i) the two year anniversary of the Closing Date, (ii) the date on which any Investor Designee that Investor is entitled to designate pursuant to Section 1.1(b) is not elected to the Board at any annual meeting of the shareholders of the Company (or at any special meeting held to elect directors in lieu of an annual meeting) and is not otherwise appointed to the Board, and (iii) the date of a Change of Control (the "<u>Standstill Termination Date</u>"). In addition, the provisions of Section 3.2(a) shall not apply at any time after (A) the Board resolves to pursue a Buyout Transaction or a transaction that is contemplated by the Board to result in a Change of Control or (B) the Board approves, recommends or accepts a Buyout Transaction or a transaction that would result in a Change of Control proposed by any Person (other than any Investor Group member); provided, however, that the provisions

of Section 3.2(a) shall again become operative at any time that the Board (1) resolves not to pursue any such transaction described in clause (A) above or (2) rejects or announces that it has withdrawn its recommendation of any such transaction described in clause (B) above.

3.3 <u>Anti-Takeover Provisions</u>. From the date hereof until the Standstill Termination Date, the Company shall take all reasonable actions to ensure that (i) to the extent permissible under Applicable Law, no "fair price," "moratorium," "control share acquisition" or other form of antitakeover statute or regulation under Wisconsin law, (ii) no anti-takeover provision in the articles of incorporation or by-laws of the Company or other similar organizational documents of its subsidiaries, and (iii) no shareholder rights plan, "poison pill" or similar measure, in each case that contains restrictions that are different from or in addition to those contained in Sections 3.1 and 3.2 (including with respect to the time periods specified in Section 3.1), is applicable to Investor's ownership of Common Stock.

3.4 <u>Buyout Transactions</u>. So long as Investor is in compliance with Section 3.1, nothing set forth in Section 3.1 or Section 3.2 shall prohibit Investor from (i) selling or transferring shares of Common Stock pursuant to the terms of a Buyout Transaction, (ii) voting its shares of Common Stock with respect to any Buyout Transaction or (iii) endorsing a Buyout Transaction or any other transaction that would constitute a Change of Control proposed by any Person (other than any member of the Investor Group or any Controlled Affiliate of a member of the Investor Group); provided that, in the case of clause (iii) above, (A) no member of the Investor Group or any Controlled Affiliate of a member of the Investor Group or any such transaction that constitutes a Change of Control, (B) no member of the Investor Group or any Controlled Affiliate of a member of any Controlled Affiliate of a member of the Investor Group solicits or induces such Person to propose such a transaction and (C) no member of the Investor Group or any Controlled Affiliate of a member of the Investor Group is providing equity or debt financing in connection with such transaction.

#### 3.5 Preemptive Rights.

(a) <u>Sale of New Stock</u>. Until the date on which the Investor's Investor Percentage Interest is less than 10%, if the Company at any time or from time to time makes a Qualified Equity Offering, Investor shall be afforded the opportunity to acquire from the Company for the same price and on the same terms as such securities are proposed to be offered to others, in the aggregate up to the amount of New Stock required to enable it to maintain its Investor Percentage Interest.

# (b) Notice.

(i) In the event the Company intends to make a Qualified Equity Offering that is an underwritten public offering or a private offering made to

financial institutions for resale pursuant to Rule 144A, no later than five business days after the initial filing of a registration statement with the Commission with respect to such underwritten public offering or the commencement of marketing with respect to such Rule 144A offering, it shall give Investor written notice of its intention (including, in the case of a registered public offering and to the extent possible, a copy of the prospectus included in the registration statement filed in respect of such offering) describing, to the extent then known, the anticipated amount of securities, range of prices, timing and other material terms of such offering. Investor shall have five business days from the date of receipt of any such notice to notify the Company in writing that it intends to exercise such preemptive purchase rights and as to the amount of New Stock Investor desires to purchase, up to the maximum amount calculated pursuant to Section 3.5(a) (the "<u>Designated Stock</u>"). Such notice shall constitute a non-binding indication of interest of Investor to purchase the Designated Stock so specified at the range of prices and other terms set forth in the Company's notice to it. The failure to respond during such five Business Day period shall constitute a waiver of the preemptive rights in respect of such offering.

(ii) If the Company proposes to make a Qualified Equity Offering that is not an underwritten public offering or Rule 144A offering (a "<u>Private</u> <u>Placement</u>"), the Company shall give Investor written notice of its intention, describing, to the extent then known, the anticipated amount of securities, price and other material terms upon which the Company proposes to offer the same. Investor shall have five Business Days from the date of receipt of the notice required by the immediately preceding sentence to notify the Company in writing that it intends to exercise such preemptive purchase rights and as to the amount of Designated Stock Investor desires to purchase, up to the maximum amount calculated pursuant to Section 3.5(a). Such notice shall constitute the binding agreement of Investor to purchase the amount of Designated Stock so specified (or a proportionately lesser amount if the amount of New Stock to be offered in such Private Placement is subsequently reduced) upon the price and other terms set forth in the Company's notice to it. The failure of Investor to respond during the five Business Day period referred to in the second preceding sentence shall constitute a waiver of the preemptive rights in respect of such offering.

#### (c) Purchase Mechanism.

(i) If Investor exercises its preemptive purchase rights provided in Section 3.5(b)(ii), the closing of the purchase of the New Stock with respect to which such right has been exercised shall be conditioned on the consummation of the Private Placement giving rise to such preemptive purchase rights and shall take place simultaneously with the closing of the Private Placement or on such other date as the Company and the Investor shall agree in writing; <u>provided</u>, that



the actual amount of Designated Stock to be sold to the Investor pursuant to its exercise of preemptive rights hereunder shall be reduced if the aggregate amount of New Stock sold in the Private Placement is reduced and, at the option of the Investor (to be exercised by delivery of written notice to the Company within three Business Days of receipt of notice of such increase), shall be increased if such aggregate amount of New Stock sold in the Private Placement is increased. In connection with its purchase of Designated Stock, Investor shall execute an instrument in form and substance reasonably satisfactory to the Company containing representations, warranties and agreements of Investor that are customary for private placement transactions.

(ii) If the Investor exercises its preemptive purchase rights provided in Section 3.5(b)(i), the Company shall offer the Investor, if such underwritten public offering or Rule 144A offering is consummated, the Designated Stock (as adjusted to reflect the actual size of such offering when priced) at the same price as the New Stock is offered to the underwriters or initial purchasers and shall provide written notice of such price to Investor as soon as practicable prior to such consummation. Contemporaneously with the execution of any underwriting agreement or purchase agreement entered into between the Company and the underwriters or initial purchasers of such underwritten public offering or Rule 144A offering, Investor shall enter into an instrument in form and substance reasonably satisfactory to the Company acknowledging Investor's binding obligation to purchase the Designated Stock to be acquired by it and containing representations, warranties and agreements of Investor that are customary in private placement transactions, and the failure to enter into such an instrument at or prior to such time shall constitute a waiver of the preemptive rights in respect of such offering. Any offers and sales pursuant to this Section 3.5 in the context of a registered public offering shall be also conditioned on reasonably acceptable representations and warranties of the Investor regarding its status as the type of offeree to whom a private sale can be made concurrently with a registered public offering in compliance with applicable securities laws.

(d) <u>Failure of Purchase</u>. In the event the Investor fails to exercise its preemptive purchase rights provided in this Section 3.5 within the applicable five Business Day period or, if so exercised, the Investor does not consummate such purchase within the applicable period, the Company shall thereafter be entitled during the period of 120 days following the conclusion of the applicable period to sell or enter into an agreement (pursuant to which the sale of New Stock covered thereby shall be consummated, if at all, within 60 days from the date of such agreement) to sell the New Stock not purchased pursuant to this Section 3.5 at a price which is at a discount (expressed as a percentage) to the market price of the shares of the Company that does not exceed by more than 5% the discount (expressed as a percentage) to the market price offered in the Qualified Equity Offering giving rise to such preemptive purchase rights

hereunder (if such a discount was so offered). In the event the Company has not sold the New Stock or entered into an agreement to sell the New Stock within said 120 day period, the Company shall not thereafter offer, issue or sell such New Stock without first offering such securities to Investor in the manner provided in this Section 3.5.

(e) The Investor shall not have any rights to participate in the negotiation of the proposed terms of any Private Placement, underwritten public offering or Rule 144A offering.

(f) The Company and the Investor shall cooperate in good faith to facilitate the exercise of the Investor's preemptive rights hereunder, including securing any required approvals or consents, in a manner that does not jeopardize the timing, marketing, pricing or execution of any offering of the Company's securities.

# ARTICLE IV

#### INDEMNIFICATION

#### 4.1 Indemnification.

(a) The Company agrees to indemnify and hold harmless each Shareholder, its officers, directors and managers and each Person who is a controlling Person of such Shareholder within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act (each such person being referred to herein as a "<u>Covered Person</u>") against, and pay and reimburse such Covered Persons for, any losses, claims, damages, liabilities, joint or several, to which such Covered Person may become subject under the Securities Act or otherwise, insofar as such losses, claims, damages or liabilities (or actions or proceedings, whether commenced or threatened, in respect thereof) arise out of or are based upon (<u>i</u>) any untrue or alleged untrue statement of material fact contained or incorporated by reference in any Registration Statement, prospectus or preliminary prospectus or any amendment thereof or supplement thereto, or any document incorporated by reference therein, or (<u>ii</u>) any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, and the Company will pay and reimburse such Covered Persons for any legal or any other expenses actually and reasonably incurred by them in connection with investigating, defending or settling any such loss, claim, liability, action or proceeding, <u>provided</u> that the Company shall not be liable to a Covered Person in any such case to the extent that any such loss, claim, damage, liability (or action or proceeding in respect thereof) or expense arises out of or is based upon an untrue statement or alleged untrue statement, or omission or alleged omission, made or incorporated by reference therein, in reliance upon, and in conformity with, written information prepared and furnished to the Company by such Covered Person expressly for use therein or arises out of or is based on

such Shareholder's failure to deliver a copy of the Registration Statement or prospectus or any amendments or supplements thereto after the Company has furnished such Shareholder with a sufficient number of copies thereof. In connection with an underwritten offering, the Company, if requested, will indemnify such underwriters, their officers and directors and each Person who controls such underwriters (within the meaning of the Securities Act) to the same extent as provided above with respect to the indemnification of the Covered Persons.

(b) In connection with any Registration Statement in which a Shareholder is participating, each such Shareholder will furnish to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such Registration Statement or prospectus and, will indemnify and hold harmless the Company, its directors and officers, each underwriter and any Person who is or might be deemed to be a controlling person of the Company, any of its subsidiaries or any underwriter within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act against any losses, claims, damages, liabilities, joint or several, to which the Company or any such director or officer, any such underwriter or controlling person may become subject under the Securities Act or otherwise, insofar as such losses, claims, damages or liabilities (or actions or proceedings, whether commenced or threatened, in respect thereof) arise out of or are based upon (i) any untrue or alleged untrue statement of material fact contained in the Registration Statement, prospectus or preliminary prospectus or any amendment thereof or supplement thereto or (ii) any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement or omission is made in such Registration Statement, any such prospectus or preliminary prospectus or any amendment or supplement thereto in reliance upon and in conformity with written information prepared and furnished to the Company by such Shareholder expressly for use therein, and such Shareholder will reimburse the Company and each such director, officer, underwriter and controlling Person for any legal or any other expenses actually and reasonably incurred by them in connection with investigating, defending or settling any such loss, claim, liability, action or proceeding, <u>provided</u> that the obligation to indemnify and hold harmless will be individual

(c) Any Person entitled to indemnification hereunder will (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party will not, without the indemnified party's prior consent,

settle or compromise any action or claim or consent to the entry of any judgment unless such settlement or compromise includes as an unconditional term thereof the release of the indemnified party from all liability, which release shall be reasonably satisfactory to the indemnified party. An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim will not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim.

(d) The indemnification provided for under this Agreement will remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling Person of such indemnified party and will survive the registration and sale of any securities by any Person entitled to any indemnification hereunder and the expiration or termination of this Agreement.

(e) If the indemnification provided for in Section 4.1(a) or Section 4.1(b) is held by a court of competent jurisdiction to be unavailable to an indemnified party with respect to any loss, liability, claim, damage or expense referred to therein, then the indemnifying party, in lieu of indemnifying such indemnified party thereunder, will contribute to the amount paid or payable by such indemnified party as a result of such loss, liability, claim, damage or expense in such proportion as is appropriate to reflect the relative fault of the indemnifying party on the one hand and of the indemnified party on the other hand in connection with the statements or omissions which resulted in such loss, liability, claim, damage or expense as well as any other relevant equitable considerations. The relevant fault of the indemnifying party and the indemnified party will be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission to state a material fact relates to information supplied by the indemnifying party or by the indemnified party and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission. Notwithstanding the foregoing, the amount any Shareholder will be obligated to contribute pursuant to this Section 4.1(e) will be limited to an amount equal to the net proceeds to such Shareholder of the Registrable Securities sold pursuant to the Registration Statement which gives rise to such obligation to contribute (less the aggregate amount of any damages which the Shareholder has otherwise been required to pay in respect of such loss, claim, damage, liability or action or any substantially similar loss, claim, damage, liability or action arising from the sale of such Registrable Securities). No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any person who was not guilty of such fraudulent

#### ARTICLE V

#### DEFINITIONS

5.1 <u>Defined Terms</u>. Capitalized terms when used in this Agreement have the following meanings:

"<u>Acquiring Person</u>" has the meaning set forth in the definition of Change of Control; <u>provided</u>, <u>however</u>, that for purposes of Section 3.4, an Acquiring Person shall not include any Investor solely by reason of Investor's taking or agreeing to take any action permitted under Section 3.4.

"<u>Affiliate</u>" means, with respect to any Person, (i) any Person directly or indirectly Controlling, Controlled by or under common Control with such Person or (<u>ii</u>) any officer, director, manager, general partner or trustee of any of the foregoing; <u>provided</u>, <u>however</u>, that for purposes of this Agreement the Company and any Person directly or indirectly Controlled by the Company shall not be deemed to be Affiliates of Investor or of the Investor Group.

"Agreement" has the meaning set forth in the preamble.

"<u>Applicable Law</u>" means all applicable provisions of (<u>i</u>) constitutions, treaties, statutes, laws (including the common law), rules, regulations, ordinances, codes or orders of any Governmental Entity, (<u>ii</u>) any consents or approvals of any Governmental Entity, and (<u>iii</u>) any orders, decisions, injunctions, judgments, awards, decrees of or agreements with any Governmental Entity.

"Beneficially Own" with respect to any securities shall mean having "beneficial ownership" of such securities (as determined pursuant to Rule 13d-3 under the Exchange Act), including pursuant to any agreement, arrangement or understanding, whether or not in writing.

"Board" has the meaning set forth in Section 1.1(a).

"Business Day" means any day on which banks are not required or authorized to close in the City of New York.

"<u>Buyout Transaction</u>" means a tender offer, merger, sale of all or substantially all the Company's assets or any similar transaction, except such a transaction that is proposed by or involves a member of the Investor Group or an Affiliate of any member of the Investor Group and has not been approved by the Board, that offers each holder of Voting Securities (other than, if applicable, the Person proposing such transaction) the opportunity to dispose of Voting Securities Beneficially Owned by each such holder for the same consideration or otherwise

contemplates the acquisition of Voting Securities Beneficially Owned by each such holder for the same consideration.

"<u>Change of Control</u>" means the consummation of any transaction or series of related transactions involving (i) any purchase or acquisition (whether by way of merger, share exchange, consolidation, business combination or similar transaction or otherwise) by any Person or group (within the meaning of Section 13(d)(3) of the Exchange Act) (such other Person or group, an "<u>Acquiring Person</u>"), of any of (A) securities representing a majority of the outstanding voting power of the Company entitled to elect the Board, (B) the majority of the outstanding shares of common stock of the Company, or (C) all or substantially all of the assets of the Company and its Subsidiaries, taken together as a whole, (ii) any sale, lease, exchange, transfer, license or disposition of all or substantially all of the assets of the Company and its Subsidiaries, taken together as a whole, to an Acquiring Person or (<u>iii</u>) any merger, consolidation or business combination in which the holders of voting securities of the Company immediately prior to the transaction, as a group, do not hold securities representing a majority of the outstanding voting power entitled to elect the board of directors of surviving entity in such merger, consolidation or business combination.

"Class A Common Stock" has the meaning set forth in the recitals.

"Closing Date" has the meaning set forth in the Investment Agreement.

"Commission" means the Securities and Exchange Commission or any other federal agency administering the Securities Act.

"<u>Common Stock</u>" means the common stock, par value \$0.01 per share, of the Company (<u>i</u>) into which the Class A Common Stock held by the Investor shall automatically convert pursuant to its terms and (<u>ii</u>) purchased by Investor pursuant to the exercise of the Purchase Rights and any securities issued in respect thereof, or in substitution therefor, in connection with any stock split, dividend or combination, or any reclassification, recapitalization, merger, consolidation, exchange or other similar reorganization.

"Company" has the meaning set forth in the preamble.

"Control" means the power to direct the affairs of a Person by reason of ownership of Voting Securities, by contract or otherwise.

"Covered Person" has the meaning set forth in Section 4.1(a).

"Demand Registration" has the meaning set forth in Section 2.1(a).

"Designated Stock" has the meaning set forth in Section 3.5(b)(i).

"Exchange Act" means the Securities Exchange Act of 1934, as amended, or any similar federal statute and the rules and regulations thereunder, as in effect from time to time.

"Governmental Entity" means any federal, state, local or foreign court, legislative, executive or regulatory authority or agency.

"<u>Holdback Period</u>" means, with respect to any registered offering covered by this Agreement, (<u>i</u>) ninety days after and during the ten days before, the effective date of the related Registration Statement or, in the case of a takedown from a shelf registration statement, ninety days after the date of the prospectus supplement filed with the Commission in connection with such takedown and during such prior period (not to exceed ten days) as the Company has given reasonable written notice to the holder of Registrable Securities or (<u>ii</u>) such shorter period as Investor, the Company and the underwriter of such offering, if any, shall agree.

"Independent Director" means an individual who, as a member of the Board following the Closing Date, would be independent of the Company under the rules of the New York Stock Exchange, Inc. or such other securities exchange on which the Common Stock is listed.

"Initial Unaffiliated Director" has the meaning set forth in Section 1.1(a).

"Investment" has the meaning set forth in the recitals.

"Investment Agreement" has the meaning set forth in the recitals.

"Investor" has the meaning set forth in the preamble.

"Investor Affiliate" means an Affiliate of Investor other than any "portfolio company" (as such term is customarily used among institutional investors) of Investor or any Affiliate of Investor.

"Investor Cessation Date" has the meaning set forth in Section 6.1.

"Investor Designees" has the meaning set forth in Section 1.1(a).

"Investor Fund" shall mean Warburg Pincus Private Equity IX, L.P., a Delaware limited partnership, or any Successor Fund that Beneficially Owns Common Stock.

"Investor Group" means Investor, Investor Fund and any Investor Affiliate.

"<u>Investor Percentage Interest</u>" means the percentage of Total Voting Power, determined on the basis of the number of Voting Securities actually outstanding, that is controlled directly or indirectly by Investor Group, including as Beneficially Owned.

"<u>Investor Permitted Transferee</u>" means each of (<u>i</u>) Investor Fund, (<u>ii</u>) an Investor Affiliate, (<u>iii</u>) the owners of Investor, including Beneficial Owners of any owners of Investor, in connection with any liquidation of, or a distribution with respect to equity interests owned in, Investor (including but not limited to any distributions by the owners of Investor to their Beneficial Owner) or (<u>iv</u>) any financial institution that acquires shares of Common Stock pursuant to Section 3.1(a)(iii).

"<u>MI Corp.</u>" has the meaning set forth in the recitals.

"MI Designees" has the meaning set forth in Section 1.1(a).

"<u>New Stock</u>" means common stock of the Company or securities convertible into or exchangeable for common stock of the Company offered in a public or nonpublic offering by the Company.

"<u>Person</u>" means an individual, a partnership, a joint venture, a corporation, a limited liability company, a trust, an unincorporated organization or a government or department or agency thereof.

"Piggyback Registration" has the meaning set forth in Section 2.2(a).

"Private Placement" has the meaning set forth in Section 3.5(b)(ii).

"Public Offering" means an offering of Common Stock pursuant to a Registration Statement filed in accordance with the Securities Act.

"Purchase Rights" has the meaning set forth in the recitals.

"Qualified Equity Offering" means a public or nonpublic offering of common stock of the Company or securities convertible into or exchangeable for common stock of the Company (collectively, "New Stock") solely for cash and not pursuant to a Special Registration; <u>provided</u>, <u>however</u>, that none of the following offerings shall constitute a Qualified Equity Offering: (i) any offering pursuant to any stock purchase plan, stock ownership plan, stock option plan or other similar plan where stock is being issued or offered to a trust, other entity or otherwise, to or for the benefit of any employees, officers, consultants, directors, customers, lenders or vendors of the Company, or (<u>ii</u>) any offering made as part of or in connection with a merger or acquisition, a partnership or joint venture or strategic

alliance or investment by the Company or a similar non-capital-raising transaction.

"<u>Register</u>," "<u>registered</u>" and "<u>registration</u>" refers to a registration effected by preparing and filing a Registration Statement in compliance with the Securities Act, and the declaration or ordering of the effectiveness of such Registration Statement, and compliance with applicable state securities laws of such states in which Shareholders notify the Company of their intention to offer Registrable Securities.

"<u>Registrable Securities</u>" means (i) all Common Stock, (ii) any other stock or securities that the Shareholders of the Common Stock may be entitled to receive, or will have received pursuant to such Shareholders' ownership of the Common Stock, in lieu of or in addition to Common Stock, or (iii) any equity securities issued or issuable directly or indirectly with respect to the securities referred to in the foregoing clauses (i) or (ii) by way of conversion or exchange thereof or share dividend or share split or in connection with a combination of shares, recapitalization, reclassification, merger, amalgamation, arrangement, consolidation or other reorganization. As to any particular securities constituting Registrable Securities, such securities will cease to be Registrable Securities when ( $\underline{w}$ ) they have been effectively registered or qualified for sale by prospectus filed under the Securities Act and disposed of in accordance with the Registration Statement covering therein, ( $\underline{x}$ ) they have been sold to the public pursuant to Rule 144 or Rule 145 or other exemption from registration under the Securities Act or ( $\underline{y}$ ) they have been acquired by the Company.

"Registration Expenses" has the meaning set forth in Section 2.4(a).

"<u>Registration Request</u>" has the meaning set forth in Section 2.1(a). The term Registration Request will also include, where appropriate, a Short-Form Registration request made pursuant to Section 2.1(c).

"Registration Statement" means the prospectus and other documents filed with the Commission to effect a registration under the Securities Act.

"<u>Rule 144</u>" means Rule 144 under the Securities Act or any successor or similar rule as may be enacted by the Commission from time to time, as in effect from time to time.

"<u>Rule 144A</u>" means Rule 144A under the Securities Act or any successor or similar rule as may be enacted by the Commission from time to time, as in effect from time to time.



"<u>Rule 145</u>" means Rule 145 under the Securities Act or any successor or similar rule as may be enacted by the Commission from time to time, as in effect from time to time.

"<u>Rule 415</u>" means Rule 415 under the Securities Act or any successor or similar rule as may be enacted by the Commission from time to time, as in effect from time to time.

"Securities Act" means the Securities Act of 1933, as amended, or any similar federal statute and the rules and regulations thereunder, as in effect from time to time.

"Selling Expenses" means all underwriting discounts, selling commissions and transfer taxes applicable to the sale of Registrable Securities hereunder and any other Registration Expenses required by law to be paid by a selling Shareholder.

"Shareholder" means any Investor Permitted Transferee who holds outstanding Registrable Securities and is or becomes a party to this Agreement.

"Shareholders' Counsel" has the meaning set forth in Section 2.4(b).

"Short-Form Registrations" has the meaning set forth in Section 2.1(c).

"Special Registration" means the registration of (i) equity securities and/or options or other rights in respect thereof solely registered on Form S-4 or Form S-8 (or successor form) or (ii) shares of equity securities and/or options or other rights in respect thereof to be offered to directors, members of management, employees, consultants, customers, lenders or vendors of the Company or its direct or indirect subsidiaries or in connection with dividend reinvestment plans.

"Standstill Termination Date" has the meaning set forth in Section 3.2(c).

"Successor Fund" means one or more successor funds to the Investor Fund, each of which is Controlled by Warburg Pincus LLC and/or Warburg Pincus & Co. (or a Controlled Affiliate of one of such entities) and is managed by Warburg Pincus LLC or its Affiliates.

"Supermajority Vote" means the affirmative vote of at least eight members of the Board.

"Total Voting Power" at any time shall mean the total combined voting power in the general election of directors of all the Voting Securities then outstanding.

"Transactions" has the meaning set forth in the Investment Agreement.

"<u>Voting Securities</u>" means, at any time, shares of any class of equity securities of the Company, which are then entitled to vote generally in the election of directors.

5.2 <u>Terms Generally</u>. The words "hereby", "herein", "hereof", "hereunder" and words of similar import refer to this Agreement as a whole and not merely to the specific section, paragraph or clause in which such word appears. All references herein to Articles and Sections shall be deemed references to Articles and Sections of this Agreement unless the context shall otherwise require. The words "include", "includes" and "including" shall be deemed to be followed by the phrase "without limitation". The definitions given for terms in this Article V and elsewhere in this Agreement shall apply equally to both the singular and plural forms of the terms defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. References herein to any agreement or letter (including the Investment Agreement) shall be deemed references to such agreement or letter as it may be amended, restated or otherwise revised from time to time.

# ARTICLE VI

#### MISCELLANEOUS

6.1 <u>Term</u>. This Agreement will be effective as of the date hereof and will continue in effect thereafter until the earliest of (<u>a</u>) its termination by the consent of all parties hereto or their respective successors in interest (with the consent of a majority of Independent Directors who are not Investor Designees), (<u>b</u>) except for those provisions of this Agreement that terminate as of a date specified in such provisions, which provisions shall terminate in accordance with the terms thereof, the date on which Investor Group ceases to hold any shares of Registrable Securities ("<u>Investor Cessation Date</u>") and (<u>c</u>) the dissolution, liquidation or winding up of the Company.

6.2 <u>No Inconsistent Agreements</u>. The Company will not hereafter enter into any agreement with respect to its securities which is inconsistent with or violates the rights granted to the holders of Registrable Securities in this Agreement or grant any registration rights to any other Person without obtaining the prior approval of Investor.

## 6.3 <u>Legend</u>.

(a) All certificates representing the shares of Common Stock held by each Shareholder shall bear a legend substantially in the following form:

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO A SHAREHOLDERS AGREEMENT (A COPY OF WHICH IS ON FILE WITH THE SECRETARY OF THE COMPANY). NO TRANSFER, SALE, ASSIGNMENT, PLEDGE, HYPOTHECATION OR OTHER DISPOSITION OF THE SECURITIES

# REPRESENTED BY THIS CERTIFICATE MAY BE MADE EXCEPT IN ACCORDANCE WITH THE PROVISIONS OF SUCH SHAREHOLDERS AGREEMENT AND (<u>A</u>) PURSUANT TO A REGISTRATION STATEMENT EFFECTIVE UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR (<u>B</u>) PURSUANT TO AN EXEMPTION FROM REGISTRATION THEREUNDER. THE HOLDER OF THIS CERTIFICATE, BY ACCEPTANCE OF THIS CERTIFICATE, AGREES TO BE BOUND BY ALL OF THE PROVISIONS OF SUCH SHAREHOLDERS AGREEMENT."

(b) Upon the permitted sale of any shares of Common Stock pursuant to (<u>i</u>) an effective Registration Statement under the Securities Act or pursuant to Rule 144 or (<u>ii</u>) another exemption from registration under the Securities Act or upon the termination of this Agreement, the certificates representing such shares of Common Stock shall be replaced, at the expense of the Company, with certificates or instruments not bearing the legends required by this Section 6.3 <u>provided</u> that the Company may condition such replacement of certificates under the foregoing clause (ii) upon the receipt of an opinion of securities counsel reasonably satisfactory to the Company.

6.4 <u>Amendments and Waivers</u>. Except as otherwise provided herein, the provisions of this Agreement may be amended or waived only upon the prior written consent of the Company (to the extent approved by a majority of Independent Directors who are not Investor Designees) and Investor. A copy of each such amendment shall be sent to each Shareholder and shall be binding upon each party hereto, <u>provided</u> that the failure to deliver a copy of such amendment shall not impair or affect the validity of such amendment.

6.5 <u>Successors and Assigns</u>. This Agreement will be binding upon and inure to the benefit of and be enforceable by the Company and its successors and permitted assigns and will be binding upon Investor and its successors and permitted assigns. This Agreement will inure to the benefit of and be enforceable by Investor and solely with respect to Article II and Article IV, any Shareholder who is a permitted assignee hereunder. Notwithstanding the foregoing, no member of Investor Group may assign its rights under this Agreement without the prior written consent of the Company, provided that, subject to Section 3.1, Investor may assign its rights under Article IV, absent such consent, in connection with a sale, transfer or disposition to any Investor Permitted Transferee who is a Shareholder. Notwithstanding anything to the contrary in this Agreement, the Company may assign this Agreement in connection with a merger, reorganization or sale, transfer or contribution of all or substantially all of the assets or shares of the Company to any Person; provided, that such Person expressly or by operation of law or otherwise assumes the due and punctual performance and observance of every covenant, agreement and condition of this Agreement to be performed and observed by the Company.

6.6 <u>Severability</u>. Whenever possible, each provision of this Agreement will be interpreted in such manner as to be effective and valid under Applicable Law, but if any provision of this Agreement is held to be invalid, illegal or unenforceable in any respect under any Applicable Law or rule in any jurisdiction, such invalidity, illegality or unenforceability will not affect any other provision or the effectiveness or validity of any provision in any other jurisdiction, and this Agreement will be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision had never been contained herein.

6.7 <u>Counterparts</u>. This Agreement may be executed simultaneously in two or more counterparts, any one of which need not contain the signatures of more than one party, but all such counterparts taken together will constitute one and the same Agreement.

6.8 Descriptive Headings. The descriptive headings of this Agreement are inserted for convenience only and do not constitute a part of this Agreement.

6.9 <u>Governing Law</u>. This Agreement will be governed by and construed in accordance with the laws of the State of Wisconsin regardless of the laws that might otherwise govern under applicable principles or rules of conflicts of law to the extent such principles or rules are not mandatorily applicable by statute and would require the application of the laws of another jurisdiction.

6.10 <u>Consent to Jurisdiction</u>. Each party irrevocably submits to the exclusive jurisdiction of any federal or state court located in the State of Wisconsin, for the purposes of any suit, action or other proceeding arising out of this Agreement or any transaction contemplated hereby (and agrees not to commence any such suit, action or other proceeding except in such courts). Each party further agrees that service of any process, summons, notice or document by U.S. registered mail to such party's respective address set forth or referred to in Section 6.14 shall be effective service of process for any such suit, action or other proceeding in the above-named courts, or that any such suit, action or other proceeding brought in any such court has been brought in an inconvenient forum.

6.11 <u>Waiver of Jury Trial</u>. Each party hereby waives, to the fullest extent permitted by Applicable Law, any right it may have to a trial by jury in respect of any suit, action or other proceeding arising out of this Agreement or any transaction contemplated hereby. Each party (<u>a</u>) certifies and acknowledges that no representative, agent or attorney of any other party has represented, expressly or otherwise, that such other party would not, in the event of litigation, seek to enforce the foregoing waiver and (<u>b</u>) acknowledges that it understands and has considered the implications of this waiver and makes this waiver voluntarily, and that it and the other parties have been induced to

enter into the Agreement by, among other things, the mutual waivers and certifications in this Section 6.11.

6.12 Enforcement; Attorneys' Fees. Each party hereto acknowledges that money damages would not be an adequate remedy in the event that any of the covenants or agreements in this Agreement are not performed in accordance with its terms, and it is therefore agreed that in addition to and without limiting any other remedy or right it may have, the non-breaching party will have the right to an injunction, temporary restraining order or other equitable relief in any court of competent jurisdiction enjoining any such breach and enforcing specifically the terms and provisions hereof, provided that no Shareholder will have any right to an injunction to prevent the filing or effectiveness of any Registration Statement of the Company. In any action or proceeding brought to enforce any provision of this Agreement, the successful party shall be entitled to recover reasonable attorneys' fees in addition to its costs and expenses and other available remedies.

6.13 <u>No Third Party Beneficiaries</u>. Nothing in this Agreement shall confer any rights upon any Person other than the parties hereto and each such party's respective heirs, successors and permitted assigns, all of whom shall be third party beneficiaries of this Agreement, <u>provided</u> that the Persons indemnified under Article IV are intended third party beneficiaries of Article IV.

6.14 <u>Notices</u>. All notices, requests, demands, waivers and other communications required or permitted to be given under this Agreement shall be in writing and shall be deemed to have been duly given if (<u>a</u>) delivered personally, (<u>b</u>) mailed, certified or registered mail with postage prepaid, (<u>c</u>) sent by reputable overnight courier or (<u>d</u>) sent by fax (<u>provided</u> a confirmation copy is sent by one of the other methods set forth above), as follows (or to such other address as the party entitled to notice shall hereafter designate in accordance with the terms hereof):

If to the Company, to it at:

Metavante Corporation 4900 West Brown Deer Road Milwaukee, WI 53223-2459 Attention: Frank Martire President and Chief Executive Officer Facsimile: 414-362-1705 with a copy to (which shall not constitute notice):

Metavante Corporation 4900 West Brown Deer Road Milwaukee, WI 53223-2459 Attention: Norrie Daroga Executive Vice President, Chief Risk Officer and Secretary Facsimile: 414-362-1705

If to Investor, to it at:

WPM, L.P. c/o Warburg Pincus Private Equity IX, L.P. 466 Lexington Avenue New York, New York 10017 Attention: James Neary Facsimile: 212-878-9351

with a copy to (which shall not constitute notice):

Wachtell, Lipton, Rosen & Katz 51 West 52nd Street New York, New York 10019 Attention: Andrew R. Brownstein Igor Kirman Facsimile: (212) 403-2000

If to any other Shareholder, to its address set forth on the signature page of such Shareholder to this Agreement with a copy (which shall not constitute notice) to any party so indicated thereon.

All notices and other communications hereunder shall be in writing and shall be deemed duly given ( $\underline{w}$ ) on the date of delivery if by personal delivery, ( $\underline{x}$ ) upon confirmation of receipt if delivered by facsimile, (y) on the first Business Day following the date of dispatch if delivered by a recognized next-day courier service (z) when received if delivered by certified or registered mail, return receipt requested, postage prepaid.

6.15 <u>Entire Agreement</u>. This Agreement, constitutes the entire agreement and supersedes all prior agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof.

[the remainder of this page left intentionally blank]

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement by their authorized representatives as of the date first above written.

# METAVANTE TECHNOLOGIES, INC

By: /s/ Randall J. Erickson

Name: Randall J. Erickson Title: Vice President and Secretary

WPM, L.P.

By: WPM GP, LLC, its general partner

By: /s/ James Neary

Name: James Neary Title: Managing Director

SIGNATURE PAGE TO SHAREHOLDERS AGREEMENT

## METAVANTE TECHNOLOGIES, INC. STOCK PURCHASE RIGHT AGREEMENT

Stock Purchase Right Agreement, dated as of November 1, 2007, (as it may be amended from time to time, this "<u>Agreement</u>") between Metavante Technologies, Inc., a Wisconsin corporation (the "<u>Company</u>"), and WPM, L.P., a Delaware limited partnership ("<u>Investor</u>").

WHEREAS, pursuant to an Investment Agreement, dated as of April 3, 2007 (the "<u>Investment Agreement</u>") among the Company, Marshall & Ilsley Corporation, a Wisconsin corporation ("<u>MI Corp</u>"), Metavante Corporation, a Wisconsin corporation, Montana Merger Sub Inc., a Wisconsin corporation, and Investor, Investor has agreed to acquire, on the terms and subject to the conditions set forth in the Investment Agreement, newly issued shares of Class A common stock, par value \$0.01 per share, of the Company, which shares shall be converted into shares of common stock, par value \$0.01 per share, of the Company (the "<u>Common Shares</u>"); and

WHEREAS, the parties intend that on the terms and subject to the conditions hereof, Investor will own 25% of the Common Shares, on a fully diluted basis, upon consummation of the Share Issuance (as defined in the Investment Agreement) and the purchase of all the Subject Shares, and are entering into this Agreement in furtherance of that connection;

WHEREAS, this Agreement shall be effective as of the Closing Date of the Investment Agreement (the "Effective Time").

1. <u>Purchase Right</u> Upon the terms and subject to the conditions set forth in this Agreement, the Company hereby grants to the Investor the right to purchase from the Company (the "<u>Purchase Right</u>") the Subject Shares at the Purchase Prices; provided, however, that notwithstanding anything to the contrary contained in this Agreement, the total number of Subject Shares that may be purchased under this Agreement shall equal one third of the aggregate number of Common Shares that may be issued under the Subject Employee Options as of immediately following the Distribution, subject to reduction, if any, pursuant to Section 3.1(b) hereof. Immediately prior to the Effective Time, there were (i) options to purchase 3,833,566 shares of common stock of MI Corp. outstanding that will be converted into Subject Employee Options pursuant to Section 6.2(a) of the Employee Matters Agreement and (ii) options to purchase 1,898,750 shares of common stock of MI Corp. outstanding that will be converted into Subject Employee Options pursuant to Section 6.2(c) of the Employee Matters Agreement (the options referred to in clause (i) and (ii) being referred to collectively herein as the "<u>Applicable MI Options</u>"). Within five business days after the determination of the number of Subject Employee Options into which the Applicable MI Options are convertible pursuant to the Employee Matters Agreement, the Company shall deliver to Investor a schedule setting forth, with respect to each Subject Employee Option into which the Applicable MI Options subject Employee Options underlying such Subject Employee Option.

2. <u>Expiration Date</u>. In no event may the Purchase Right be exercised, in whole or in part, after the earlier of (i) the date that is forty-five days after the Quarterly Notice (as defined herein) is given in respect of the calendar quarter in which all Subject Employee Options expire, (ii) the date that all Subject Shares (as they may have been reduced pursuant to Section 3.1(b)) have been purchased by the Investor or (iii) ten years from the date hereof, unless the Board shall extend the expiration date of any of the Subject Employee Options beyond the end of such ten-year period, in which case the Purchase Right shall be similarly extended (the "Expiration Date").

#### 3. Exercise of Purchase Right.

#### 3.1. Quarterly Notice and Reduction of Right

(a) No later than the last day of each month following the end of each calendar quarter prior to the Expiration Date, the Company shall give the Investor a notice setting forth the following: (i) the aggregate number of Common Shares issued during such quarter upon the exercise of Subject Employee Options (ii) the aggregate exercise price of such Subject Employee Options for such Common Shares, and (iii) the Subject Employee Options that expired unexercised or were forfeited during such quarter (the "<u>Quarterly Notice</u>"). The Quarterly Notice shall be accompanied by a schedule setting forth, in the form of tranches of the same exercise dates and exercise prices, all unexercised Subject Employee Options as of the end of such quarter.

(b) The Subject Shares shall be automatically reduced by a number equal to one third of the Common Shares issuable (x) under Subject Employee Options that expire unexercised or are forfeited and (y) under Out of the Money Options as provided in Sections 3.2(a) and 3.2(c).

#### 3.2. Method of Exercise.

(a) The Purchase Right shall automatically be deemed exercised to purchase a whole number of Subject Shares equal to one third of the aggregate number of Common Shares issued under the Subject Employee Options during each calendar quarter the exercise prices of which equal or are less than the Fair Market Value as of the date of exercise of the Purchase Right for such Subject Shares (each such Subject Employee Option, an "<u>In-the Money Option</u>") and for an aggregate Purchase Price equal to one third of the aggregate exercise prices of such In-the-Money Options for such Common Shares, in each case as specified in the Quarterly Notice with respect to such quarter (it being understood that this number shall not be reduced for any such Common Shares that are withheld from employees to pay the exercise price of such Subject Employee Options, or any withholding taxes due, pursuant to net vesting settlement and similar provisions). Such purchase shall take place 45 days following the date the Quarterly Notice is given (or the first business day following such 45<sup>th</sup> day, if such day is not a business day). Following the Quarterly Notice and prior to such date of purchase, the Investor may deliver to the Company a notice (the "<u>Cash Payment Notice</u>") electing to pay such Purchase Price by a Cash Payment, in which case the Cash Payment is not so

made with respect to any Quarterly Notice, such Purchase Price shall be paid by the Company withholding from the number of Subject Shares to be delivered to the Investor a number of Subject Shares having an aggregate Fair Market Value, determined as of the close of business on the business day immediately before the date of purchase, equal to such Purchase Price, which date shall also be deemed the date of exercise of the Purchase Right for purposes of determining the In-the Money Options and Out of the Money Options. Any fraction of a Subject Share which would be required to pay such Purchase Price shall be disregarded and the remaining amount due shall be paid in cash by the Investor. Upon the purchase of any Subject Shares pursuant to this Section 3.2(a), the number of Subject Shares remaining shall be reduced by the number of Subject Shares so purchased. The Subject Shares shall also be reduced by a number equal to one third of the number of Common Shares issued during each calendar quarter pursuant to Out of the Money Options.

(b) In the event the Investor sells, transfers, assigns or otherwise disposes of (whether by operation of law or otherwise) (but only in the event that the Purchase Right is not accelerated under Section 3.2(c) in connection with such event), to a third party that is not an affiliate of the Investor or distributes to its limited partners (collectively, "<u>Transfers</u>"), any of the Common Shares it acquired on the date of the Distribution, but not any Common Shares that it thereafter acquired in excess of such Common Shares, it may exercise the Purchase Right for a whole number of Subject Shares equal to the applicable Acceleration Subject Shares and for a purchase price equal to the related Acceleration Purchase Price, by delivering to the Company an irrevocable exercise notice within 10 days of such sale (the "<u>Acceleration Notice</u>"). The Acceleration Notice shall set forth the number of Common Shares that have been sold by the Investor, the dates of sales thereof, shall certify that such Notice is being given in accordance with Section 3.2(b), and shall specify whether the Investor wishes to pay the Purchase Price by a Cash Payment or through the Company withholding from the Subject Shares to be delivered to the Investor a number of Subject Shares having an aggregate Fair Market Value, determined as of the date the Acceleration Notice is given, equal to the aggregate Acceleration Purchase Price. Within 10 business days of receiving the Acceleration Notice, the Company shall give the Investor notice (the "<u>Acceleration Details Notice</u>") of the Acceleration Purchase Price applicable to the Acceleration Notice as well as of its calculation of the number of Acceleration Subject Shares being purchase by the Investor pursuant to such Acceleration Notice. In the event that Investor elected to pay the Acceleration Details Notice. Upon the Acceleration Subject Shares perice in cash, it shall deliver the Acceleration Subject Shares pursuant to this Section 3.2(b), the number of Subject Shares remaining shall be reduced b

(c) Immediately prior to (i) any event causing the simultaneous acceleration of the vesting, or automatic exercise, of all the Subject Employee Options or (ii) a merger or other business combination involving the Company in which the Common Shares are converted into the right to receive cash in exchange for such Common Shares, the Purchase Right shall automatically be deemed exercised for all Subject Shares then still subject to the Purchase Right. The Purchase Price shall be an amount equal to the related Acceleration Purchase Price. Such Purchase Price shall be paid by the Company withholding from the number of Subject Shares to be delivered to the Investor a number of Subject Shares having

an aggregate Fair Market Value, determined as of three business days before the date of such acceleration, equal to such Purchase Price. Any fraction of a Subject Share which would be required to pay such Purchase Price shall be disregarded and the remaining amount due shall be paid in cash by the Investor. The Subject Shares shall be reduced by a number equal to one third of the number of Common Shares subject to Out of the Money Options as of the date of an acceleration pursuant to this Section 3.2(c).

(d) The Purchase Right may be exercised by the Investor solely as and to the extent expressly set forth in this Section 3.2. In no event may the Purchase Right be exercised after it terminates as set forth in Section 2. No certificate representing a Subject Share shall be delivered until the full purchase price therefore has been paid. Notwithstanding anything to the contrary contained in this Agreement, the Company shall have no obligation to issue any fraction of a Subject Share under this Agreement, all of which shall be disregarded.

### 4. Additional Terms and Conditions of Purchase Right.

4.1. <u>Nontransferability of Purchase Right</u>. The Purchase Right is exercisable only by the Investor. The Purchase Right may not be sold, transferred, assigned, pledged, hypothecated, encumbered or otherwise disposed of (whether by operation of law or otherwise) or be subject to execution, attachment or similar process. Any attempt to so sell, transfer, assign, pledge, hypothecate, encumber or otherwise dispose of the Purchase Right, shall be null and void.

# 4.2. Investment Representation.

(a) The Investor hereby represents and warrants that (a) any Common Shares purchased upon exercise of the Purchase Right will be purchased for investment and not with a view to the distribution thereof within the meaning of the Securities Act of 1933, as amended (the "<u>Securities Act</u>"), unless such purchase has been registered under the Securities Act and any applicable state securities laws; (b) any subsequent sale of any such shares shall be made either pursuant to an effective registration statement under the Securities Act and any applicable state securities laws, or pursuant to an exemption from registration under the Securities laws; and (c) if requested by the Company, the Investor shall submit a written statement, in form reasonably satisfactory to the Company, to the effect that such representation (x) is true and correct as of the date of purchase of any shares hereunder or (y) is true and correct as of the date of any sale of any such shares, as applicable.

(b) All Subject Shares issued under this Agreement shall bear the legend specified in Section 6.3 of the Shareholders Agreement.

4.3. <u>Adjustment</u>. In the event of any adjustment (i) in the Common Shares issuable upon exercise of Subject Employee Options or (ii) the terms of any of the Subject Employee Options, including the exercise prices, in each case including as a result of stock split, stock dividend, recapitalization, reorganization, merger, consolidation, combination, exchange of shares, liquidation, spin-off or other similar change in capitalization or event, the Subject Shares

and the terms and conditions thereof (including without limitation the Purchase Price thereof) shall be equitably adjusted by the Board in the same manner as the Subject Employee Options.

4.4. <u>Compliance with Applicable Law</u>. The Purchase Right is subject to the condition that if the listing, registration or qualification of the Subject Shares upon any securities exchange or under any law, or the consent or approval of any governmental body, or the taking of any other action is necessary or reasonably desirable as a condition of, or in connection with, the purchase or delivery of Subject Shares, the Purchase Right may not be exercised, in whole or in part, unless such listing, registration, qualification, consent or approval shall have been effected or obtained. The Company and the Investor agree to use reasonable efforts to effect or obtain any such listing, registration, qualification, consent or approval.

4.5. <u>Delivery of Certificates</u>. Upon the exercise of the Purchase Right, in whole or in part, the Company shall deliver or cause to be delivered one or more certificates representing the number of shares purchased against full payment therefore, subject to Section 4.2(b).

4.6. <u>Purchase Right Confers No Rights as Stockholder</u>. The Investor shall not be entitled to any privileges of ownership with respect to Subject Shares unless and until purchased and delivered upon the exercise of the Purchase Right, in whole or in part, and the Investor becomes a stockholder of record with respect to such delivered shares; and the Investor shall not be considered a stockholder of the Company with respect to any such shares not so purchased and delivered previously.

4.7. <u>Company to Reserve Shares</u>. The Company shall at all times prior to the expiration or termination of the Purchase Right reserve and keep available, either in its treasury or out of its authorized but unissued common shares, the full number of shares subject to the Purchase Right from time to time.

4.8. Shareholders Agreement. Any Subject Shares issued upon exercise of the Purchase Right shall be subject to the provisions of the Shareholders Agreement, and shall be shares of "Common Stock" that are "Beneficially Owned" by Investor for purposes of the Shareholders Agreement; provided, however, that no exercise of the Purchase Right shall in itself constitute a violation of Section 3.2(a) of the Shareholders Agreement. Without limiting the generality of the foregoing, such Subject Shares shall be subject to (i) the registration rights provisions of Article II of the Shareholders Agreement, (ii) the transfer restriction provisions of Section 3.1 of the Shareholders Agreement, and (iii) the provisions of Section 6.3.

4.9. Defined Terms. Capitalized terms used in this Agreement have the following meanings:

"<u>Acceleration Purchase Price</u>" shall mean with respect to any Acceleration Subject Shares, one third of the aggregate exercise price of the Subject Employee Options to the extent used in determining such Acceleration Subject Shares.

"<u>Acceleration Subject Shares</u>" shall mean (x) in the case of Section 3.2(b) a number of Subject Shares equal to one third of a percentage of the Reference Common Shares



that is equal to the percentage of the Common Shares transferred by the Investor and in respect of which an Acceleration Notice had not been delivered previously, and (y) in the case of Section 3.2(c) a number of Subject Shares equal to one third of all Common Shares subject to then outstanding Subject Employee Options the exercise prices of which equal or are less than the Fair Market Value as of the date of an acceleration pursuant to Section 3.2(c).

"Board" shall mean the Board of Directors of the Company, excluding any Investor Designees (as defined in the Shareholders Agreement).

"Cash Payment" shall mean a wire transfer of immediately available funds to such account as the Company may specify from time to time.

"Distribution" shall have the meaning ascribed thereto in the Employee Matters Agreement.

"<u>Employee Matters Agreement</u>" shall mean that certain Employee Matters Agreement, dated as of April 3, 2007, between the Company, New M&I Corporation, and the other parties thereto, as amended.

"Fair Market Value" shall mean the closing transaction price of a Common Share as reported in the New York Stock Exchange Composite Transactions (or the equivalent reporting system for any other national securities exchange on which the Common Shares are primarily listed) on the date as of which such value is being determined or, if there shall be no reported transactions for such date, on the next preceding date for which transactions were reported; provided, however, that if the Common Shares are not listed on any national securities exchange, the Fair Market Value may be determined by the Board by whatever means or method as the Board, in the good faith exercise of its discretion, shall at such time deem appropriate.

"MVT Option" shall have the meaning ascribed thereto in the Employee Matters Agreement

"<u>Out of the Money Options</u>" shall mean (x) in the case of Section 3.2(a), Subject Employee Options the exercise prices of which are greater than the Fair Market Value as of the date of exercise of the Purchase Right for such Common Shares, and (y) in the case of Section 3.2(c), Subject Employee Options the exercise prices of which are greater than the Fair Market Value as of the date of an acceleration pursuant to such Section 3.2(c).

"Purchase Prices" shall mean the purchase prices for which the Investor may purchase Subject Shares hereunder.

"<u>Reference Common Shares</u>" shall mean, as of any time of determination, the Common Shares subject to those Subject Employee Options (i) that are outstanding, unexercised and vested, (ii) the exercise prices of which equal or are less than the Fair Market Value as of such date, (iii) not previously used in determining the Acceleration Subject Shares in connection with any Acceleration Notice, and (iv) have the earliest grant dates (when compared to other Subject Employee Options that meet the specifications in clause (i) – (iii) immediately above).

"<u>Shareholders Agreement</u>" shall mean that certain Shareholders Agreement, dated as of November 1, 2007, among the Company, the Investor and any other Shareholders (as defined therein) that become a party thereto, as amended from time to time.

"<u>Subject Employee Options</u>" shall mean the MVT Options outstanding effective immediately after the Distribution.

"Subject Shares" shall mean the Common Shares issuable pursuant to Section 3 hereof.

#### 5. Miscellaneous Provisions.

5.1. <u>Successors</u>. This Agreement shall be binding upon and inure to the benefit of the Investor, the Company and the successors and assigns of the Company. The Investor may not assign any of its rights or obligations under this Agreement, whether by operation of law or otherwise. Nothing in this Agreement, express or implied, is intended to or shall confer upon any other person any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement

5.2. Notices. All notices and other communications hereunder shall be in writing and shall be deemed duly given (a) on the date of delivery if delivered personally, (b) upon confirmation of receipt if delivered by telecopy or telefacsimile, (c) on the first business day following the date of dispatch if delivered by a recognized next-day courier service, or (d) on the date received if delivered by registered or certified mail, return receipt requested, postage prepaid. All notices hereunder shall be delivered as set forth below, or pursuant to such other instructions as may be designated in writing by the party to receive such notice:

if to the Company to:

Metavante Technologies, Inc. 4900 West Brown Deer Rd. Milwaukee, Wisconsin 53223 Fax: (414) 362-1705 Attention: Norrie J. Daroga, Esq.

with a copy to:

Quarles & Brady LLP 411 East Wisconsin Avenue Milwaukee, Wisconsin 53202-4497 Fax: (414) 978-8786 Attention: Conrad G. Goodkind, Esq.

if to Investor, to:

WPM, L.P. c/o Warburg Pincus & Co. 466 Lexington Avenue New York, New York 10017 Fax: (212) 878-9351 Attention: James Neary

with a copy to:

Wachtell, Lipton, Rosen & Katz 51 West 52nd Street New York, New York 10019 Fax: (212) 403-2000 Attention: Andrew R. Brownstein, Esq. Igor Kirman, Esq.

or to such other persons or addresses as may be designated in writing by the party to receive such notice as provided above.

5.3. <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of Wisconsin (without giving effect to choice of law principles thereof).

5.4. <u>Consent to Jurisdiction</u>. Each of Investor and the Company irrevocably agrees that any legal action or proceeding with respect to this Agreement, any provision hereof, the breach, performance, validity or invalidity hereof or for recognition and enforcement of any judgment in respect hereof brought by another party hereto or its successors or permitted assigns may be brought and determined in any federal or state court located in the State of Wisconsin, and each of Investor and the Company hereby irrevocably submits with regard to any such action or proceeding for itself and in respect to its property, generally and unconditionally, to the exclusive jurisdiction of the aforesaid courts. Each of Investor and the Company hereby irrevocably waives, and agrees not to assert, by way of motion, as a defense, counterclaim or otherwise, in any action or proceeding with respect to the jurisdiction of the above-named courts for any reason other than the failure to lawfully serve process, (ii) that it or its property is exempt or immune from jurisdiction of judgment, execution of judgment or otherwise) and (iii) to the fullest extent permitted by applicable laws, that (A) the suit, action or proceeding in any such court is brought in an inconvenient forum, (B) the venue of such suit, action or proceeding is improper and (C) this Agreement, or the subject matter hereof, may not be enforced in or by such courts.

5.5. <u>Waiver of Jury Trial</u>. EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE

EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (a) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, TO IT THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (b) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (c) EACH PARTY MAKES THIS WAIVER VOLUNTARILY AND (d) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 5.5.

5.6. <u>Counterparts</u>. This Agreement may be executed in two counterparts each of which shall be deemed an original and both of which together shall constitute one and the same instrument.

5.7. Descriptive Headings. The descriptive headings of this Agreement are inserted for convenience only and do not constitute a part of this Agreement.

5.8. <u>Amendments and Waivers</u>. The provisions of this Agreement may be amended or waived only upon the prior written consent of the Company (to the extent approved by a majority of Independent Directors who are not Investor Designees, each as defined in the Shareholders Agreement) and Investor .

5.9. <u>Entire Agreement</u>. This Agreement constitutes the entire agreement and supersedes all prior agreements, understandings, representations and warranties, both written and oral, among the parties with respect to the subject matter hereof and thereof.

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement by their authorized representatives as of the date first above written.

METAVANTE TECHNOLOGIES, INC.

By: /s/ Randall J. Erickson

Name: Randall J. Erickson Title: Vice President and Secretary

WPM, L.P.

By: WPM GP, LLC, its general partner

By: /s/ James Neary

Name: James Neary Title: Managing Director